### DESIGN OF AN E-LEARNING SYSTEM USING SEMANTIC INFORMATION AND CLOUD COMPUTING TECHNOLOGIES

by

### AHMED BADAWY FATHELBAB MOHAMMED

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in

Telematics Engineering

Universidad Carlos III de Madrid

Advisor:

Jesus Arias Fisteus

May 2023

This thesis is distributed under license "Creative Commons Attribution – Non Commercial – Non Derivatives".



# Design of an E-learning system using semantic information and cloud computing technologies

Prepared by: Ahmed Badawy Fathelbab Mohammed, Minia University (Egypt), Universidad Carlos III de Madrid (Spain). Contact: <u>ahmed.badawy2000@mu.edu.eg</u>

Under the advice of: Jesus Arias Fisteus, Telematic Engineering Department, Universidad Carlos III de Madrid, (Spain).

Unless otherwise indicated, the content of is thesis is distributed under a Creative Commons Attribution-ShareAlike 4.0 International (CC BY-SA).

### ACKNOWLEDGEMENTS

Thank Allah for his many blessings and grace; thank you professor Jesus Arias Fisteus, my PhD supervisor, thank you for your help, time and effort; I also thank my beloved wife, thank you for all your support, assistance and patience.

### **Published and Submitted Content**

This thesis is based on the following published papers:

[1] Ahmed Badawy; Jesus A. Fisteus; Tarek M. Mahmoud; Tarek Abd El-Hafeez. Topic Extraction and Interactive Knowledge Graphs for Learning Resources. Published in: Sustainability 2022, Volume 14, Issue 1, 226

- This work is wholly included and its content is reported in different parts of this thesis.
- The author's role in this work is focused on the design, implementation and simulations of the concepts proposed in the paper.

### **Project Acknowledgements**

Part of the research in this thesis has been supported by the Egyptian Ministry of Higher Education and Minia University in the Arab Republic of Egypt.

### Abstract

Humanity is currently suffering from many difficult problems that threaten the life and survival of the human race. It is very easy for all mankind to be affected, directly or indirectly, by these problems. Education is a key solution for most of them. In our thesis we tried to make use of current technologies to enhance and ease the learning process.

We have designed an e-learning system based on semantic information and cloud computing, in addition to many other technologies that contribute to improving the educational process and raising the level of students. The design was built after much research on useful technology, its types, and examples of actual systems that were previously discussed by other researchers.

In addition to the proposed design, an algorithm was implemented to identify topics found in large textual educational resources. It was tested and proved to be efficient against other methods. The algorithm has the ability of extracting the main topics from textual learning resources, linking related resources and generating interactive dynamic knowledge graphs. This algorithm accurately and efficiently accomplishes those tasks even for bigger books. We used Wikipedia Miner, TextRank, and Gensim within our algorithm. Our algorithm's accuracy was evaluated against Gensim, largely improving its accuracy.

Augmenting the system design with the implemented algorithm will produce many useful services for improving the learning process such as: identifying main topics of big textual learning resources automatically and connecting them to other well defined concepts from Wikipedia, enriching current learning resources with semantic information from external sources, providing student with browsable dynamic interactive knowledge graphs, and making use of learning groups to encourage students to share their learning experiences and feedback with other learners.

### Table of Contents

Acknowledgements	
Published and Submitted Content	
Project Acknowledgements	
Abstract	
Table of Contents	
List of Figures	
List of Acronyms	
Chapter 1: Introduction	2
1.1 Motivation	2
1.1.1 Importance of education for society	2
1.1.2 Importance of education for individuals	10
1.1.3 Importance of education in the research field	12
1.2 Education vs. learning	14
1.2.1 Education	14
1.2.2 Learning	16
1.2.3 Learning theories	18
1.2.4 Learning vs. Education	18
1.2.5 E-learning	19
1.2.6 Evolution of distance learning and e-learning	21
1.3 Objective	22
1.3.1 Challenges	22
1.3.2 Contributions	22
1.4 Outline of the thesis	23
Chapter 2: Background and State of the Art	24
2.1 E-learning	24
2.1.1 Adaptive e-learning	25
2.1.2 Augmented reality (AR)	27
2.2 Semantic Information	28
2.3 Semantic information and e-learning	30
2.4 Cloud Computing	35
2.4.1 Cloud computing and e-learning	39
2.5 Learning object recommendation	45
2.5.1 Types of recommender systems	46
2.6 Learning object annotation	49
2.6.1 Metadata Standards	50
2.6.2 Examples	51
2.7 Topic identification	53
2.7.1 Data mining techniques for topic identification	54
2.7.2 Research examples on topic identification	55
2.7.3 Enrich topic identification with new ideas	56
2.8 Important tools for our system	57
2.8.1 Wikipedia Miner	57
2.8.2 TextRank	58
2.8.3 BM25	59
2.8.4 Gensim	60

Chapter 3: Our system Contribution	61
3.1 E-learning perspective	61
3.2 Semantic web perspective	63
3.2.1 Learner	63
3.2.2 Learning domain	65
3.2.3 Learning object	66
3.2.4 Enriching learning content	68
3.2.5 Knowledge discovery	69
3.2.6 Learning path generation	70
3.2.7 Semantic recommender system	71
3.3 Cloud computing perspective	75
3.4 Augmenting all the three perspectives	79
Chapter 4: An algorithm for topic identification in textual learning	90
resources	80
4.1 Introduction	80
4.2 Algorithm phases	81
4.2.1. Extract main keywords	81
4.2.2. Identify main topics	81
4.3 Dynamic interactive knowledge graphs	84
Chapter 5: Experiments and Results	87
5.1 Experiment 1	87
5.2 Experiment 2	89
5.2.1 Computation time	90
5.2.2 Accuracy	93
5.2.3 Accuracy vs. text size	94
· · · · · · · · · · · · · · · · · · ·	
Chapter 6: Conclusion and Future Work	97
6.1 Research Restrictions and Practical Implications	98
6.2 Future work	98
References	100

	-	
1.1	South Korea in the past and now	5
1.2	Singapore in the past and now	5
0.1	number of researches on "e-learning" in SCOUPUS	25
2.1	database during 1990-2022	25
2.2	number of researches on "semantic web" in SCOUPUS	20
2.2	database during 1990-2022	29
2.2	number of researches on "ontology" in SCOUPUS	20
2.3	database during 1990-2022	29
2.4	number of researches on "semantic" and "e-learning" in	20
2.4	SCOUPUS database during 1990-2022	30
2.5	e-learning vs. semantic e-learning	31
2.6	Learner, Learning Content, and Domain Ontologies	34
2.7	The recommendation process	34
20	number of researches on "Cloud computing" in	20
2.0	SCOUPUS database during 1990-2022	30
2.0	"cloud computing", "Internet of things", "fog computing"	26
2.9	on Google trends	30
2.10	The relationship between cloud services	37
2.11	SaaS, PaaS, and IaaS	37
2 12	number of researches on "cloud based learning" in	30
2.12	SCOUPUS database during 1990-2022	57
2 13	Cloud computing infrastructure-based e-learning ecosystem	43
2.13	architecture	ъJ
2 14	the framework of the cloud-based reflective learning	44
2.17	environment	TT
	number of researches on "Learning object	
2.15	recommendation" in SCOUPUS database during 1990-	45
	2022	
2.16	The general e-learning recommendation process	46
2.17	number of researches on "Learning object annotation" in	50
	SCOUPUS database during 1994-2022	
2.18	Architecture of KIM platform	52
2.19	Flow diagram of KIM semantic information extraction	53
2.20	number of researches on "Topic extraction" in SCOUPUS	54
	database during 1990-2022	
2.21	number of researches on "Entity extraction" in SCOUPUS	54
0.00	database during 1990-2022	
2.22	a description of the Wikipedia miner toolkit's architecture	57
3.1	required steps before logging into the system	62
3.2	learning resources preparation process	62
3.3	the continuous observation process	62
3.4	Student group recommendation	63
3.5	Some leaner's properties from FOAF Ontology	64
3.6	Learner Ontology	64
3.7	Our representation for Learner Ontology	65
3.8	learning domain example for computer science subject	66
3.9	Learning object representations	66

### List of Figures

3.10	Example of required Learning object representations	67
3.11	IEEE LOM elements	67
3.12	Learning Content Ontology	68
3.13	Components of LOD (linked open data) tool [154] for	69
3 1/	The knowledge discovery process of the aIBM tool	70
3.14	The training path generation process of the effect tool	70
3.15	The customizable learning path generation process	70
3.10	The customizable learning path generation process	71
3.17	The recommendation process	71
3.10	Our recommendation process	72
3.19	Service models for cloud computing	75
3.20	Pervice models for cloud computing	76
3.21	Required cloud services models for our system	70
3.22	Class have reacting in fragment models for our system	/0
3.23	architecture	77
3.24	Framework of the cloud-based reflective learning	77
3.25	Our system from the cloud computing perspective	78
3.26	Our system design from the three perspectives	79
4.1	Structure of our algorithm	82
	Relation between learning resource (in blue), and main	
4.2	topics (in yellow)	84
4.3	Relation between learning resources (in blue), and main topics (in yellow)	85
4.4	Relation between one learning resource (blue), main topics (red), and chapters (yellow)	85
4.5	Relation between learning resources (blue), main topics (yellow), and chapters (red)	86
5.1	Comparison between the four levels of parent-category	88
5.2	Number of pages vs. number of books	89
5.3	Text size vs. number of books	90
5.4	Computation time vs. text size	91
5.5	Average computation time vs. text size	91
5.6	Gensim (computation time vs. text size)	92
5.7	Our algorithm (computation time vs. text size)	92
5.8	Top-K accuracy	94
5.9	Accuracy vs. text size (all)	95
5.10	Average accuracy vs. text size (detailed)	95
5.11	Average accuracy vs. text size	96

### List of Tables

Table 1.1: Japan after the war and now	4
Table 1.2: Germany after the war and now	6
Table 1.3: United States of America after the war and now	6
Table 1.4: Open School viewpoint in Denmark	7
Table 1.5: Education rankings by country 2021	8
Table 1.6: Education rankings sorted by population 2021	9
Table 1.7: Number of published documents by year (1990-2022)	13
Table 1.8: Number of published documents by country	13
Table 1.9: Number of published documents by subject area	14
Table 1.10: Differences between Education and Learning	18
Table 1.11: Definitions of e-learning	20
Table 2.1: Examples of cooperation from the nature	35
Table 4.1. Examples of our algorithm's results	83
Table 5.1: Comparison between the four levels	88
Table 5.2: Total computation time	92
Table 5.3: Example of results	93
Table 5.4: Top-10 accuracy	93
Table 5.5: Top-k accuracy	94

### List of Acronyms

AI	artificial intelligence
AR	Augmented reality
ASP	Application Service Provider
AWS	Amazon Web Services
CBT	computer-based training
DaaS	Data as a Service
eJRM	electronic Justice
FOAF	friend of a friend ontology
GAE	Google App Engine
HaaS	Hardware as a Service
IaaS	Infrastructure as a Service
IS	Information Systems
LDA	Latent Dirichlet allocation
LMS	learning management system
LOD	linked open data
LOM	Learning Object Metadata
LSI	Latent semantic indexing
MBTI	Myers-Briggs Type Indicator questionnaire
MOOCs	massive open online courses
NER	Named Entity Recognition
NLP	natural language processing
PaaS	Platform as a Service
RDF	Resource Description Framework
SaaS	Software as a Service
SCROM	Shareable Content Object
Storage aaS	Storage as a service
SVD	Singular Value Decomposition
XML	eXtensible Markup Language

## 1. Introduction

### **1.1 Motivation**

Humanity is currently suffering from many difficult problems that threaten the life and survival of the human race. It is very easy for all mankind to be affected, directly or indirectly, by these problems, for two main reasons:

1-As a result of the great development in the means of communication and transportation, and because we live on one planet, these problems do not threaten one community or one town. 2-Because we did not seek to solve them for decades, these problems have become huge and hard to be solved.

The misuse of great scientific progress may be a major cause of many of these problems, but there is no doubt that mankind is the main actor and the only cause of this huge corruption. Humanity has taken the path of progress and scientific research. Unfortunately, we cannot confirm how this path will be ended, but the end is either prosperity and happiness, or destruction and annihilation. The confirmation of the result depends on the strength of the teams working in both directions (the use of education to build, or the use of education to destroy). Any effort for reform will be effective, even if it is small. What is more important than the amount of change is continuity and perseverance in doing it.

Here, a pivotal conclusion can be reached that education is an effective and powerful weapon that can be used for reconstruction and prosperity, or for sabotage and destruction.

"Education is the most powerful weapon which you can use to change the world." Nelson Mandela [1]

But there is no escape from using education to fix what has been corrupted and is still corrupting by mankind.

Was the use of education effective against mankind made problems? To answer this question, it is necessary to research the subject and refute historical facts about the impact of education on societies and individuals, and this is what we are about to do in this chapter.

#### 1.1.1 Importance of education for society

To study the effect of education on societies we are going to explore some history facts of some countries such as Spain, Finland, Japan, South Korea, Singapore, Copenhagen, and Denmark.

Starting with an example of underdeveloped countries in the 1900, Clàudia Canals [2] presented a comparison between Spain and Finland. She argued that both countries were largely agricultural countries, most of the people were not educated, and the illiteracy rate was high with a percent of 40%. The per capita income level was also similar between the two countries.

The surprise appeared after fifty years, where Finland's income doubled Spain's income, all Finnish people were educated, and an obvious widespread of secondary education could be mentioned in all social classes. At the same time, illiteracy was still dominant in Spanish society, and secondary education was rare among the people.

After seventy years of education development and progress and the rise of the Spanish economic level, Finland is still ahead of Spain in terms of education and economic level.

According to these facts and results, what is the secret?

The two countries started the race at the same time and from the same conditions and available capabilities. But the difference was huge, about seventy years, equal to the age of an entire generation, who could have lived a comfortable life in light of better education, health and financial standard. But they were deprived of this life only because of the delay in the decision to develop education. If this is not the main reason, then at least it must be a big factor in it. As Clàudia argues "Education directly affects economic growth insofar as it is essential to improve human capital." [2]

It is worth noting that Spain was suffering from a devastating civil war during the period from 1936 to 1939, and this war may be the reason for part of the difference between the two countries. [3]

Many factors could affect the economic level of the country and individuals: natural resources, available technological tools, number of labor force and demographic distribution, especially the level of efficiency and training of this workforce or "human capital" that is determined by the stock of knowledge, skills and habits. Here, the direct impact of education quality appears on the level of the workforce, and consequently on improving and raising the economic level of the individual and society. Clàudia summarized that as "The increase in workers' educational level improves their human capital, increasing the productivity of these workers and the economy's output." [2]

Finland has introduced some key points to make sure that education remains effective such as: (1) Respecting education is a part of their identity, (2) Not every individual has the ability to be a teacher, (3) Less study time and more enjoy time, (4) There is no separation between students on the basis of their academic level, (5) The intensity of the connection between the student and the teacher, (6) Equality between students.

Another example that shows the miraculous effect of education and the need to rely on it for the advancement of nations is the Japanese model. They relied on education to solve and face many massive problems such as the societal transformation from a closed society to an open industrial society that wants to compete to reserve a seat among the major industrial countries. They used education to face challenges presented by the need to quickly absorb Western ideas, science, and technology in the Meiji period (1868-1912) [4], when the Japanese people moved from being an isolated feudal society to the new paradigm of a modern, industrialized nation state. Meiji leaders acted rapidly to put a new educational system as a keystone to catch up with the West and promote national unity [5].

The role of education did not end here for them, as it was still effective in facing the next challenges, which were wars and their devastating impact on countries. They left them in the worst conditions, especially with the presence of a new lethal and destructive weapon called the nuclear bomb. Infrastructure, economic resources, manpower, and any foreign desire to invest are destroyed by wars. This is exactly what happened to Japan after the defeat in World War II 1945.

If any nation is in these circumstances, it has two expected options: crying over the ruins and heavy losses, pleading and begging other nations for help, or getting up and reorganizing forces and resources available for reconstruction and development.

The second choice was the right choice and the winning bet for Japan. And because they realized the importance of investing in education and its influence in building and restoring strength, it was their main weapon against the total destruction that subdued their necks. So, directly after

the war ended they restarted classes upon the destroyed schools [6]. Table 1.1 presents images of Japan after the war and now.



#### Table 1.1: Japan after the war and now

https://en.wikipedia.org/wiki/File:HiroshimaPeaceMemorialPanorama-2.jpg

Their equation for renaissance was: Education + morals + work = renaissance

The destruction they were subjected to was not the end of the war: after that they were subjected to the cultural occupation that almost dominated the minds of their children and erased their identity and their cultural history from the minds that were still forming. Therefore, they relied once again on their effective weapon (education) in confronting the Western cultural invasion.

As we have seen in the previous examples, the power of education in facing many challenges and its effectiveness against difficulties of all kinds became clear. It was no secret to the sincere people of other countries the need to use education for the advancement and continuity of their countries in the forefront, or to improve the general country and the people. So other countries sought to catch up with the train of evolution through its only gateway, which is education, such as South Korea, Singapore, and Denmark.

South Korea had keystones for a powerful education: good organization, good upbringing, morals and equality are the basis of their education system. Figure 1.1 shows South Korea in the past and now.

Because of their weak economic resources and lack of water, Singapore faced a severe economic crisis when it separated from Malaysia. Singapore had people of different races, nationalities, religions and languages who had to cooperate and unite to build their country. The goal and language were unified for the advancement of the country by selecting the English language as the first language of the country, and setting the general goal of the different races on improving the economic level with diligence at work.

Singapore also realized that the economic and scientific renaissance can only come with an advanced educational system. Therefore, education was among their basic tools for renaissance and construction. Their principle was to study less and learn more, encouraging the student to self-learn. Internal motivation and passion for knowledge was their main driver of learning. Figure 1.2 shows Singapore in the past and now.



Figure 1.1: South Korea in the past and now



Figure 1.2: Singapore in the past and now

Education was and still is one of the factors greatly affecting the rise of nations from great destruction to development and prosperity, as we saw in the previous examples, and as we will see in many other countries such as Germany and the United States of America. Table 1.2 shows images of Germany after the war and now and Table 1.3 presents images of the United States of America after the war and now.

 Table 1.2: Germany after the war and now



Table 1.3: United States of America after the war and now



Denmark had presented a new viewpoint regarding education and its importance in the life of society and the individual by giving students the freedom to choose or create their own educational path. The Open School in Copenhagen, Denmark, is a current futuristic example of education development. In this school they give students the ability to set their own educational

plan, where the school is only an advisory body to help them achieve their academic goals. Table 1.4 shows images of the Open School viewpoint in Denmark.





#### Emphasizing the importance of education

After mentioning historical facts and events that happened to specific nations, which show the power and importance of education for survival and success, we will explore some opinions about the importance of education.

In World Population Review [1], and according to many references they mention, prosperity of nations and their people depends mainly on education. Moreover, the economic level is closely related to the country's level of education. By observing fully developed countries and comparing their level of education with developing countries, we find that developed countries provide their people with the best level of education. On the other hand, developing countries seek to provide a good level of education to raise the economic level of their people and improve the future of the country. [1]

The Global Partnership for Education organization mentions [7] that education, from an economic point of view, has great benefits for society, in addition to being one of the most important forms of investment in which a nation's wealth can be used to obtain the highest return in the future. It gives society a kind of reassurance and hope for the future and social stability, ensures progress and development in the growth of the nation's economy, improves mental and physical health and provides sustenance at a high level of quality and prosperity. It is also an effective and vital course to achieve the goals of sustainable growth for future generations. [7]

From Ajay's [8] point of view, education has great importance in the lives of individuals for many reasons, including: (1) education transforms the usual daily information in the lives of individuals into knowledge, and knowledge acquisition is the only thing that gives people the ability to define their position and opinion from the different situations they encounter during their lives; (2) education is responsible for transforming humans from primitive beings fighting over food and other animal instincts to sublime beings that think, research and contemplate life and the environment; (3) humans, compared to the nature in which they live, are in the midst of physical weakness, but education gives humans confidence in themselves as beings that have the ability to learn, explore, and invent to deal with forces greater than themselves and overcome difficult situations; (4) education gave people a good opportunity to communicate, integrate and interact, either by providing the required means for that or basic personal skills for expression and communication; (5) also, good education in the early stages can give the opportunity for buried

personal and individual talents to appear, grow and shine; (6) education is almost a guaranteed instrument for the successful future of the members of society and therefore the entire nation; (7) mind and personality also have a large share of the benefits of education, by opening the mind to different cultures and multiple societies and accepting new useful ideas; education is also useful in forming a normal and beneficial personality for the surrounding society; (8) education helps in learning to manage the greatest wealth of any individual and invest it properly; which is time; managing time and investing it optimally is one of the great benefits of education, which enables individuals to work within specific schedules and under different work pressures. **[8]** 

In addition to all the previous advantages, worldvision [9] highlighted more advantages of education at the level of individuals and societies. For the individual, education develops problem-solving and confrontation skills, and supports the individual to be self-reliant. As for society, education helps spread equality among individuals, supports economic stability and security, and facilitates the way for future economic growth for the nation; which prompts the nation to build a strong and effective economy in facing crises. [9]

As mentioned by UNESCO [10] as well, education leads to the economic growth of the individual, which leads to the economic growth of the entire nation. Also, the countries that enjoy high rates of education are the countries with the largest share in achieving great growth in the field of human and economic development. [10]

So far, we see that a high level of education clearly helps individuals and societies, another source stated that citizens with low education or low educational capabilities have a high probability of being unemployed [11]. Therefore, education has become a weapon that must be used and provided to every child from the beginning of their life to give them the ability to improve their personal life and advance their society and nation. This made education a major and fundamental human right that plays a vital and fundamental role in the development of the individual, society and the economy according to the Global Partnership for Education organization [12]. Therefore, the progress and development of countries is measured according to many criteria, but the most important one is the level, quality, and spread of education [13] [14]. With the great spread of poverty and hunger, the role of education emerges as an effective exterminator of these epidemics and a donor of opportunities for a better life. [9] [15]

Despite the enormous advantages that nations enjoy with education, the high cost of quality education is a major obstacle for many developing countries, especially with the difficult circumstances that accompanied and followed the corona virus COVID-19 pandemic [16] [17]. Therefore, we see in **Table 4** that the advanced countries in education are economically prosperous countries, which have the capabilities that qualify them to provide requirements and costs, such as the United States of America, United Kingdom, and Germany [1]. **Table 1.5** presents education rankings by country 2021. [1]

Country	Rank (2021)	Rank (2020)	2022 Population
United States	1	1	334,805,269
United Kingdom	2	2	68,497,907
Germany	3	4	83,883,596
Canada	4	3	38,388,419
France	5	5	65,584,518
Switzerland	6	6	8,773,637
Japan	7	11	125,584,838

Table 1.5: Education rankings by country 2021 [1]

Australia	8	7	26,068,792
Sweden	9	8	10,218,971
Netherlands	10	9	17,211,447
New Zealand	11	14	4,898,203
Denmark	12	10	5,834,950
Norway	13	12	5,511,370
Italy	14	16	60,262,770
Finland	15	15	5,554,960
Austria	16	13	9,066,710
Spain	17	18	46,719,142
Belgium	18		11,668,278
South Korea	19	22	51,329,899
Ireland	20		5,020,199

When studying the level of economic growth according to the number of people, we notice that China and India are at the top of the list: with more than one billion people, in 1995 the World Bank classified these two countries as low-income countries. But now, the same two countries have an economic classification of middle income level [18]. Interest in the educational level may be the reason for this achievement, in addition to many other factors.

As we can see in table 1.6, the education rankings sorted by population for years 2020 and 2021, China was ranked 24th in 2020 and advanced to 22nd place in 2021, and India was ranked 34th in 2020 and advanced to 32nd place in 2021. [1]

Country	Rank (2021)	Rank (2020)	2022 Population
China	22	24	1,448,471,400
India	32	34	1,406,631,776
United States	1	1	334,805,269
Indonesia	54	55	279,134,505
Brazil	36	30	215,353,593
Russia	23	21	145,805,947
Mexico	37	36	131,562,772
Japan	7	11	125,584,838
Philippines	55	52	112,508,994
Egypt	39	42	106,156,692
Vietnam	59	64	98,953,541
Turkey	31	31	85,561,976
Germany	3	4	83,883,596
Thailand	46	48	70,078,203
United Kingdom	2	2	68,497,907
France	5	5	65,584,518
South Africa	33	32	60,756,135
Italy	14	16	60,262,770
Kenya	68	68	56,215,221
Myanmar	76	71	55,227,143

Table 1.6: Education rankings sorted by population 2021 [1]

Here, an important question arises: Does economic power lead to a good level of education? Or does a good education lead to prosperity and economic progress?

It is clear from what we have seen and presented from examples such as Japan, Singapore, United States of America, and Germany that education is the basis and origin of economic power. And after discussing the importance of education for society, we will now see the importance of education at the level of individuals.

#### 1.1.2 Importance of education for individuals

To discuss the impact of education on the lives of individuals, we must study the intersection between the educational domain and the individual's domain, which clearly intersect in the category of scientists. Albert Einstein and Marie Curie are considered among the most famous and prominent scientists of our time, who rose to prominence in difficult circumstances and a harsh life. Was education a role in improving the lives of these individuals?

Beginning with some benefits provided by education for individuals, according to the World Vision for Education organization [9] and Lawrence [19], education improves a person's communication skills, uses logic to solve problems, sharpens innovation skills and creativity, develops time management skills, provides basic skills to compete in the labor market to obtain a good job and improve his life. Education also promotes equality between men and women, thus preserving the basic building block of society and reducing infant mortality. According to a UNESCO report, a mother's education increases the chance of a newborn child reaching the age of five by fifty percent [9]. Thus, education is one of the reasons for escaping from death literally and at the level of every individual.

In addition, the World Bank reports that pregnancy of minor girls decreases by six percent for each additional year of schooling for girls. It also helps mothers realize the responsibility of having new children and control the number of births for each family by increasing awareness and a sense of responsibility [19], and the future income of each individual increases by ten percent with each additional year of education. [20]

Now going back to our question, about the role of education in individuals such as Einstein and Curie life, we will find the answer after exploring some facts about their lives.

Marie Curie: Marie Skodowska was born in Warsaw, Poland, on November 7, 1867. Her parents were educators who supported education because of its importance, and its role in the renaissance of nations. Because of their interest in education and their realization of its importance, the girl traveled at a young age to another country to complete her education, and the chosen country was France. There she met Pierre Curie, who was her colleague and later husband, to be her companion in personal and educational life. They decided to conduct their research in the field of radioactive elements, which was her passion. [21]



Marie Curie [22]

Indeed, in 1903, their research and efforts yielded fruitful results for both spouses, as they shared the Nobel Prize in Physics. During World War I, Marie Curie harnessed her knowledge to serve her community by making the first mobile X-ray machine to help war-wounded and sick people everywhere. Her scientific achievement did not stop at that point, but after eight years, specifically in 1911, she won her second Nobel Prize in Chemistry, maintaining her place in history as the first person to receive two Nobel Prizes. [21]

In order to emphasize the legacy that she learned from her family about the interest in education and the realization of its importance, Mary bequeathed this passion to her daughter, who continued the path of science and research, and her efforts also resulted in obtaining the Nobel Prize in Chemistry in 1935 jointly with her husband Frederic Joliot. [21] [23]

Albert Einstein: On March 14, 1879 Albert Einstein was born in Ulm, in the Kingdom of Württemberg in the German Empire in a family with difficult economic conditions. Introversion and difficulty speaking with poor memory were the characteristics of Einstein, the child whose parents thought he was mentally retarded. He was suffering from the difficulty of integrating into the strict educational system of contemporary schools, which prompted him to create problems to leave them. Electrical engineering was the specialty that his father sought Einstein to join. But the rebellious Einstein did not like the system and the method of teaching, so he clashed again with those in charge of his education. The reason he mentioned later for being rejected and rebellious against the strict traditional educational system or "rote learning" is that it kills thought and creativity in the minds of students. [24]

Despite all these difficulties and problems created by the rebellious genius child, he showed extraordinary genius in the field of mathematics and physics at a young age. At the age of twelve and during one summer course the child taught himself principles of algebra and Euclidean geometry. In the same period, he studied and understood the Pythagorean theorem and discovered his own original proof for it. When the teenage boy completed his journey to the age of fourteen, he said he had mastered integral and differential calculus. The child Einstein loved science and showed superiority and genius in it, but he preferred to study it on his own, away from the strict traditional education systems. **[24]** 



Albert Einstein [https://historia.nationalgeographic.com.es/personajes/Einstein]

At the age of sixteen, Einstein was presented with an educational resource that completely changed his life, which was a series on science for children by Aaron Bernstein, (1867–1868; Popular Books on Physical Science). What was distinctive about this educational resource was the use of imagination and non-adherence to traditional forms of education, as the author imagined riding the electric current passing through the telegraph wire. [24]

And here the genius did not stop at the limits of what he saw, but he thought beyond that, the light.

If you could run parallel to the light beam, how would it appear to you?

This question remained in control of his life and thoughts for the next ten years. At that time Einstein also wrote his first "scientific paper" ("The Investigation of the State of Aether in Magnetic Fields"). [24]

There are two remarkable things that profoundly influenced Einstein's life, according to what he wrote. The first one was about the invisible force acting on a compass needle that he saw at the age of five, which made him fond of exploring the invisible forces in nature to discover their sources and understand their rules. The second one was the book that he found about geometry at the age of twelve, which showed great passion in reading it and caring for it to the point that he called it his "sacred little geometry book". Eventually, he traced his passion and produced the most prominent assumptions and physical laws that changed the history of science and invention. **[24]** 

As we have seen in the events mentioned about the lives of Albert Einstein and Marie Curie, education was a pivotal influence in their lives. It had a huge positive impact on the course of their lives and the nucleus of their scientific achievements whose fruits we are still reaping to this day.

As another evidence of the important impact of education on the lives of individuals, researchers have studied the effect of different levels of education on identical twins where they have the same family environment, so their skills and habits should be very similar. The results of these studies confirmed this importance, showing an increase in earnings and productivity per individual by between 6% to 10% for each additional year of education. [2]

#### 1.1.3 Importance of education in the research field

After exploring needs and requirements for main elements of the learning process (learners and teachers), we explored the Scopus database for research about education and e-learning to analyze their relation. Results presented in next figures. In table 1.7 we can see the number of published documents is increasing from 1990 till 2022 for both keywords. The number of documents with the "education" keyword in 1990 was higher than with the "e-learning" keyword (2,500 vs. 0), and in (2020-2021) the number of documents in e-learning reached 12,500 while education reached 15,0000.



Next we will analyze the number of research in education and e-learning on the Scopus database grouped by country. In table 1.8 we can see that among the top 10 countries there are 7 out of 10 countries in both fields.



Table 1.8: Number of published documents by country

Next we analyzed the results according to the subject area. In **table 1.9** we found that among the top 5 subjects there are 4 out of 5 subjects that are common between "education" and "e-learning".



#### Table 1.9: Number of published documents by subject area

### 1.2 Education vs. learning

Education and learning are two different operations that could occur at the same time or separately. There is a clear difference between the two, on which we will elaborate in the next section.

#### 1.2.1 Education

Many definitions had been used for education, such as:

- "Education is a planned activity geared at accomplishing specific goals, such as imparting knowledge or developing skills and character qualities." [31]
- "The process of receiving or imparting systematic instruction, particularly at a school or university, is called education." [Definitions from Oxford Languages]
- "Education is the process where people acquire or disseminate fundamental knowledge to another." [9]
- "Education is the process of receiving or imparting structured instruction in order to gain knowledge." [32]
- "Education is a formal conservative process that aids in the development of talents, attitudes, and other behavioral practical values in the society from which they belong, for the objective of gaining social competency and maximizing the potential of an individual". [33]

However, no matter how different the definition is, education has basic goals such as: giving the individual the ability to distinguish the right thing to do, encourage it, and distinguish the wrong one to move away from it and reject it; giving individuals the ability to collect the basic and daily skills necessary for life and coexistence; spreading social norms among the members of society to coexist in peace, security and order; improving thinking, reasoning, judgment and deduction abilities; helping to build society by encouraging individuals to build, participate and cooperate, which is the ultimate goal of education. [9]

#### Types of education

There are three basic types of education: formal education, informal education, and non-formal education. [9] [33]

1- Formal education, present in academic institutions inside regular classrooms. In formal education only qualified teachers have the ability and the right to teach students the standard curriculum that contains basic skills or even advanced academic lessons. Formal education has specific steps and levels. It starts in elementary school, and advances to post-secondary education. Among other education types, formal education is the most popular type because of its effectiveness and its ability to develop generations' capabilities by passing education and knowledge from one generation to the next in a systematic, repeated and stereotyped manner. [9] [33]

Education of this type can be obtained or delivered through two methods: face-to-face, where there is direct communication between teachers and students at the same time and place; or distance education, where contact between the members of the educational system is indirect, in terms of time, place, or both. Different means of communication methods can be used based on needs and available resources. With the different means of communication and types of educational resources, the sub-types of distance education differ.

One of the sub-types of distance education was education by correspondence, where the educational content was prepared by the educational institution and sent to the student, then the student studied and his academic progress was followed up through assignments. Another type of distance education was teleconferences, where communication was taking place between teachers and students at the same time, but across distant places through satellites and other available technological means of communication. This type was very similar to video conferencing that used the same means and methods. Distance education using mass media was another type where education resources were disseminated by television, radio, or newspapers. With the advent of the Internet and educational websites, online education emerged as one of the new types that allow sharing educational resources on the website of the educational institution and sending them to students to study them at any time and place that suits them. By merging interactive electronic educational resources based on computers and the Internet as basic tools for communication and education, e-learning emerged. [9] [33]

- 2- Informal education, where academic institutions have no role and student do not need them in any way. Home, libraries, nature or Internet websites may be the source from which students acquire knowledge and learn skills. This type of education is not limited by any schedule, academic content, or educational path. Therefore, it is a natural form of education and can even happen involuntarily.
- 3- Non-formal education, where the organization, scheduling, and the systematic implementation are essential, but it does not have to be implemented in a school system. In terms of quality, this type can potentially provide the same level of quality as the previous two, but it presents more advantages than them in terms of flexibility and ease of implementation, and it does not set conditions or limits on the age of the students.

As a result, it is a hybrid of formal and informal education in which informal learning takes place in a formal setting.

#### 1.2.2 Learning

On the other hand, learning is a different process and many definitions had been used for it, such as:

- The process of obtaining new information, skills, actions, beliefs, attitudes, and preferences is known as learning. [34] [35]
- Learning is defined as the process by which knowledge or abilities are acquired through study, practice, or instruction. [Definitions from Oxford Languages]
- Learning is the sustained modification of behavior brought about by practice or experience. [33]
- Learning is the process of acquiring information, attitudes, skills, beliefs, or preferences that aids an individual in adapting to their environment. [32]

With the passage of time and the change of age stages, the nature of people changes, with growth or puberty, which affects individuals biologically and behaviorally, and this change may be positive or negative. Here, the effect of learning appears, which also contributes to the change, but behavioral change is not counted as a learning outcome unless the change is permanent and continuous with the person. Learning is structured around a unified goal and purposeful centered point, but if the learning process is done effectively and successfully, it can be used to serve multiple goals and harness it to serve different situations. [33]

People can learn from their surroundings, continuously through their whole lives, such as continuous or coincidental life experiences, education of various types and from different sources, school life, training and self-development. All of these things can be the reason for learning, in addition to many other sources that are not limited to a specific period or a specific age. It is interesting that the gift of learning is not exclusive to humans. Animals and plants also have the ability to learn, and now machines can also learn if the required conditions and resources are provided. And with the realization of the importance of learning for any living organism, the question arises: what is the motivation for learning, why is learning so urgent that some animals seek it? And we find the answer lies simply in two words, curiosity and motivation. Most of the time, humans or animals have a curiosity for knowledge and understanding, which drives them to watch, observe, experiment and draw conclusions. On the other hand, motivation could be a main driver for learning, where the pursuit to obtain a specific pleasure, solve a problem, or overcome a difficulty, could constitute an incentive for learning. [32]

#### Learning types

There are six types of learning according to [33] : verbal, motor, concept serial, problem solving, and paired associate learning.

- 1. In verbal learning, signs, symbols, figures, words and pictures are used to help in acquiring verbal behavior.
- 2. In motor learning, instruments are used to help learning a task such as swimming, driving, riding, flying, playing.
- 3. In concept learning, our primary tool for conjuring an image of things, people, or events in our minds is imagination.
- 4. When learning to solve problems, our cognitive skills like reasoning, comprehension, observation, generalization, and inference are our instruments for tackling a particular issue in a particular circumstance.
- 5. In serial learning, school education provides students with the educational material that they have to follow in the same sequence to learn the skills they need.
- 6. In paired associate learning, the learner receives learning tasks on account of his association to achieve some learning goals.

#### Learning styles

Moreover, in [36] Felder and Soloman stated that there are four dimensions of learning styles where each dimension have two opposite categories as follows:

#### 1- Active and Reflective learners:

Active learners understand information better by practicing it actively with some of the active practices such as discussing with other learners, explaining it to other students, or applying the information in use in the field of study.

On the other hand, reflective learners tend to take enough time to think and reflect on the content to understand and comprehend it. [36]

#### 2- Sensing and Intuitive learners:

Sensing learners want to learn from scientific facts without resorting to surprises and complications. They want to solve problems with proven scientific methods that have been tried over and over again. They are very upset about being tested and questioned in parts of the course that are not covered and explained directly and extensively in the classroom. Their skills appear in good memorization of scientific content, patience in studying complex details, and skillful manual experiments. Therefore, they are more practical and have greater care and caution in laboratory work and they have difficulty studying and understanding courses that are not directly related to practical life.

By contrast, intuitive learners love innovation and conclusion, get bored, and even flee from repetitive and routine education. They have the drive and enthusiasm to search for the relationships between things and their interactions and to explore their possibilities and capabilities by themselves. They do not prefer the content that is explained extensively, boring and monotonous, and that is presented with a spoon, which contains a lot of texts that need to be remembered without understanding and comprehension. They can absorb new concepts and definitions more easily than sensing learners. Mathematical formulations are not an obstacle for them either. They prefer to work faster and more creatively than sensing learners because they are not bound by the steps and routine traditional methods, but therefore, their care and caution are less than their counterparts. And they can accept with patience the tests in the parts that have not been explained extensively because they are not bound by the explained content, and they always strive to conclude and expand perceptions. [36]

#### 3- Visual and Verbal learners:

Visual learners are the learners who absorb more and faster from visual educational content such as pictures, diagrams, flow charts, time lines, films, and demonstrations.

On the other hand, verbal learners learn better from verbal educational content such as written texts or spoken lectures. Unfortunately, the majority of learners need visual content to obtain the largest percentage of the content, so that cognitive achievement is very little if only verbal content is used. Knowing that, by increasing the diversity of the types of scientific content provided to the learner, the percentage of cognitive achievement increases significantly. So, the good learner is the one who has the ability to comprehend and understand all types of educational content in an equal manner, whether the available content is verbal content or visual content. [36]

#### 4- Sequential and Global learners:

Sequential learners prefer to follow sequential steps in a logical, systematic and organized way, so that each step is a logical conclusion to the step that preceded it in order to solve the problem in question and study. In general, their approach is to follow clear and sequential logical steps to reach solutions.

On the other hand, global learners prefer to learn in a random and non-sequential way to reach solutions without following any organized educational path and with steps in the form of large leaps between different methods and paths. What is important for them is to comprehend the content, even if it is not organized, and if they do not realize the relationship between the non-sequential contents. One of the strengths of learners using this style is that they may find solutions quickly, using innovative and unprecedented ways to do so, only when they can understand the big picture. But in many cases it is very difficult for them to explain how they reached this solution or to show it to other students to follow the same method that they used. [36]

#### **1.2.3 Learning theories**

To complete the whole picture about learning and education we must go through a quick review on learning theories. In [37] Phillips et al. clearly stated Plato's theory, which assumes that every child born has their own knowledge accompanying them from birth. It is called the Theory of Recollection or Platonic epistemology.

Conversely, Brain et al. discussed in [38] the Locke's theory that assumes that humans are born without any innate knowledge and without any prior experiences, but they are born as a blank white page that is filled with experiences and knowledge from the environment surrounding them. There are many other learning theories such as Behaviorism [37], Social learning theory [39], Cognitivism or Gestalt theory [40], Constructivism [41] or Transformative learning theory [42].

Our system will be based on the Learning as a Network learning theory [43], where learners are the basic nucleus of the learning process, and the learning process is followed up by the continuous building of network links between them and their surrounding world, which represent their experiences and personal knowledge gained during their learning journey. The Learning as a Network theory builds upon the connectivism theory [44] [45], which represents networked learning [46], and focuses on learning as making connections. We have chosen it because it is considered the most appropriate theory for the current conditions and for the tools that we will study and use in our thesis, such as the interactive knowledge graph [47] [48] that we are going to discuss in chapter 2.

#### 1.2.4 Learning vs. Education

Now that we are fully aware of what learning and education are, we give a comprehensive list of their distinctions in table 1.10 as mentioned in **[33] [32]**:

	EDUCATION	LEARNING	
Definition	Received or given systematic	Learning is the process by which	
	instruction, particularly in a formal	knowledge or abilities are obtained	
	context, is what is referred to as	through study, practice, or	
	education.	instruction.	
Purpose	To alter the student's behavior in a	To create new knowledge and to	
	way that is more desirable.	manage future behaviors.	
Knowledge	Via a teacher or a textbook, one	Through a variety of sources,	
_	gains knowledge in education.	knowledge is acquired during	
		learning.	

 Table 1.10: Differences between Education and Learning [33] [32]

Guidance	Under the direction of an instructor,	No teacher or mentor is necessary	
	education frequently occurs.	for learning.	
Motivation	Extrinsic incentive drives education.	Intrinsic motivation drives learning.	
Elicited by	Extrinsic Motivation	Intrinsic Motivation	
Process	Education is a process that is	Learning is not a methodical	
	organized.	procedure.	
Age	Up until a particular age, education	Age is not a factor in learning.	
	is obtained.		
Guidance	Required	May or may not be required	

#### Education from the learner's point of view

With different ages, generations, and available technology, the learner's needs vary. To understand the learners' needs and requirements, the educational system must be viewed from their point of view. In principle, each learner may have their own personal preferences about the educational system, and it is also important to be commensurate with the content they want to study. The current generation, as a result of the contemporary portable and easy-to-use means of communication, does not have the ability to be bound by what is being studied, whether in space or time. They prefer to access educational material anywhere any time.

In addition to their rebellion against the traditional educational systems, they need new systems that are interesting, not boring and capable of stimulating the student's mind and developing their capabilities in an innovative way. But they want all these features at the lowest cost.

#### Education from the teacher's point of view

In order to face all the current challenges and provide new requirements for learners, educational systems must also be studied from the point of view of teachers. Teachers are looking for an educational system capable of managing information more efficiently, effectively and quickly, gain the ability to communicate with a large number of students quickly, provided with the appropriate tools to explain different educational contents in different ways that suit the needs of different students, supported by the appropriate technologies for monitoring the educational level of learners individually, and to test students in appropriate ways to measure their different abilities with each educational content.

#### "Passion" is the key to excellence

None of the above will be possible if there is no passion, either from students or from teachers, because forcing someone to learn anything they do not like is a wasted effort. If students love what they learn and have their own motivation and passion to learn, this will be much easier and fruitful. They could even help each other and cooperate to achieve better results. Persons will be interested in learning when they are prompted by curiosity and intrinsic motivation. If they are well motivated they could find their passion that might guide them towards improvement. Accordingly, the educational system - that should start by looking for the areas of passion of each child - should be different for each person according to their own interests.

#### 1.2.5 E-learning

Depending on how and where e-learning is being implemented, it can be defined in different ways. According to the European Commission [49], It is "the technologies that enhance learning quality by facilitating distant exchanges and cooperation, access to facilities and resources, and both"

Other definitions of e-learning are mentioned in table 1.11 [50].

	8	
1	In a well-designed course with acceptable	(Choudhury &
	accreditations, e-learning is defined as the transfer of	Pattnaik, 2020)
	knowledge and skills via electronic media like the	
	Internet, intranets, and extranets.	
2	A web-based framework called e-learning employs	(Rodrigues et al.,
	digital technology and other forms of educational	2018)
	content to give learners a personalized, student-	
	centered, easily available, enjoyable, and interactive	
	learning experience that supports and improves	
	learning processes.	
3	In the current technological environment, e-learning	(Vershitskaya et
	is a crucial instrument for addressing the demand for	al., 2020)
	highly skilled workers.	·

#### Table 1.11: Definitions of e-learning

Since different terminologies vary depending on the researcher's area of expertise and area of interest, there are different terminologies used to define e-learning. Online education is also known as e-learning, mobile learning, cloud learning, massive open online courses (MOOCs), virtual learning, distance learning, online interactive learning, virtual learning, and web-based education [49] [51] [52]. And, according to [53], since it is a user-centric framework that concerns material and how it is interpreted, e-learning varies from general Information Systems (IS) in this regard.

#### E-learning from different Perspectives

Instructors are crucial in ensuring that technology is successfully incorporated into teaching and learning in terms of implementation and curriculum preparation. And as stated by Kamla et al. [54] teacher satisfaction has a significant impact on how effective e-learning is. [49]

On the other hand, students are a key element in e-learning system success because they interact directly and continuously with it and without their participation e-learning systems would not be able to develop the learning process [55]. But the most relevant hurdles and difficulties a faculty has while deploying an emerging innovation, however, are reluctance to change and a strong refusal to learn [56]. Several research findings showed that the instructors' top concern throughout the evolution of e-learning has been a lack of student motivation [57] [49].

Current e-learning systems, with their current capabilities, are required to be evolved and make use of new technologies to provide better performance for the growing number of users and the evolving learning materials. Like every research branch, e-learning starts as a simple idea and is developed over time thanks to the growing number of interested researchers, the emergence of new technologies that can improve the existing systems, the increasing pressure on e-learning systems, and the increased financial support directed at developing technology for economic and commercial purposes.

By integrating different technologies into the e-learning system we can harvest the advantages of all of them. E-learning is a good choice because it can be used by educational systems as an affordable, safe, and progressing alternative to classical education. It is available anytime and anywhere with the lowest costs. Although e-learning is not new, the current state of affairs requires more capabilities from e-learning systems.

To select the required technologies we first have to explore the evolution of the e-learning ecosystem to find their pros and cons from different perspectives (teacher and student), analyze

the available technologies and their integration with the e-learning system. In addition, we could investigate learning theories and information system theories.

E-learning is also a great facilitator of self-learning, as it helps learners to autonomously get appropriate materials for the skills they want to develop, without the need to take full courses at formal education institutions such as schools and colleges. Self-learning has the potential to change the way people are going to be educated. As every student is unique, the education system has the opportunity to adopt new tools that help it to adapt uniquely, according to every individual student's needs. As reported in [58], 91% of elementary school students and 71% of middle and high school students use e-learning for learning things on their own.

So far, we can note that students in the current era are lucky to have all these possibilities, tools and educational resources at their fingertips. Moreover, several studies showed that the number of available learning resources is growing every day [59] [60].

#### 1.2.6 Evolution of distance learning and e-learning

Education and the development of its means have grabbed the attention of mankind a long time ago. With the emergence of any new means of communication, humans try to adapt and exploit it in developing and improving contemporary educational means. This happened with every development throughout history.

Distance learning development started in 1728 by regular mail, where the first weekly maildelivered distance learning course was developed by Bostonian Caleb Philips. In 1840 Issac Pitman used letters to teach his students shorthand. Later, in 1922, courses began to be broadcasted over radio stations. The first testing equipment was developed in 1924. It was automated to allow students to test themselves. In 1953, educational courses were on TV channels. And the teaching device GLIDER, created by Harvard professor BF Skinner in 1954, allowed schools to provide students with pre-recorded introductions.

In 1960, Prof. Don Bitzer of Illinois University presented the first computer-based training (CBT) program, called PLATO. It was a big heavy machine that could be used by one student at a time. After five years, in 1965, Phone-based courses were introduced. In 1966 a psychology professor from Stanford University created computer aided instruction.

In 1969, the ARPANET, which was the main core of the Internet, had been created by interconnecting four university computers. The Apple II personal computer was released in 1977. Then, in 1984 Toronto University offered the first-ever completely online course.

Finally, the World Wide Web -that was invented by Tim Berners-Lee- was born in 1989. The Phoenix University won the educational organization race by launching the first all online collegiate school, while Drexel University started one of the first online programs, and the Open University in UK was among the initial universities in the world to provide online distance learning, and the first online university, the Jones International University, opened in 1999 until it closed in 2015.

In the 2000s, businesses realized that they had a lot of content they could share with employees, so they adopted e-learning. 81% of colleges in Drexel University offered at least one course via the Internet in 2003, where more than a quarter of US college students were taking courses online [61]. In 2012, 6.7 million (32.5%) of the 20.6 million students enrolled in higher education, were taking an online course, according to the fall 2012 survey of top academic leaders [62] [63]

[64]. And in 2013, one-third of all college pupils were pursuing at least one course online. In 2016 Drexel University celebrated 20 years of online courses [65].

Over 75% (11.8 million) of all undergraduate students in the US were enrolled in at least one distant education course as of fall 2020. [66]

In 2019, the market size of online learning was approximately \$187.87 billion having a 400% increase over 2013 with the use of virtual reality and game-based learning. Experts predict the next wave of online education will happen in emerging markets (Africa, India and China).

### 1.3 Objective

Learning resources are an important element of e-learning. Learning resources are available in a wide range of formats that cover almost anything that one could need to learn [67] [47]. However, the growing number of learning resources is a double-edged sword: it will provide a great opportunity for students to learn anything they wish in different styles and ways of exploration. But it also represents a challenge for students to find and select the required one among this growing ocean of resources. Therefore, learners would benefit from having new tools and techniques for exploring, browsing and searching learning resources. Those tools would not only save learners time and effort, but also protect them from boredom and the massive progress of technologies such as Artificial Intelligence [68], Machine Learning [69], Natural Language Processing [70], Semantic Web [71] and Virtual Reality [72] could facilitate building such tools.

In this thesis, we aim to design an e-learning system to help learners and researchers on electronic educational resources to obtain the required knowledge in the fastest time and with the least effort. In addition to ridding the learner of boredom and monotony that learners face with current learning systems through integration with learning groups and dynamic interactive knowledge graphs. The use of current technological means may contribute significantly and effectively to the development of the old idea, which is still in use until now, about e-learning systems. In addition to presenting the design of our e-learning system, we will present our algorithm that we designed, implemented and tested to help in the same field, and we will mention that in detail in the coming chapters.

#### 1.3.1 Challenges

Many challenges could face us during our journey towards our goal, such as:

- The required infrastructure of technology, Internet connection, and computational power. The required training for students and teachers is also a big challenge especially in developed countries.
- Converting all learning materials into valid learning objects in order to be used in our elearning system.

#### **1.3.2 Contributions**

We plan to apply results in these areas to build an e-learning system, inspired by the Learning as a Network learning theory [43], which aims to facilitate the learning process for students in different ways: (1) a topic extraction system, to automatically obtain the main topics from learning resources and build a multilayer topic structure; (2) a clustering system, to connect learning resources based on the relationships between their topics; (3) dynamic interactive knowledge graphs, to show learning resources in an easy and searchable dynamic connected graph; and (4) a social learning network, to connect different users with common interests or common learning goals, and to facilitate self-learning.

Our main contribution is a topic extraction algorithm that is able to analyze the text of learning resources, including large books, and identify their main topics as unambiguous references to categories in the Wikipedia collaborative encyclopedia. Based on these extracted topics, the system is able to connect and group related learning resources. These connections can be presented to learners in the form of knowledge graphs that they can browse graphically and interactively in order to search and explore resources they might be interested in.

Automatic Topic Identification is vital in the current information era for finding learning material among this daily giant growth of data on the Web. To establish unequivocal topic identifiers we need a comprehensive and up-to-date vocabulary, knowledge source, or ontology that could be used for representing any possible topic a learning resource might be about. In our opinion, the online encyclopedia Wikipedia is the most feasible choice for this task. As such, it represents a giant multilingual database of concepts and semantic relations. A specific topic of an article can be identified with the identifier of the Wikipedia article that describes that topic

After giving an introduction about the main topic we are going to give a more focused view about the topics that we are going to use in our system in the next chapters.

### 1.4 Outline of the thesis

The rest of this thesis is organized as follows. Chapter 2 reviews related research. Chapter 3 describes the research methodology we followed and the design of our e-learning system. Chapter 4 includes a thorough description of our topic extraction algorithm and some examples of the knowledge graphs produced by the system. Chapter 5 describes the experiments we performed on the topic extraction algorithm and their results, which are discussed and compared to research work. Finally, Chapter 6 concludes the article and describes our future lines of work.
# 2. Background and State of the Art

# 2.1 E-learning

How people acquire new knowledge and abilities, as well as how those they already possess are updated to address challenges, are the key focuses of the learning process, as noted in [73], and this process could be accomplished through different learning methods.

Where learning methods are "the ways through which instructors deliver instructions and learners access these instructions" according to K. D. et al. [74], and those methods include:

- 1- Traditional learning, where the teacher meets with students at the same time and location to provide course material.
- 2- E-Learning, where the learning process is supported by technologies for communication and information globally.
- 3- Blended learning that employs a wide range of learning techniques in conjunction to increase the learner's potential.
- 4- Mobile learning, where portable technologies are used for learning or delivery of content.
- 5- Personalized learning, where individualized adaptive learning is supported and provided for each learner according to their specific needs and capabilities.

According to [75] [76], e-learning is now a modern method for promoting the learning and teaching process in universities, and it has been a deliberate concept and has been cultivated to be a significant component of education in many higher education institutions as stated in [76], [77]. Additionally in seeking to improve teaching and learning procedures, higher education institutions have also made large expenditures in the development of learning management systems [78].

However, the investments of educational institutions were not the only reason for the development of e-learning. As stated in [74], e-learning has evolved because of the development in many sectors such as network technologies, e-learning technologies and content development. Evolution of the network technologies began with client-server networks. Next, Internet-intranet-extranet were introduced, and now wireless broadband access technologies are helping to provide better e-learning management system (LMS), next multimedia and virtual communities, and game authoring tools, and now personalized tools are the current trend in e-learning technologies. Finally, the evolution of content development is presented. It started with text based content, then text based/multimedia content, after that learning objects were introduced, and now online games and personalized content are being evolved to provide more personalized and adaptive e-learning systems.

The desire for e-learning was born out of the need for the education industry to advance along with technological advancement. Nevertheless, all of a sudden, the COVID-19 pandemic struck the world, creating unmanageable declines and disruptions in many industries, including education [79]. 160 countries implemented nationwide closures. As reported by UNESCO, closures affected around 87% of higher education students [80]. It has hampered roughly 1 billion students, who ultimately depend on direct, in-person communication within these educational facilities. E-learning was the only solution for all those students in order to not lose the entire school year. Video and teleconferencing platforms such as Google Classroom, Zoom, Webex, and Microsoft had received great attention [81].

The choice was previously available, but after the pandemic e-learning was inevitable. Education is a vital need for humanity, whether there is an emergency, war or a natural disaster [82].

When we searched the SCOPUS database for "e-learning" from 1990 to 2022, we got 102,422 document results. By analyzing the results by year, we got an increasing curve as shown in figure 2.1.



Figure 2.1: number of researches on "e-learning" in SCOUPUS database during 1990-2022 [83]

As we can see in figure **2.1**, the research in e-learning is a growing research field from 2000 and till now, producing many e-learning systems, but many researchers such as **[84] [85] [86] [87] [88] [89]** state that the key difficulty in e-learning is to boost initiatives aimed at encouraging a permanent and continuous use.

In this direction, Tania et al. [90] defined the main factors for the continued use of e-learning as follows: satisfaction, perceived usefulness, perceived ease of use, attitude, self-efficacy, system quality, confirmation, service quality, information quality, flow, trust, task technology fit, social influence, performance expectancy, personal innovation, habit, and technological construct.

But what is the benefit of the continuous use of e-learning systems if they provide the same content in the same way to all students and do not take into account the differences between them?

Students are humans, not machines. They have different feelings, different back knowledge, and different learning styles. We must take that into account, if we want to improve the learning process, previous research around the adaptive e-learning concept.

### 2.1.1 Adaptive e-learning

As stated in [91], based on the most recent research, a learning environment that can recalibrate to each student's particular requirements, strengths, and capabilities speeds up the learning process and improves learning outcomes and performance.

And here we need to introduce the term "Adaptive learning", which was defined in [92] as a strategy for delivering individualized learning, with a propensity to offer capable, effective, and

custom learning pathways for capturing the attention of each learner. Kerr P. [93] presented three types of adaptivity:

- 1- Learning contents adaptivity, where the personal preferences of students are taken into account such as their preferred learning style, what students need to learn, educational background and knowledge of students, in addition to their personal skills and experience.
- 2- Adaptivity of presentation mode, where the required adaptations are applied to the method of presenting the content according to the personal preferences of the students and according to the appropriate methods for the educational content that is presented.
- 3- Complete adaptivity that combines the two earlier categories.

AHA, APeLS, and 3DE are a few examples of extant adaptive web-based educational systems [94] [95] [96] that offer different degrees of adaptivity. But despite their adaptivity, they have a low usage because of providing a meager selection of learning options, a lack of collaboration, and the inability to reuse their learning materials.

Nonetheless, despite having little to no adaptivity, LMSs like Blackboard, WebCT, or Moodle are often and effectively used in e-learning because they provide a variety of tools that help teachers manage and develop their online courses [97].

Hatami presented different learning styles in [98], but the Felder-Silverman Learning Styles Model (FSLSM) is the most used one because of its adaptivity within e-learning environments in addition to learning styles. FSLSM uses four dimensions to describe student's learning styles with a lot of details. Active/reflective, sensing/intuitive, visual/verbal, and sequential/global are the four dimensions [99], as mentioned in chapter 1. Other research identifies three learning styles, which are [74], [100]:

- 1- Visual learners: where learning is most effectively accomplished by visual means, such as visuals, demonstrations, facial expressions, and instructor body language.
- 2- Auditory learners: where listening to lectures, conversations, talking things through, and hearing what other people have to say is the best method to learn.
- 3- Tactile/Kinaesthetic learners: where the best way to learn is through experiencing, reflecting, interacting, and doing things, and exploring through simulation environments is a good choice for this learning style.

For each learner of the above three learning styles there is a preferable style, but it is better for students to try different styles interchangeably while going through the learning process to gain the best result and avoid getting bored [74]. Considering Felder and Silverman [101] opinion, students should learn through different operations: the ability to hear, perceive, reflect, and act; to reason rationally or instinctively; to memorize spoken or written materials; and the ability to recall seen images

After exploring different learning styles and operations, we are going through some research examples on adaptive e-learning.

One example on adaptive e-learning is proposed by Arsovic et al. [91], where they introduced an adaptive e-learning model that consisted of five main parts: students' model, adaptation module, expert system for data analysis and decision making, repository of learning objects, and database of educational methods. Their proposed model utilizes a combination of static and dynamic personalization techniques.

The static part uses a specific questionnaire when students first register on the course to determine each one's preferred learning style, and a pre-test to determine each student's prior

knowledge. According to the student's needs and results in previous courses, the new course's contents and structure are adapted to provide the best performance and satisfaction. On the other hand, the dynamic part is based on the continuous observation of the students' attitude in real time. Using data mining techniques on the obtained student's data provides very useful information that could be used for the adaptation and the personalization process.

Another example on adaptive e-learning includes [102] where Surjono presented many important factors that could enhance the performance of the adaptive e-learning such as student's personality, learning in social groups, active student participation, and motivation. He mentioned that, by understanding a student's personality, there will be a better identification of learning style in adaptive e-learning. Learning in groups also was one of the factors that he found useful for the learning process, where the distance between individuals is not important [103]. Another core factor in the success of e-learning is the active participation of students in the learning process, as JW You mentioned in [104]. And, finally, motivation, which is very important for students to keep while using the e-learning system. According to Al-Dujaily et al. [105] computer based learning students must be motivated because they invest their time and effort to gain more knowledge. Furthermore, Gardner [106] mentioned that students who learned with motivation and reasoning could reach a higher level.

To summarize, the explored research confirms through their results that adaptive e-learning is more efficient and useful than traditional e-learning, and it is more useful for students. But it could be more complicated to establish and operate, so that it is less widespread among educational institutions. But it is our choice to pick all the advantages of adaptive e-learning and endure the extra cost and effort, or use traditional e-learning after augmenting extra technologies to add more advantages. As e-learning helps improve the learning and the education process by overcoming the distance between teacher, student and the administrative staff, specific majors and courses, medical education, which involves in-person instruction and contact, is adversely impacted in this period [70]. So, other tools or technologies had to be integrated into current systems to overcome the educational obstacles of these practical courses, such as augmented reality.

### 2.1.2 Augmented reality (AR)

Many technologies have evolved during the last years because of the rapid development of the hardware and computational power. Augmented reality is one of those technologies that could add extra value to current e-learning systems. It could fill a huge gap where simulation and practice are required in specific courses such as medical education. It also could be used for adding motivation for a specific kind of students who prefer this kind of learning style.

As a result, many learning and educational institutions have expended a lot of time and money developing augmented reality (AR) applications as a creative and clever way to improve student learning outcomes [107] [108] [109] [110] [111] [112]. We can see that through its wide usage in primary schools [113] [114], along with its usage in higher education institutions [115] [116]. According to the Statista company, in their analysis of the forecasted spending on cutting-edge education technologies from 2018 to 2025, it is anticipated that by 2025 educational institutions would have invested 12.6 billion dollars in augmented reality technology, up from their 2018 investment of 1.6 billion dollars. [117]

One of the studies that looked at the use of augmented reality on e-learning was done by Baabdullah et al. [118], where they assessed 500 undergraduate students' experiences with augmented reality learning apps (AR.LRP) using a sample of the four main universities in Saudi

Arabia, and they found four major benefits from the usage of augmented reality learning applications:

- 1- Personal interactive benefits
- 2- Social interactive benefits
- 3- Affective benefits
- 4- Cognitive benefits

In addition, their results support the huge impact of those benefits on students' experience and performance.

In addition to the usage of augmented reality on e-learning, it also had earned growing global attention in a variety of key areas such as: healthcare [119], retailing [120], digital marketing [121], travel and tourism [122], manufacturing [123], and maintenance [124].

The enormous advantages of augmented reality on e-learning were not isolated from scientific research where many researchers such as [125], [126], [127], [128] reported that the natural interactive properties of AR give students greater possibilities for exploring the real world content, and [129] [130] [131] [132], who stated that "AR makes the learning process more enjoyable and valuable for students". Hsiao et al. [133] also confirmed that students' level of physical practice is improved as a result of their learning using augmented reality technology. Moreover, Wu et al. [134] mentioned that applications for augmented reality aid students in observing and analyzing phenomena that are challenging to observe or study in the physical realm.

# 2.2 Semantic Information

Tim Berners-Lee et al. introduced the semantic web in [71]. They introduced it as "a new form of web content that is meaningful to computers and will unleash a revolution of new possibilities". As they mentioned the semantic web was not a new web or a replacement of the current web, but it was an extension of the current web where the information in use takes extra metadata to add more meaning that enables humans and computers to work together on it. Because of that new representation of the information, the machines will have the ability to understand that information, and that will launch a new area of vital functionality. But this requires the computers to have, in addition to an enormous amount of structured information, a set of inference rules to be able to reason automatically new facts.

To enable the semantic web to come into being those two technologies were very important: eXtensible Markup Language (XML) and Resource Description Framework (RDF). XML handles the structure of the information documents, and RDF handles the meaning of this information in the shape of triples, where each triple represents (subject-verb-object) of a meaningful sentence [71].

With the spread of semantic technology, more and more semantic data were produced, and the need to infer new relations among this data was important to make the best use of the developing data. So there was a need for something that has the ability to connect concepts with well-defined and clear relations, and here the third basic component of the semantic web will solve that problem which is called ontologies. A typical ontology has a taxonomy (to define classes of objects and relations between them), and a set of inference rules [71].

Tim Berners-Lee et al. in their research stated that "If properly designed, the Semantic Web can assist the evolution of human knowledge as a whole" [71].

When we searched SCOPUS for "semantic web" during 1990-2022 we got only 60,970 document results, and by analyzing the result by year we got figure 2.2.



Figure 2.2: number of researches on "semantic web" in SCOUPUS database during 1990-2022 [135]

When searching SCOPUS for "ontology" during 1990-2022 we got 164,751 document results, by analyzing the result by year we got figure 2.3.



Figure 2.3: number of researches on "ontology" in SCOUPUS database during 1990-2022 [136]

From figures 2.2 and 2.3 we can see that the research in the semantics field is growing and branching in many other fields because of its reusability and diversity. It has been used in many different fields such as:

Commerce in [137], image segmentation [138], medical aspects [139], networks [140], security [141], cryptography [142], and many other fields.

As we saw, semantic information could be useful in different areas and it could be especially useful in the learning field to provide a better representation for learning contents, which will lead to faster searching and better recommendation. So we are going to explore this point in more detail.

# 2.3 Semantic information and e-learning

E-learning has been the focus of researchers for a long time. With every emergence of a new technology, attention was directed to thinking about how to use it to develop e-learning. Examples of this abound, from cloud computing to the Semantic Web, machine learning, artificial intelligence and virtual reality. We will talk and mention some of these examples around which the topic of our PhD thesis revolves.

In this section, we will explore applications and research related to the semantic web with elearning. By searching Scopus database for the two terms "semantic" and "e-learning" and analyzing the results by year we got 2648 documents as shown in figure 2.4.



Figure 2.4: number of researches on "semantic" and "e-learning" in SCOUPUS database during 1990-2022 [143]

We can see in figure 2.4 that the research publication in those two fields started to grow from the year 2002 and till now with around 150 new publications per year. By analyzing the results by country we found that China is the first country with more than 400 publications, and next comes the United States with around 200 publications. India, Germany, and Spain come next.

As the computational technology and Internet speed develops over time there is a great improvement in the e-learning sector. More learning resources are available for learners, higher download and upload speeds are provided, and the quality of materials increases. But with all these advantages we have the problem of finding appropriate learning resources among an ocean of learning materials. To bypass this obstacle we must improve the e-learning system with more advanced tools and technologies such as semantic technology, which have the ability to facilitate the searching ability and provide a better recommendation service, as we will explore below.

In [144] Huang et al proposed a new context-aware semantic information management system that provides a knowledge-enhanced learning model and learner personality representation. They

tried to integrate learning content, learner personality, and the learning process to produce a semantic e-learning framework with the use of XML, RDF, and Learning Object Metadata (LOM). They cited [145] to confirm that the more semantic aware the computing technology is, the more intelligent e-learning is expected to be.

Other studies related to the same topic were mentioned also, such as Nilson [146], who investigates the Semantic Web's possible effects on e-learning, and Henze et al. [147] and their presentation of a logic-based method based on RDF annotations for resource representation and reasoning, and Simic et al. [148], who discussed how the semantics of course content and student information could be presented using Semantic Web technologies. In figure 2.5 they compared a traditional e-learning system with a semantic e-learning system.



Figure 2.5: e-learning vs. semantic e-learning [144]

In figure 2.5 we can see that Huang et al. [144] added to the traditional e-learning system personal agents, semantic web, ontologies (knowledge bases), learning models, and learning activities to enhance the e-learning system and convert it into a semantic e-learning system. For future work they suggested more development of the semantic aware services, integration of Shareable Content Object Reference Model (SCROM), supporting of new learning models such as Scenario-based Learning and Problem-based Learning.

Gladun, Anatoly, et al. [149] presented a multi-agent system that is based on semantic web technologies aiming at controlling student's acquired knowledge automatically in e-learning frameworks and educational contexts. And to enhance their system they planned to improve their algorithms and use multi-agent technologies to enhance ontology analysis. They also planned to use the learning history to improve feedback between the learner and the teacher. Finally, they suggested the personalization of the learning strategy for each learner by the use of inductive inference methods. Junuz [150] presented some facts to be considered while generating metadata about learning objects to be used in a semantic e-learning environment. They discussed many learning object standards such as IEEE LOM, Dublin Core, IMS, and SCORM. Rui et al. [151] also discussed semantic web and metadata standards. They proposed a semantic e-learning system that could solve problems in traditional e-learning platforms with the use of semantic technology and ontology usage for learning content, resource inquiry and sharing.

On the same research track Dutta [152] also focused on the potential uses of semantics with elearning by describing learning materials with ontologies, mentioning that the reusability of semantic information, which can be used by both humans and machines, is a great advantage that could be helpful in the e-learning domain. They used semantic information in three ways: content to describe learning materials, context to define how the topic presented, and structure, the logical structure of the learning material. Sheeba et al. [153] discussed the application of semantic technologies in learning content and the possible development of e-learning activities. They have made sure that the ontology usage in e-learning content had a great focus on the research and the enormous development that it could provide for e-learning.

In this direction also, Garcia et al. [154] described a new tool called LOD (linked open data) for education based on semantic web technologies, with which they made an experiment with more than 30 students. The students who used the new tool had better results and performance. And they intended to provide more extended experiments for a better and more reliable platform for future work.

In their study they used: Sakai [155] LMS for offering instructional technology, as an entity recognizer, Apache Stanbol [156], which performs NLP, is utilized. It produces a list of URIs with some pertinent attributes. The tools necessary for updating RDF provided by Apache Marmotta [157] as a (RDF) triple store, and it also customizes a MySQL database to support triple persistence. And Wikipedia data is gathered and converted into RDF via DBpedia [158].

With the huge production of e-learning systems and architectures, the sharing and integration of data between these systems have been given less attention. So Masud et al. [155] presented a solution for this problem with semantic data interoperability and distributed metadata management. Their solution aims at supporting learning content exchange between different e-learning systems.

But not only semantic technology was used to improve the e-learning systems, there were other research fields that merged with e-learning to enhance and develop it such as machine learning, data mining, and social networks.

To determine the appropriate knowledge level for the student, Ghatasheh proposed in [156] a concept for a dynamic e-learning environment based on machine learning and the study of user activities. The chosen classification algorithm for estimating knowledge level was examined in this system using many experiments and the most recent e-learning trends.

In [157], Cuéllar et al. applied social network analysis and data mining to the semantic web and social networks to enhance e-learning settings. To combine data from several learning management systems and create a social network of learning systems, they created ontology. With the help of that system, they were able to gather more data regarding the social organization of teachers and students, which they used in conjunction with their ontology to enhance the learning process. The ontologies' hierarchical structure also allowed them to perform SNA (Social Network Analysis) at various levels. Finally, they recommended undergoing additional social network analysis in the e-learning space, so that they can support future work by providing more understanding of students' needs and offering more motivation methods individually and among learning groups.

Regarding the huge amount of available learning resources, there is an urgent need to improve recommendation systems. Semantic technologies can help in this regard as well. So, by utilizing intra and extra semantic linkages between learning items and learners' demands, Fraihat et al. [158] introduced a semantic recommender system for e-learning to assist learners in discovering suitable learning objects for their field and interests. Their suggested system enhances learning quality by requiring less time and effort to locate necessary learning resources.

Another overview of ontology-based recommenders for e-learning from studies published between 2005 and 2014 was presented by Tarus et al. [159]. In that work, they showed how incorporating ontologies into e-learning recommender systems may improve the caliber of recommendations.

In their discussion of the use of ontology in the recommendation process, Rahayu et al. [160] noted that ontologies were a useful addition to many other disciplines, including social science, artificial intelligence, computing technology, education, and psychology of learning. The results of their study are in favor of adding ontology-based recommendations to learning systems. Also, they discovered in their study that different artificial intelligence (AI) approaches helped semantic reasoners produce recommendations.

While some studies [161] [162] [163] utilized only semantic reasoners to generate suggestions, other studies made use of a variety of artificial intelligence approaches, including: 1- Software agents [164] [165] [166], 2- Collaborative filtering [167], 3- Cognitive diagnosis method [168], 4- Fuzzy linguistics [169], 5- Generalized sequential pattern algorithm [170], 6- Hierarchical Task Network Planning [171], 7- Natural language processing [172]. We will elaborate more in this area because of its importance for our research.

Capuano et al. in [172] unveiled a smart learning system based on approaches to knowledge discovery and cognitive computing, where several services were incorporated into their system, including the automatic conceptualization and classification of textual legal cases using natural language processing, the creation of learning paths based on legal ontologies, and the management of legal knowledge bases through editing, versioning, integration, and enrichment.. They highlight the growing prominence of semantic approaches and technologies in technology-enhanced learning, backing up their point with studies like [173] [174] [175] [176] [177] [178] [179].

They created a system, called electronic Justice Relationship Management (eJRM), that manages the resources that are currently available based on legal ontologies, enriching them with data from the thesauri EuroVoc [180], ItalGiure [181], and DeJure [182], as well as Wikipedia's common sense knowledge. Over 13,000 concepts make up the defined ontologies, which are connected to one another through both informative relationships.

The editing and versioning functions are provided by a graphical ontology management system (SEMANTO) [183] on their behalf. Also, they used the method developed by Capuano et al. [184] to automatically augment Wikipedia topics and words with the concepts on the system's ontologies for the enrichment function, which involved adding more terms and topics from outside sources. With an algorithm described in [185] for matching and integration, they matched ontological elements discovered by using a variety of similarity criteria.

For the semantic search, their eJRM system uses a knowledge discovery technique to extract pertinent concepts from user-input text, while including a portion of the algorithm described and put into practice in [186], [187].

Comparing our system against the previous one, we used natural language processing with machine learning through the Wikipedia miner tool to extract main topics from learning materials that not only related to legal ontologies, but could be any learning object in any topic.

Yu, Zhiwen, et al. [188] conducted another related study and suggested an approach for semantic content recommendation based on ontologies toward context-aware e-learning. In it, they wanted to improve learning effectiveness and make it easier to find relevant learning materials among the expanding collection of educational resources.

To generate customized and enhanced recommendation results, their suggested recommendation system incorporates knowledge about the learner's context (without naming the information

resource), knowledge about content, and knowledge about the learning domain. The semantic similarity between the learning content and the learner's aim serves as their system's ranking criterion. Figure 2.6 depicts the learner ontology, learning content ontology, and domain ontology that they created for their system.



Figure 2.6: Learner, Learning Content, and Domain Ontologies [188]

They noted that the use of ontologies gives programs the ability to interpret learner context and content aspects based on their semantics, and their process for recommending learning content consists of four steps, as illustrated in figure 2.7.



Figure 2.7: The recommendation process [188]

Their recommendation process starts by:

- Semantic relevance calculation. In this initial stage, all of their ontologies are used to determine how semantically related are the learner and the learning materials, in order to produce a list of recommendations.
- Recommendation refining occurs in the second step, when the results are adjusted using the content ontology, until a number of viable options are found.
- Learning path generation, where learner and content ontologies are both used to get the target learning content and its prerequisite content.
- Recommendation augmentation, enhancing recommendations by aggregating content related to the main course using a content ontology.

From the explored research we conclude that the implementation of semantic technologies and the integration of ontology-based recommendations into learning technologies have a great impact and development on enhancing the learning process. It facilitates the search process, enriches the learning content, interconnects different e-learning systems, removes redundancy and saves time creating learning objects that are already created, augments social networks advantages into the learning and educational process, and opens new possibilities for further development into semantic e-learning systems.

# 2.4 Cloud Computing

From nature we can take a lot of lessons, we can see groups of ants cooperating to build their huge colonies, swarms of bees cooperate to curb food and defend their habitat from the biggest invader, and flocks of wolves cooperate to catch prey that is larger than the size of their members as shown in table 2.1.



#### Table 2.1: Examples of cooperation from the nature

From this, mankind learns that cooperation and working in groups often leads to better results in less time, and we began to form our own groups, to divide tasks and to specialize to serve the general interest of society. According to Marinescu a network of entities may arrange themselves to accomplish tasks that an individual entity cannot, even if each unit may only have moderate aptitude. This is how human people have learnt to aggregate themselves and create man-made machines. [192]

With this way of thinking, mankind knew that the integration of multiple tools and resources into a single system gives better results and can solve more complex problems, and it was logical to apply this principle in the computing field with the development of computing power that helped to solve more complex challenges. Over time, multiple examples of the integration of many computers appeared to work on solving one problem or to implement one goal. For example, time sharing servers, grid computing, global computing, Internet computing, parallel computing, distributed systems [192], and many other examples and paradigms that evolved due to the need for more computing power. Cloud computing was, among them, a strong competitor and a smart choice. [193]

In the following ways, cloud computing differs from other computing paradigms: With its lightweight client software, location independence, and ease of access, the cloud interface does not need users to alter their working practices or settings.

It ensures user satisfaction, hardware performance, CPU speed, I/O bandwidth, and memory capacity. The cloud's management is transparent to users and it is an independent system. It is a pay-as-you-go on-demand solution that offers flexibility and scalability [193]. Application designers are liberated from the constraints of a single system by its elasticity and impression of endless computing resources [192].

With all these advantages, cloud computing became usable in many application scenarios, which made it a tool for business promotion by lots of companies [194]. As a result, it encouraged investing in software, networking, storage, and processing technologies. It also helped focusing on corporate computing, whose adoption by commercial groups might have a significant impact on the economy [192].

Given its wide range of uses, cloud computing has been given many alternative definitions. It is defined by the National Institute of Standards and Technology as a strategy for providing easy, on-demand access to a shared pool of reconfigurable computing resources, such as networks, servers, storage, applications, and services, which may be quickly supplied and released with little administration work or service provider involvement [195], [196].

By searching Scopus database for the term "Cloud computing" and analyzing the results by year we got 124,477 document results through the years 1990 till 2022 as shown in figure 2.8.



Figure 2.8: number of researches on "Cloud computing" in SCOUPUS database during 1990-2022 [197]

Additionally, searching "cloud computing", "Internet of things", and "Fog computing" on Google trends we got the following chart as shown in figure 2.9.



Figure 2.9: "cloud computing", "Internet of things", "fog computing" on Google trends [198]

According to [196] [193] [194] and [192], there are three service models for cloud computing. The following is how software as a service, platform as a service, and infrastructure as a service are defined:

- Software as a Service (SaaS) refers to the hosting and online delivery of software to customers. It enables the use of service provided apps in a cloud architecture. Examples include the Application Service Provider (ASP), SalesForce.com, Google Mail, and Google Docs.
- Platform as a Service (PaaS) is the hosting and provision of a development platform to support the entire software lifecycle and allow users to create cloud services and apps. It provides the ability to deploy user-created apps using the provider-supported programming languages and tools. Examples include Google App Engine (GAE), [194], and Zimki [199].
- Infrastructure as a Service (IaaS) is the hosting and provisioning of processing, storage, networks, and other essential computing resources. Examples include Amazon Web Service<sup>™</sup> (AWS) [200], and Microsoft Azure [201], [194].

Other categories could be branched from the previous main three such as:

- Storage as a service (Storage aaS): hosting and delivery of virtualized storage on demand. Examples include Amazon S3, Google BigTable, and Apache HBase.
- Data as a Service (DaaS): hosting and provision of data in various formats and from numerous sources. Examples include Google Docs, Adobe Buzzword, ElasticDrive.
- Hardware as a Service (HaaS): This refers to the hosting and provision of IT hardware or perhaps a whole data center as a pay-per-use subscription service. Examples include Amazon EC2, IBM's Blue Cloud project, Nimbus, Eucalyptus and Enomalism.

Figures 2.10 and 2.11 show the relation between the different cloud services and the differences between them.



After presenting the different service models of cloud computing we are going to explore the deployment models.

According to [196] and [192], there are four types of cloud computing deployment models.

- Private cloud: This type of cloud infrastructure is only used by one company, and either the organization itself or a third party manages it.

- Community cloud: This is a shared cloud architecture that includes shared policies, needs, values, and concerns between multiple companies.
- Public cloud: A cloud infrastructure that is made available to the general public, and whose entire ownership is held by the cloud service provider, who also controls its policies, value, profits, costs, and pricing models. Examples include Amazon EC2, S3, Google AppEngine, and Force.com.
- Hybrid cloud: This type of cloud infrastructure is created by combining two or more distinct clouds. However, proprietary technology that links different clouds allows for the portability of data and applications. Organizations utilize the hybrid cloud approach because it allows them to oversee corporate operations on the cloud without sacrificing private activities or resource optimization.

Cloud computing, as we saw, has many details and properties, and we just scratched its surface in the last paragraphs, but it won't be possible to see cloud computing without other technologies. Those technologies were the base or the leader that led us to cloud computing.

According to Wang et al. [193] some of these technologies are:

- Virtualization technology that enables hardware partitioning, thus providing flexible and scalable computing platforms.
- Service flow and workflow orchestration enables computing clouds to automatically orchestrate services from many sources and types, establishing a dynamic workflow.
- Web service that presents the main idea of cloud computing as a web service.
- Service oriented architecture, that enables the management, organization, and orchestration of the different services provided by cloud computing.
- Web 2.0, which improves user collaboration and information exchange, so using cloud computing completes the necessary tools for this collaboration.
- The worldwide distributed storage system where network storage systems and distributed data systems allow cloud computing to administer its storage services globally.

And there are still more contributions coming from various technological fields, so it's not the end.

According to Qian et al. [194], who enumerated some of the benefits of cloud computing, it's a win-win situation for the service provider and the customer, and the benefits they mentioned are:

- The satisfaction of business requirements on demand,
- The reduction of costs and energy-saving,
- The improvement of resource management efficiency.

However, there are still several important issues that need to be researched, such as privacy, security, and service continuity; these issues could result in Internet problems, power outages, service interruptions, and system bugs.

#### 2.4.1 Cloud computing and e-learning

E-learning on its own proved to be vital especially during the COVID-19 pandemic. Many universities use it and many countries depend on it for improving their educational level. While e-learning has the power of connecting teacher and student anywhere and anytime, this advantage is getting weaker because of the gigantic learning content size. Learning content evolves in many types and many formats, so the e-learning systems have to get better to handle this advance; or else, it won't be useful for students and we will lose all its advantages.

To keep up with the current technology development, current e-learning systems must be enhanced with the required computational power, which vary from system to another depending on many factors such as number of users, number of available courses, and provided services. This variation and scalability could be provided easily through the use of cloud computing. Cloud computing could make e-learning advantages more powerful again, especially with rapid development in computing power [203]. Therefore, the use of cloud based e-learning was logical and inevitable, especially from a practical point of view: it is more economic and scalable [90]. As a result, a lot of educational institutions use cloud based e-learning platforms to harvest all these benefits [204] [205] [206] [207] [208].

By searching Scopus database for the term (cloud based learning) and analyzing the results by year we got 22,326 document results through the years 1990 till 2022 as shown in figure 2.12.



Documents by year

Figure 2.12: number of researches on "cloud based learning" in SCOUPUS database during 1990-2022 [209]

As we can see in the previous figure, there is an increasing research activity in this field. This is due to the high importance of developing education and the growing necessity for providing it for a higher number of students over a wider geographical space, may be to ease the learning process, or it could be to reduce the learning costs, or even to continue the learning process in case of COVID-19 quarantine [**210**] [**211**].

Developing education and learning systems requires a high budget due to the cost of infrastructure, computers, servers, and maintenance. Hence, cloud computing is the ideal choice for academic institutions and enterprises. [212]

Through the Internet, third parties offer cloud services to academic institutions. Cloud-based learning platforms demand, however, a fast and dependable Internet connection.

There are several companies that offer services, including Google, Amazon, Microsoft, Yahoo, and others, that could assist educational systems [213] [214].

A tremendous variety of e-learning tools and platforms were developed as a result of the enormous market for cloud-based learning. In their analysis of various Cloud-Based technologies, Siddiqui et al. [212] list the following:

- Easy generator [215] [216], a comprehensive cloud-based learning platform that has a straightforward tool for building interesting courses. No coding experience is necessary. The user interface is quite inviting, and courses are responsive to mobile devices.
- Lectora Online [215] [217] [218], a cloud-based authoring tool that enables the creation of assignments, the direct inclusion of expert animated movies into media libraries, and quick, unrestricted resource downloads.
- eCoach [215] [219] [220], an authoring tool that works across multiple devices and enables anyone to make online courses. It makes it simple and quick to create online courses using materials and media from all around the Internet. It offers a private URL for sharing learning materials, including mobile-friendly courses and resources.
- Ruzuku [215] [221] [222], an authoring tool that enables creating and sharing online courses by allowing users to add content, files, and activities. When used as a learning repository, Ruzuku does not track student data.
- iSpring [223] [224], a user-friendly cloud-based learning management application that offers complete control over learning solutions to organize training and learning resources.

Other examples from [215] include:

- Litmos [225], a straightforward, user-friendly cloud-based solution for co-authoring courses. A training courses library that is available for uploading can be used in conjunction with audio, video, text, graphics, and animations to quickly and easily develop courses.
- Smart Sparrow [226], a platform that enables the development, delivery, analysis, and sharing of adaptive classes based on students' past behavior, knowledge, and experiences. Also, it can adapt to any type of learning activity thanks to its dynamic adjustment to user feedback.
- Coassemble [227], a multi-device creation tool that enables anyone to swiftly create new online courses from the ground up by using content and resources from the Internet.
- Elucidat [228], a fully cloud-based e-learning authoring tool that enables writers to construct extremely engaging HTML5 e-learning for any device in addition to highly engaging courses with additional features like social polls, rules, branching, and tailored content clips.

Learning management systems (LMSs) also have a great impact on developing and managing the learning process, as they provide the required administration services and tools for both institutions and students. There are many LMSs but the most popular systems as Imed Bouchrika mentioned in [229] are:

- Google Classroom [230], a component of Google Apps for Education, that represents a good LMS for schools that are heavily involved with Google services. It only works in academic environments, though. Together with Google Classroom, Google offers a number of other applications as part of its G Suite services to support education.
- Google for education [231], a Google service that offers separately customizable versions of a number of Google products, including Gmail, Drive, Hangouts, Sheets, Meet, Google Calendar, Groups, Docs, Slides, News, Play, Sites, and Vault.
- Blackboard Learn [232], the learning management system for assessment and content reporting/analytics that is appropriate for K–12 schools, colleges, government and military initiatives, and corporations as well. It can be used in software as a service, managed hosting, or self-hosting implementations. Also, it works with portable electronics. [229]

- Canvas [233], the innovative cloud-based learning management system for education for simple learning and productivity. Given that it was built especially for K–5 to higher education institutions, with a hybrid or totally online learning environment, it could be used by schools of all shapes and sizes, from small classrooms to big institutions. [229]
- Moodle [234], an effective open-source learning management system for education in terms of flexibility. For the majority of students in secondary education, it is also a cost-effective and modular learning management system choice. It includes robust, secure, and safe tools that may be used to develop flexible training programs. Both the academic and commercial sectors are served with expanded features. [229]

Other LMS examples include Mindflash, Edmodo, Quizlet, Schoology, NEO, and many more; but one of them represents a new trend in the education field, which is a game-based learning system named D2L Brightspace [235]. D2L Brightspace is the LMS for video assignments and game-based learning. The dynamic platforms, solutions and assistance users require to have the optimum learning experience are combined into it. Additionally, 15% of enrollments and 13% of institutes currently utilize this LMS. [229]

Returning to the main topic of this part of the thesis, which is e-learning systems based on cloud computing, as it became a business field and a good attractive for investment so many big companies such as Google starts to invest in it, Amazon as well took a part of the market through providing its Amazon Web Service for education.

Amazon Web Services (AWS) cloud computing for education [236] aids in making education accessible, personalized, and lifelong for everyone. With its help, more than 14,000 educational institutions of all sizes, from elementary and secondary schools to colleges and universities, are able to fulfill their essential missions and achieve their most important institutional priorities.

Some live examples from higher education institutions and some schools could be found using some of the previous platforms and infrastructures to support cloud based learning such as:

- Arizona State University employed AWS to enhance the educational experience for students and offer 24/7, anywhere access to applications. It offered customers extended reality environments, digital assistants, speech technology, and virtual computer labs.
- The University of Notre Dame modernized administrative processes and student information systems using AWS. They were given scalable resources and capacity as needed, which helped them save money thanks to pay-as-you-go pricing and strict security standards. Also, they reduced downtime and data loss through quick, dependable disaster recovery.
- New Jersey school district and Fife School districts are another example. In addition to
  raising security standards with dependable, scalable, and secure data storage, they leveraged
  Amazon to reduce the costs of IT infrastructure by 50–75%.

Other educational institutions used Google's educational solutions [237] such as:

- University Carlos III de Madrid (UC3M), which uses Google services in addition to Moodle and Blackboard collaborate [238] to provide students and professors with the required infrastructure and services for online courses, assignments, online meetings, sharing and modifying common files, and many other operations.
- Brown University uses virtual reality to immerse students in American history. And according to Adam Blumenthal Virtual Reality Professor and Artist-in-Residence, Brown University, VR is very important for students to open their mind.
- Google for Education is another resource used by The Carroll School to assist pupils who have learning impairments. And according to Margaret Kuzmicz, Director of Technology, Carroll School, Google Workspace helps students interact with their teacher and fellow

classmates on a regular basis and frequently in real time. With assisting people in overcoming challenges they may not have previously received assistance with, this connection makes them more productive.

Now we can explore some research articles that present cloud based learning systems to get an overview about how it works.

In [239] an e-learning ecosystem based on cloud computing was presented by Dong et al. They combined cloud computing and e-learning because they believe that present e-learning systems do not have the necessary underlying infrastructures, and that combining the two will result in the dynamic allocation of the computation and storage resources that are needed. According to them, the main contributions included offering infrastructures with quality of service guarantees, service level agreements, resource allocation that was focused on priorities, support for a variety of applications, and automatic real-time configuration and resource management.

Their system is composed of three layers:

- Infrastructure layer, which provides processing and storage resources for upper levels, where cloud computing platforms are in charge of managing them.
- Content layer, which includes the instructional materials. For upper levels, it exposes the common APIs and interfaces of the contents.
- Application layer, which houses the e-learning tools, services, and other components offers features and user interfaces for other programs or users.

In addition to the three layers, it also has four ad hoc modules:

- The monitoring module monitors the requests that are processed, the real-time configuration data, and resource usage, including the state of the CPU, RAM, I/O, and other components.
- The scheduling of resources, the run-time, and the teaching and learning methodologies are all established and maintained by the policy module.
- Certain policies in the arbitration module are created manually by experts, and user requests are fulfilled.
- Finally, the provision module, which begins implementing resource allocation strategies decided upon by the policy module and arbitration module.

Their system architecture is shown in figure 2.13.



Figure 2.13: Cloud computing infrastructure-based e-learning ecosystem architecture [239]

Another example, presented in [206], is a cloud-based reflective learning environment to assist instructors and students in enhancing their capacity for reflection, both during and following actual class sessions.

Figure 2.14 depicts the three main components of the framework for the cloud-based reflective learning environment:

- The cloud service, which includes:
  - Course website, which provides tools for educators and students to use in teaching and learning.
  - Web applications, which aid educators and learners in giving and carrying out reflective learning exercises.
  - Course database, a repository for materials and course data.
  - User profile database, which keeps track of the learning profiles, learning records, and individual student information.
  - Course administration, which outlines the administrative procedures.
  - Reflection participation, which provides comments and information about how students have interacted with recent and earlier courses.
- The teacher side, which permits instructors to provide course materials and carry out reflective learning exercises to aid in student learning reflection.
- The student side, which allows pupils to engage in different reflective learning activities.



Figure 2.14: the framework of the cloud-based reflective learning environment [206]

Due to a range of evaluation techniques, such as tests, questionnaires, and interviews, their learning environment is able to successfully support student reflective capacities and improve their interest in learning. Furthermore, it enables learners and instructors to manage and conduct their learning journey.

To conclude, we can see that there are a huge number of platforms and services that support cloud based learning, not only to help develop education and society, but also because it is a good investment. None of the above systems is perfect: every one of them has advantages and disadvantages and the user or the institution have to choose the most adequate ones for their needs. But here comes our role to develop the e-learning ecosystem by researching the current problems and the missing needs of students, teachers and educational institutions to help take a new step in developing and improving the educational process. We will draw on the past experiences of researchers and keep pace with the current development in technology to harness the service of the educational process.

# 2.5 Learning object recommendation

Owing to the massive and expanding number of educational resources, finding the required one is a very hard task to do. Because of that, there is a great interest in researching this area to find a solution.

By searching the Scopus database for the terms "Learning AND object AND recommendation" and analyzing the results by year we got 1,563 document results through the years 1990 till 2022 as shown in figure 2.15.



Figure 2.15: number of researches on "Learning object recommendation" in SCOUPUS database during 1990-2022 [240]

To begin our exploration of the different types of recommendation systems we will start by the general structure of the recommendation process as described by Nafea et al. [241] in figure 2.16

The process starts at the LMS interface, where the student completes a questionnaire or a behavior pattern from previous interactions is captured by the system. The learner profile and the learning object model created from the previous step is fed into the algorithm, which applies one or more of the techniques that include data mining, machine learning, and artificial intelligence. In addition to the algorithm there are many adaptation rules that help in the adaptation process. The adaptation outcome created from all previous elements is fed into the recommendation module, which contains the learning objects list and the learner profiles list. The recommendation module matches both lists and uses the adaptation outcomes to generate the recommendation result.



Figure 2.16: The general e-learning recommendation process. [241]

#### 2.5.1 Types of recommender systems

Now we will explore different types of recommender systems. Four main types of them are identified in [241], [242], [159]:

1- Collaborative filtering, where recommendations are generated based on the similarity of user's opinions, ratings and behavior on a set of learning objects [241]. Collaborative filtering also uses the learning objects that were liked by similar users to be recommended for the current one [243]. It requires a full record of user information in their profile such as age, country, previous learning history, background. With this information, the recommender system can recommend similar students the same learning objects [244] [245]. Two types of algorithms could be used within collaborative filtering; memory-based and model-based. Memory-based algorithms use an entire database to extract information, which makes it slow, so that model-based algorithms are preferred, as they are faster because of the selection of specific items to extract [242]. According to [246] [247] [245] [248] correlation-based, cosine-based similarities and the k-Nearest Neighbors are the most popular similarity measures. [159]

#### - Examples of use:

Many researchers explored the use of collaborative filtering recommendation and did significant work to improve it. Based on the effect of e-learning group behavior, Liu [249], [250] developed a collaborative filtering method. And a problem matrix and noise management techniques were suggested by Wang et al. [251], [244], [252]. Additionally, sequential pattern mining was used by Bourkoukou et al. [253] for creating a scoring model and gathering input from students to extract preferences in order to assign weights for learning items to produce the most suitable ones. Where "sequence behavior patterns", that was also used by Fatahi et al. [254], has been utilized to forecast users' preferred learning styles. The K-means algorithm for unsupervised model learning was used by El-Bishouty et al. [255] to identify the learning sequence and link learning items to the learner's preferred learning style. Trust relationships of target users with neighbors were used by Guo et al. [256] in addition to a user-item rating matrix and harmonic parameters to get the recommended reading list. Genetic algorithms were used by Dwivedi et al. [257] to provide an effective learning route for students. Ant colony optimization techniques were used by Kamsa et al. [258] for enhancing a path of collaborative learning. Deep belief networks in conjunction with massive open online courses were used by Zhang et al. [259]. They included prediction ratings to increase learners' effectiveness.

However, despite the great development efforts, there are still flaws and shortcomings.

#### - Limitations:

Providing accurate recommendations requires a lot of time to get the required amount of ratings for the learning object, so it is time consuming. Attributes are challenging to assign to items, which results in recommendations with ambiguous suggestions. Collaborative filtering was unable to adequately serve as a component of an effective recommender system on its own due to other issues including cold-start, scalability, and data scarcity. [241]

- 2- Content-based, where suggestions are made in accordance with the preferences of the students' profiles and the contents of the LOs [241]. In order to provide recommendations, it first determines the characteristics that a user has previously favored. Narducci et al. [260] outlines three elements needed to implement content-based recommendation: content analyzer, profile learner, and filtering component. Using feature extraction techniques, the content analyzer derives item representations from the learning object's content. Using machine learning techniques, the profile learner generalizes user input to construct user profiles based on historical preferences. Finally, matching the user profile with recommended items is done by the filtering component. [242]
  - There are two categories of content-based recommendations: case-based reasoning [247] and attribute-based technique [261]. The attribute-based strategy recommends an item if it fits the learner profile's qualities, whereas case-based reasoning techniques recommends an item if it has the strongest connection to previous things the learner liked. [159]
  - Examples of use:

There are a number of examples in this field, including the following. Correlation analysis was used by Chen et al. [262] to arrange and create groups of learning materials. A rulebased adaptive user interface was established by Kolekar et al. [263], and Shu et al. [264] made use of previous student data. Fuzzy clustering and decision trees were also used by Rahman and Abdullah [265] in order to categorize students depending on their academic performance and learning habits. Additionally, linked data in conjunction with social network interaction were used by Pereira et al. [266] for providing recommendations.

#### - Limitations:

Since content-based recommendation systems are dependent on prior user behavior and are unable to suggest fresh content, they demotivate users and cause them to have a limited focus. This shows that the content-based strategy is unable to solve issues with limited data, cold starts, scalability, time requirements, and accuracy. **[241]** 

3- Knowledge-based, whereby suggestions are made based on knowledge models about students and learning objects, with the justification that the majority of learning items are better suited to meet students' requirements [267] [241]. Users' knowledge, item knowledge, and understanding of how items and users' requirements line up are necessary for the knowledge-based recommendation process to be successful [247]. Semantic or knowledge-based recommendation algorithms are employed in several e-learning applications, where ontology-based and context-based methods are among them [268] [242]. An advantage of knowledge-based recommendation systems, especially within e-learning recommenders, is its ability to be combined with other recommendation methods [159].

#### - Examples of use:

Examples on knowledge-based recommendation systems include: the use of dependency ratios and parse trees by Aeiad and Meziane [269] upon an ontology-based model to provide realistic educational materials that meet the needs of learners; the use of classification and analysis by Phobun and Vicheanpanya [270] upon an ontology-based approach to suggest the best methods; the utilization of self-organization theory and a

leaner model by Wan and Niu [271] over a knowledge-based approach for suggesting learning materials; the application of semantic correlation and a dynamic key value memory network by Trifa et al. [272], depending on a knowledge tracing agent; and Nitchot et al. [273], proponents of knowledge representation, created knowledge structures and provided linkages of study materials by combining computing, ontology, and logic. [241]

#### - Limitations:

Given its complexity and cost, knowledge-based recommendations need time and money to complete. Also, not all sizes of e-learning systems can use it. [241]

4- Hybrid, where recommendations are generated based on a combination of two or more of the previous techniques to benefit from their advantages, overcome their limitations, and to improve performance. [274] [241]

#### - Examples of use:

Examples on hybrid recommendation systems include: the use of collaborative filtering in conjunction with sequential pattern mining by Chen et al. [262] for suggesting educational resources; the utilization of topic model in addition to matrix factorization by Zhao et al. [275] to address the issue of recommendations [159]; the use of similarity matrix and a 'tag and ratings' based approach by Karga and Satratzemi [276]; and the augmentation of decision trees, logistic regression, support vector machines, and artificial neural networks by Hussain et al. [277] to forecast the level of challenge pupils will experience.

#### - Advantages:

The drawbacks of both collaborative filtering and content-based filtering are solved by hybrid recommendation. When there are fewer ratings available, it also seems to be more accurate. The hybrid approach is the greatest option right now because of all the benefits. **[241]** 

As we mentioned, there are four main types of recommender systems, but other branches have evolved such as ontology-based recommenders, demographic-based recommenders, utility-based recommenders, context aware-based recommenders, trust-aware based recommenders, and social-network recommenders:

Demographic-based recommenders [278], which recommend items based on personal characteristics. Individuals with comparable characteristics are given similar suggestions.

Utility-based recommenders [279], which generate recommendations for the user by calculating each object's utility.

Context aware-based recommenders [280], which use recommendation algorithms once data sets are filtered based on contextual information.

Trust-aware based recommenders [281], where suggestions are based on the relationships between individuals and their trustworthy opinions.

Social-network recommenders [282], where user relationships and profiles serve as the primary sources of recommendations. Such a recommender is employed on Facebook [283] and LinkedIn [284]. [159]

Ontology-based recommenders [285], where recommendations are generated based on the knowledge represented in ontologies. This type of knowledge-based recommender system uses

ontologies to describe user context, objects, and domain knowledge. Ontologies are used to facilitate knowledge reuse and sharing [159], through representing knowledge with the use of classes, properties, relations and rules. Ontologies might be represented in a variety of languages, although the most oftenly used are Web Ontology Language and Resource Description Framework according to [286] and [159]. Ontologies come in a variety of types, including application, domain, reference, or generic ontologies, where DBpedia is a famous example of generic ontologies [287]. On the other hand, for recommender systems to describe information about students, subjects, and relationships between them, domain ontologies are the best choice [288].

Ontology-based recommenders proliferated in e-learning because of how well they can tailor content to the unique needs of each student. It achieves this capability by making the most of student information, including background, learning preferences, and learning route [159]. And we can find many examples for using ontology recommenders in e-learning such as Sosnovsky et al. [289] and Han et al. [290].

Despite the fact that ontology-based recommenders are effective and popular, building ontologies is a challenging, expensive, and time-consuming process that calls for knowledgeable professionals in engineering and knowledge acquisition [159]. Nevertheless, solutions have been created to address this issue through automatic ontology creation such as Klaussner et al. [291] and Elnagar et al. [292].

# 2.6 Learning object annotation

With the increased availability of learning resources from multiple sources, the need for organizing them is becoming important, not only to make use of these resources, but also to prevent this blessing from turning into a curse. Here comes the role of learning object annotation, where each learning object gets annotated with metadata to make it easier to be archived, retrieved, clustered and recommended for learners. The semantic annotation of learning objects makes the produced learning resources reusable, sharable and machine understandable, and integrating this annotation with learning management systems allows teachers to discover, aggregate, and reuse more learning materials in their courses. [293]

Tagging and annotating learning objects has many advantages, such as facilitating and accelerating the process of searching, retrieving, and using of learning resources, adaptively personalizing the searching process for each learner to get the most appropriate learning resources and, consequently, easing the hole learning process, opening the door for learning object reusability, saving time to create new materials instead of recreating the same content by different institutions. **[294**]

By searching the Scopus database for the terms "Learning AND object AND annotation" and analyzing the results by year we got 3,067 document results through the years 1994 till 2022 as shown in figure 2.17.



Figure 2.17: number of researches on "Learning object annotation" in SCOUPUS database during 1994-2022 [295]

Annotating learning objects by adding metadata can be done through two ways, manually or automatically:

- Manual annotation is done by the creator, the host, or the owner of the learning object. It produces high quality and more trusted results, but this comes with a high price in the required time, effort and cost.
- Automatic annotation is done through machine processing. It provides more information in less time. On its own it has two techniques:
  - Metadata harvesting; where data is collected automatically from metadata that was associated with the learning object from the beginning by humans or by a semi-automated process.
  - Metadata extraction; where data is collected also automatically, but from the learning object's content through mining techniques to produce the required metadata.

The creation of a critical mass of reusable learning objects is difficult without automatic annotation [296]. The Library of Congress is one of the institutions that acknowledged the importance of automated annotation of bibliographic resources. [294]

#### 2.6.1 Metadata Standards

We have shown the different ways of annotation, but when we get to the implementation phase we will encounter many annotation standards from which we can select the one that best fits our objective and domain.

- Dublin Core metadata Initiative [297] [298]

It is a project of the Association for Information Science and Technology, and it offers a type of metadata, drawing from many Resource Description Framework (RDF) vocabularies. Moreover, it enables a wide range of objectives and business strategies by having two levels:

- The simple Dublin Core, which consists of fifteen elements (Title, Subject, Date, Description, Coverage, Type, Source, Creator, Language, Contributor, Relation, Rights, Format, Identifier and Publisher).

- The qualified Dublin Core, which has three extra components (Audience, Provenance and RightsHolder), in addition to a set of components that clarify the semantics of the elements.

Although the Dublin Core metadata includes features essential for general-purpose applications, it is missing characteristics that describe a document's educational viewpoint. As a result, a number of alternative metadata standards, including SCORM Metadata, CanCore, and IEEE Learning Object Metadata, that contain properties identifying the pedagogic qualities of the document, have been established to address educational issues.

- SCORM (Shareable Content Object Reference Model) Metadata [303]

Is a description of the Advanced Distributed Learning initiated from the US Department of Defense's Office. It specifies communications between client-side content and a host system (referred to as "the run-time environment"), which is frequently supported by a learning management system. Sequencing, which is a set of guidelines that dictates the sequence in which a learner may view content items, was first introduced by SCORM 2004. Where they limit a student to a predetermined set of routes through the training materials.

- CanCore (Canadian Core Metadata) [304]

In a nutshell, the Canadian Core Metadata Application Profile is a condensed and in-depth explanation of a subset of the LOM metadata elements. Although the CanCore element set openly bases itself on the LOM standard's elements and hierarchical structure, it significantly lessens this specification's complexity and ambiguity.

- IEEE learning object metadata [299] [300]

A learning object can be anything, digital or otherwise, used for education, training, or learning and this conceptual data schema specifies the structure of a metadata instance for it. This standard's goal is to make using, searching, evaluating, retrieving, sharing, and exchanging learning objects easier. A learning technology system can use a metadata instance for a learning object to manage, identify, evaluate, or exchange learning objects by using other standards' implementation specifications of the data schema as a reference. In order to address educational issues, qualities indicating the pedagogical viewpoint of a document must be extracted, and IEEE LOM has attributes describing the pedagogic aspects of the text. **[294]** 

### 2.6.2 Examples

Many researchers had presented implementations for learning object annotation and we will explore some examples [294].

- Using WebProtege [301], Koutsomitropoulos et al. [293] created and developed a learning object repository to make managing and exploring learning objects easier. In order to promote integration with other discovery techniques, digital archives, and the Web of Linked and Open Data, they also had connected learning objects with a common data model for sharing and integrating knowledge organization systems via the Semantic Web called SKOS [302]. At the Democritus University of Thrace, the resultant technology has already been put into use to facilitate online classes [293].
- A metadata generator called DC-dot [303] was created by UKOLN (UK Office for Library and Information Networking), which is situated at the University of Bath. It is open source and may be updated or shared. It also generates Dublin Core metadata and has the ability to format output using a variety of alternative metadata standards.
- In order to automatically extract Dublin Core metadata metadata, Han et al. [304] suggested a machine learning approach utilizing support vector machines.

- Using a Java Servlet, Jenkins et al. [305] presented a method for automatically creating Dublin Core metadata on a web server, and ten of the essential metadata out of the fifteen are included in the description.
- Dublin Core information is also automatically generated from web pages by Li et al. [306].
   The topic element is obtained by using a neural network, while the other nine components are produced using the same methods as Jenkins et al.'s [305] methodologies.
- An ontology-based method for automatically annotating learning objects based on IEEE LOM was presented by Jovanovic et al. [307]. In addition to some heuristics, content mining techniques were mostly employed to identify metadata items. It is possible to automatically construct pedagogic assets such as examples, summaries, and references. Sadly, only resources in slide format could be annotated.
- Kris Cardinaels et al. [**308**] developed a web service that automates the production of IEEE Learning Object Metadata.
- KIM automatic semantic annotation: The KIM platform [309] was created by Popov et al.
   [310] and offers an automated semantic annotation service. They construct it using the PROTON ontology [311] and a knowledge base that offers in-depth coverage of items of general relevance. The architecture of the KIM platform and its main components are shown in figure 2.18. [294]



Figure 2.18: Architecture of KIM platform [310]

The fundamental concept behind KIM information extraction is the recognition of named entities in relation to the KIM ontology (KIMO), where the entity classes, hierarchical ordering, and suitable properties are also defined by the KIMO ontology. KIM information extraction is built on theGATE framework [**312**], an open source software toolkit supplied by the University of Sheffield that can solve virtually any text processing problem. Several of GATE's document management features were reused, along with generic natural language processing (NLP) elements like Tokenizer, Part-of-Speech Tagger, and Sentence Separator. Figure 2.19 shows the flow diagram of KIM semantic information extraction.



Figure 2.19: Flow diagram of KIM semantic information extraction [310]

# 2.7 Topic identification

Topic identification is crucial for many fields that use the computer power for analyzing text to make best use of it, or to automate hard processes that humans used to do with hard effort in large periods of time but, on the contrary, computers with the use of different techniques start to accomplish in a notably less time. But, to harvest the advantages to using computer power in these tasks, researchers have first to invent and identify the most suitable techniques for each specific field.

Topic identification has been used in many fields. Many of them are obvious such as education, information retrieval, document summarization, topic detection and tracking, text classification or recommendation. But other new fields have been used topic identification for enhancing other applications such as customer service, tourism, movie recommendation, and application reviews.

Thanks to the need for developing and improving in this field, many researchers have presented their contributions causing a growing number of research works.

By searching the Scopus database for the terms "Topic extraction" and analyzing the results by year we got 38,583 document results through the years 1990 till 2022 as shown in figure 2.20.

And by searching the Scopus database for the terms "Entity extraction" and analyzing the results by year we got 11,843 document results through the years 1990 till 2022 as shown in figure 2.21.



By browsing the available technologies that could help in this development, data mining was very obvious to use in addition to many other technologies that evolve and appear by the time such as machine learning and artificial intelligence. Data mining was very suitable for the nature of the data that needs to be processed, which is text in most situations. So we are going to explore some of the data mining methods that could be used.

### 2.7.1 Data mining techniques for topic identification

Data mining has two main categories that depend on the used technique and the available data. It contains supervised learning and unsupervised learning methods that could be applied in machine learning as well:

- Supervised learning [315] [316] is performed utilizing a ground truth, meaning that we have advance knowledge of the anticipated outcome values for our samples. Given a sample of data and desired outputs, a function that closely approximates the connection between input and output shown in the data is what is learned by supervised learning. It may be applied to classification or regression. Additionally, the foundation for creating a model that will subsequently be used to categorize new and unclassified data records requires training datasets and manually pre-classified records that have already been classified.

One of the supervised learning approaches, Named Entity Recognition (NER) [317], assists the user in structuring a text such that pertinent information in the form of entities may be retrieved more quickly.

- Contrarily, unsupervised learning [315] [316] aims to infer the inherent organization contained within a set of data points, since it lacks labeled outputs. It is only dependent on output data and may be applied to clustering.

Identification of frequent nouns and verbs, keyword clustering, and latent semantic indexing are some examples on unsupervised learning techniques [318] [317]:

- To identify common nouns and verbs, they are recognized using part of speech tagging, followed by the extraction of common words and the presentation of the most crucial topics by the extracted word combination.
- To create clusters of keywords, the k-Means clustering method is used on a term document matrix containing TF-IDF weighted review words, using cosine similarity as the distance metric. Then, words with a high TF-IDF value inside a cluster signify terms that frequently appear together in reviews and, as a result, represent latent topics.
- Latent semantic indexing's (LSI) main objective is to minimize the diversity of terms in a text by making use of the fact that multiple words may be used to describe the same subject. LSI uses the statistical method of singular value decomposition and

TF-IDS to condense all comparable words inside a text into a concept, which is then applied as a latent semantic to all words that have the same meaning or that refer to the same subject. The last step is to employ the generated latent words to produce the main output for topic detection, which are latent topics.

### 2.7.2 Research examples on topic identification

Other research examples on topic identification include the following:

- In [319] Chris et al. presented TopCat (Topic Categories), a technique for identifying topics that recur in articles in a text corpus. Chris et al. identified related items based on traditional data mining techniques. Frequent item sets are generated from the groups of items, followed by clusters formed with a hyper graph partitioning scheme. Natural language technology was used to extract named entities from a document, and then look for frequent item sets. Next, groups of named entities were clustered, capturing closely related entities that may not actually occur in the same document. Finally, a refined set of clusters was produced, with each cluster representing a set of named entities that refer to a topic. [47]
- In [320] Veselin et al. presented an algorithm for opinion topic identification by developing a methodology for the manual annotation of opinion topics with the use of fine-grained subjectivity analysis, which could be useful for question answering, summarization, and information extraction. [47]
- In [321] Kino et al. presented a method for automatic topic identification using an encyclopedic graph derived from Wikipedia. Kino et al. used the unsupervised system Wikify [322] to identify the important encyclopedic concepts in an input text automatically. As Kino et al. mentioned "topic identification goes beyond keyword extraction, instead has to be obtained from some repositories of external knowledge". Kino et al. aimed to find topics (or categories) that are relevant to the document at hand, which can be used to enrich the content of the document with relevant external knowledge [47]. Comparing to our work we augmented Wikify with other services in addition to Gensim to produce better results and process bigger textual resources.
- In [323] Freidrich et al. presented a framework for the identification of primary research topics from within a corpus of related publications. Their method uses an unsupervised topic modeling approach to classify new and emerging topics from the entire corpus. Machine learning techniques were used, such as Non-negative Matrix Factorization for Natural Language Processing, as well as an adaptive topic model Bayesian classifier that allows for the identification of new primary topics as papers are added. [47]
- In [324] Khader et al. designed an ensemble method for automatic topic extraction from a collection of scientific publications based on a multi-verse optimizer algorithm as the clustering algorithm. [47]
- In [325] Zhou et al. proposed a topic identification approach for identifying emerging topics in application reviews. They decrease noisy data by utilizing natural language processing techniques, which also included rectifying misspelled words and lengthening acronyms. In order to handle the shortness feature of application reviews and follow their growth, they additionally employed the adaptive online biterm topic model [326]. They base their strategy on three stages:

- Preprocessing, where the text is cleaned and prepared with natural language processing tools such as Natural Language Toolkit (NLTK) [327], Wiki Dictionary [328], PyCorrector [329] for correcting misspelled words, pointwise mutual information (PMI) [330] for extracting typical phrases, and co-occurrence frequency for identifying more meaningful combination of words.
- Emerging topic identification, where AOBTM [**326**] and anomaly detection methods are used for identifying emerging topics.
- Topic interpretation, where emerging topics are interpreted by labels.
- In [**317**], with data mining techniques, Menner et al. provide a general method for identifying subjects from tourist-related user-generated content. They accomplished their goal through four phases:
  - Document Retrieval, where they use a site crawler and particular regular expressions to get the reviews from the TripAdvisor social media platform.
  - Document Extraction, where they use additional regular expressions and XPath expressions to extract reviews from the HTML pages that the web crawler has gathered.
  - Document Processing, where each review is subjected to a particular preparation, such as tokenization, stop-words filtering, stemming, and the transformation to lower case, in order to make it suitable for each data mining approach. Ultimately, depending on values for term frequency or term frequency-inverse document frequency, all reviews are converted into a term document matrix.
  - Mining, where various data mining methods, including supervised and unsupervised learning via the data mining application RapidMiner Studio [**331**], were applied to the reviews.

### 2.7.3 Enrich topic identification with new ideas

Documents that refer to a specific topic might also refer to several sub-topics of it, and then the single-layer method could present problems to define the intended topics, even when a document is analyzed manually by human experts. A new multilayer topic structure was proposed in [332], which aims at automatically building a multilayer topic structure. This technique is intended to identify sub-topics where units within the same sub-topic should be very similar and units from different sub-topics should be dissimilar. The mentioned paper used the hierarchical agglomerative clustering algorithm to establish the hierarchical topic tree. It started by using each unit as an independent cluster. Then, the similar ones merged together. This process was repeated until all clusters merged into one cluster, and from this hierarchy a tree structure was established. [47]

By using multilayer topic structure [**332**], books with common topics will be semantically similar, and could be linked together to provide a network of resources that are related to a general topic. So, we can provide this information in a Dynamic Interactive Knowledge Graph [**47**], which can be updated continuously with every new added book to the database, and can be browsed by every user. This graph-based learning technique is an emerging search field and proved to be effective and useful for the learning process for students and teachers, as Weber et al. [**48**] stated that the graph-based learning technique improves searchability for new sources and provides explainable results and result recommendations. [**47**]

Zhijun et al. [333] also used an interactive group knowledge graph to visualize the relationship between knowledge points concluding that interactive group knowledge graphs have a significant

promotional effect on teachers' online learning, that it is beneficial to teachers' professional development, and that it is an effective method of enhancing the depth and breadth of the interaction of students' learning. In addition, Zhijun et al. [333] defend that the process of generating collective knowledge graphs can enhance students' enthusiasm for autonomous learning and promote meaningful and in-depth interactions between different students. [47]

# 2.8 Important tools for our system

After exploring all the previous topics we must mention a specific set of tools that play a crucial role in our system, which are Wikipedia Miner, TextRank, BM25, and Gensim. We will elaborate more about them in the next sections.

## 2.8.1 Wikipedia Miner

Wikipedia [334] is an enormous, continuously expanding tapestry of linked pages that serves as an online encyclopedia. For programmers and academics, it is a massive, multilingual library of concepts and semantic relationships that may be used as a resource for natural language processing and a variety of other fields of study [335]. But there was a need for a tool that provided advanced services to make the best use of such an unexplored mine of knowledge. Wikipedia Miner [336] is an open-source software system that uses the articles in Wikipedia to extract and identify the main topics in text through its rich semantics. With the use of a Java API, it builds databases with condensed versions of Wikipedia's structure and content. The articles, categories, and redirects on Wikipedia are organized into classes that may be effectively searched, browsed, and iterated through. XML-based web services, machine-learned semantic relatedness metrics and annotation features, and parallelized processing of Wikipedia dumps are some of the advanced features the toolkit offers.

Furthermore, it contains a topic detector feature that gathers all labels in the document. For most documents, Wikipedia probably knows something about the topics discussed and could likely add additional information. Once Wikipedia Miner detects the topics within a document, it is easier for automatic systems to process those learning resources for tasks such as classification, recommendation, or data retrieval [47]. The Architecture of the Wikipedia miner toolkit [336] is shown in figure 2.22.



Figure 2.22: a description of the Wikipedia miner toolkit's architecture [336]

Wikipedia and Wikipedia Miner have been used in many fields such as automatic topic indexing [337], document clustering [338], document summarization [339], the classification of multilingual biomedical documents [340], converting concept-based representations of documents from one language to another [341], identifying the prerequisite relationships among learning objects [342], classifying news articles [343], evaluating and classifying Open Educational Resources and OpenCourseware based on quality criteria [344], and for group recommendation by combining topic identification and social networks [335].

### 2.8.2 TextRank

TextRank [345] is a graph-based ranking algorithm that could be used for keyword extraction, or sentence extraction, but before discussing it we have to explore graph-based ranking, which uses voting or recommendation as its fundamental concept.

To utilize graph-based ranking algorithms on natural language, it is important to create a graph that represents the text and connects words or other text elements with appropriate relationships, where text units of varying sizes and qualities, such as words, collocations, full sentences, or other text units, can be included as vertices in the graph, depending on the application at hand. Linking to another vertex is essentially a vote for that vertex. A vertex's significance increases as the number of votes cast for it increases. Moreover, the ranking model considers the significance of the vertex casting the vote, which impacts the significance of the vote itself. [345]

The following major steps are involved in applying graph-based ranking algorithms to natural language texts: [345]

- The first step is choosing the text units that best describe the current work, adding them as vertices on the graph.
- The second step is finding the relationships that link these vertices, then use those relationships to create edges between the vertices, where the edges may be weighted or unweighted, directed or undirected.
- The third step is repeating the algorithm all over the graph and text till convergence.
- And finally, vertices are sorted according to their ultimate score to be ready for selection decisions.

For determining the significance of a vertex inside a graph, TextRank considers global information that is recursively derived from the whole network, as opposed to only local vertex-specific information. Kleinberg's HITS algorithm [346], Google's PageRank [347], or Positional Function [348] are comparable to TextRank, and are simple to include into the TextRank model [349].

Rada et al. [345] explored and assessed the use of TextRank for two tasks related to natural language processing, sentence and keyword extraction. In the keyword extraction task based on lexical and grammatical properties, their system is taught to identify keywords in a document, where this strategy was initially proposed in [350] using a genetic algorithm and parameterized heuristic rules. However, in [351] better results were obtained through the use of the Naive Bayes learning scheme. A higher level of results were presented in [352] by using a supervised learning system and a mixture of lexical syntactic features.

#### TextRank works as follows: [345]

- Text processing (tokenization, annotation, tagging, and syntactic filtering).

- Adding text units to the graph as vertices, and linking every two vertices that occurred in the same word.
- Repeat the previous step until the whole graph is completed.
- Assigning default weight =1 for all vertices, then run the ranking algorithm over the graph until it converges.
- Finally, sorting vertices in reverse order according to their weights, keeping top vertices for further processing.

TextRank is known to be extremely adaptable to various domains, genres, or languages because of its limited need from linguistic knowledge and language specific annotated corpora [345]. It gains reliability and efficiency by not depending solely on the local context of a text unit, but instead relying on information drawn recursively from the entire text. It has been used in different research fields such as social networks, analysis of the link-structure of the World Wide Web, and citation analysis. [345]

From the previous discussion we can see that TextRank could be useful in our research, and could have an active role in our system, but it could be improved and enhanced with the next algorithm as we will discuss.

### 2.8.3 BM25

TextRank calculates the similarity or the relation between two phrases or words based on the content they share. To avoid boosting big sentences, this overlap is simply calculated as the number of shared lexical characters between them divided by the length of each. There are numerous metrics for identifying relations between phrases or words, which are represented on the graph by the edges connecting the vertices: overlapping words, cosine distance, and query-sensitive similarity, or by combining the previous ones with supervised learning functions [353]. And the same techniques could be used for creating summaries of documents by extracting and combining the most important sentences (vertices) from the documents [354].

The 1970s and 1980s saw the invention of one of the most effective text-retrieval algorithms, BM25, which served as the foundation for the Probabilistic Relevance Framework (PRF), a formal framework for document retrieval [**355**]. BM25 is a probabilistic variant of the TF-IDF model [**356**], where BM25+ is a variant of BM25 that modifies the penalty for lengthy documents [**357**]. And since TextRank requires a similarity calculation technique, and BM25 supports this task with a superior quality, then augmenting them both had already provided significant improvements [**345**].

According to [358], they tested many different techniques to calculate the similarity between vertices for TextRank, and the best results were provided by BM25 and BM25+. They mentioned that the robust method for artificial summarization that is produced by combining TextRank with contemporary Information Retrieval ranking functions like BM25 and BM25+ outperforms the prior standard strategies. [358]

And last, but not least, we are going to give a short hint about the very useful multi tool of data analysis, Gensim.
#### 2.8.4 Gensim

Gensim [359] is a well-known rich Python library that has a great amount of useful tools for many fields. It was mainly created for two areas: indexing and similarity searching for digital documents, or the Latent Dirichlet Allocation and Singular Value Decomposition methods, which are quick, memory-efficient, and scalable. The basic algorithms of Gensim's large-scale, distributed, online SVD and LDA are comparable to the Swiss Army knife of data analysis, despite the fact that it was designed for massive digital libraries. Additionally, it has applications that are independent of the field of natural language processing. The Latent Dirichlet allocation (LDA) [360] topic model develops topics based on word frequency. It is a three-level hierarchical Bayesian model in which each component of a collection is represented as a finite mixture over a foundational set of subjects. Every topic is modeled individually as an infinite mixture over a base set of topic probabilities. The topic probabilities give an explicit representation of a document in the context of text modeling. Singular Value Decomposition (SVD) [361] [362] is a tool for data reduction that allows us to take high-dimensional data and reduce it to its essential components for analysis, interpretation, and description.

# 3. Our system Contribution

All the proposed system is just a design without implementation except our algorithm for the topic identification part, which has been implemented, tested and evaluated as we will explain in chapters 4 and 5.

In previous chapters we introduced the main topics related to our research objectives. In this chapter we will present the design of our system, exploring it and the interaction between its parts.

We present the design from three different perspectives: e-learning, semantic web, and cloud computing. In each perspective we will see the required areas to augment into our system, how the system will manage and use these parts, and how system users will benefit from them. After presenting and elaborating our system from the three main perspectives we will show the full design.

# 3.1 E-learning perspective

In our system we want to present a student centered learning style, considering the individual differences and requirements for each learner, and taking into account their different learning styles and previous knowledge levels. The adaptive e-learning model [92] is the most suitable for our goals. In order to benefit from its advantages, we have to overcome the following challenges:

- Identifying the preferred learning style for each learner.
  - By applying a questionnaire [91] before logging into the system to prepare the relevant learning content and presentation method.
- Measure learner's previous knowledge level.
  - By giving each learner a pre-test [91] prepared and evaluated by the institution staff.
  - To determine the appropriate knowledge level for the student, Ghatasheh proposed in [156] a concept for a dynamic e-learning environment based on machine learning and the study of user activities
  - By mixing the questionnaire's result with the pre-test result useful information could be concluded about learners regarding their preferred presentation and study method, level in courses and materials to start with, and future learning path recommendation.
  - Additionally, with a good amount of information provided from the learner, we can conclude more useful and advanced information regarding skills and goals. All the required information is saved to the learner's profile on the learner's data repository for future use with the information that will be recorded in the future about their courses, evaluations, and paths. In figure 3.1 we present the required steps before logging into the system.
- Prepare each individual course in various formats and styles.
  - Where teachers, educational specialists, and information technology staff have to collaborate to produce this variety of learning resources and upload them onto the system platform repository to be ready for learner's requests.
  - The learning resources could be text, audio, video, practical exercises, virtual reality experiments, or any other format. In figure 3.2 we present the learning resources preparation process.



Figure 3.1: required steps before logging into the system



Figure 3.2: learning resources preparation process

After providing the required learning resources and learners starting their journey through our elearning system, a very important process has to work in parallel: the observation process. The continuous observation for learner's activities and evaluations helps improve the system adaptivity and makes it dynamic through the use of data mining techniques and data analysis [91], and the more the observed information the more adaptivity will be provided. In figure 3.3 we present the continuous observation process.



Figure 3.3: the continuous observation process

Other factors such as a student's personality, learning in social groups [103], active student participation, and motivation could enhance the learning experience if taken into account through the design process [102]. Additionally, there are many theories that we mentioned previously in chapter 2 such as the Learning as a Network learning theory [43] and the connectivism theory [44] [45], which proved to be effective and more productive in the learning process, and fortunately they are totally adequate for our goal of creating an interactive knowledge graph [47] [48].

For all those reasons we are going to apply the Learning as a Network learning theory [43], and the connectivism theory [44] [45] into our system through the use of students' properties, which come from the first login stage, or from the continuous observation information, which is stored

in learner's data repository, or even from commonly taken or favorite learning objects from the learning object repository.

Data mining techniques and recommendation tools are used for recommending other students with common interests to enable students forming their learning groups as shown in **figure 3.4**.



Figure 3.4: Student group recommendation

Another technology that proved to be useful is augmented reality (AR). Many researches showed its effectiveness and positive impact on students' performance [107] [108] [109] [110] [111] [112] [118]. The system will support AR materials that students can interact with individually or in groups, to make the learning process more interactive and exciting.

## 3.2 Semantic web perspective

Representing the learning resources or any kind of information on the computer without giving it the ability to understand its meaning only allows us to compare and search specific keywords. On the other hand, representing the same information semantically on the same machine opens a huge gate of abilities due to the information understanding ability. The semantic representation in the e-learning system also requires adding metadata for learners, learning objects, and learning domains. In this section we present some design guidelines for this part of the system, in addition to exploring some processes and interactions that help enrich the semantic information content.

## 3.2.1 Learner

Because our system aims to be learner centered and uses the social learning model, we must provide learner's data in both social network and e-learning structures. From the social network point of view we used the friend of a friend (FOAF) ontology [**363**], which allows groups of people to form social networks without the need for a centralized database. It enables our system to support the social learning activities and improve the interaction between learning group members, in addition to the great advantages that could be gained from social network analysis [**157**] to support the learning process and sharing information between different systems. According to the FOAF, ontology each learner has a unique identifier, name, mail, homepage, interests, and a "knows" property that links to other learners.

In addition to these properties, we must use the learner's learning history and learning preferences such as: learning style, knowledge level, taken courses, tests evaluations, and all other continuous observation information, as shown in figure 3.5.



Figure 3.5: Some leaner's properties from FOAF Ontology [363]

Another learner's ontology that could be useful for our system, which includes more required information such as knowledge, learning goal, mastered content, learning interest, learning style, learning time, and location, was presented by Yu, Zhiwen, et al. in [188] as shown in figure 3.6.



We are going to use a combination of both ontologies to represent learner information, as shown in figure 3.7.



Figure 3.7: Our representation for Learner Ontology

Moreover, the personality information should be taken into account due to its positive impact on the learning process [144]. This information could be collected from: the Myers–Briggs Type Indicator (MBTI) questionnaire [364], the continuous observation process by the system or teachers, or from the learners' friends in their learning group. The MBTI questionnaire attempts to assign a value to each of four categories: introversion or extraversion, sensing or intuition, thinking or feeling, and judging or perceiving. This evaluation helps in identifying how people perceive the world and make decisions. Other factors could be activated in the future through the use of emotional data observation technologies to be augmented into the dynamic adaptivity.

### 3.2.2 Learning domain

For describing the learning domain, each subject has its own domain that describes different courses, topics, and the relations between them. The teaching staff within the institution has the responsibility for organizing each learning subject domain to guarantee that it contains all the required topics and they are in the right order. An example of learning domain ontology is presented in [188], as shown in figure 3.8, which includes the learning domain for computer science subject.



Figure 3.8: learning domain example for computer science subject [188]

### 3.2.3 Learning object

According to [152], in order to describe a learning object semantically three representations are required: content, context and structure, as shown in figure 3.9.

- Content representation: it describes learning materials or learning objects by identifying general properties such as title, type, language, author, prerequisites, and so on.
- Structure representation: it describes the logical structure of learning objects and how they are ordered in the learning path, taking into account which learning object is a prerequisite for which.
- Context representation: it defines different presentations techniques and formats for the different learning styles.



Figure 3.9: Learning object representations [152]

The semantic representation should be replicated for the same learning object for each offered learning style when adding these semantic representations within an adaptive e-learning system. For example, if we have a learning object (LO1) that has the three semantic representations (content1), (context1), and (structure1), this learning object LO1 could be used for visual, auditory, or tactile learning style as shown in figure 3.10. Although it takes a lot of work to create, its reusability and wide range of learning styles are major advantages.



Figure 3.10: Example of required Learning object representations

The content representation should be the same. However, the context and the structure representations could be changed according to the learning style in use. For the general representation, the Dublin Core learning object metadata [150] [298] could be used. It includes metadata elements helpful for general-purpose applications, but it is devoid of information about a document's pedagogical perspective.

Consequently, other metadata standards, including IMS Metadata, SCORM Metadata, CanCore, and IEEE Learning Object Metadata, have been developed to address educational concerns. These standards contain attributes that describe the pedagogical characteristics of the learning object, so that it is very suitable for our representation. The IEEE LOM elements are shown in figure 3.11.



Figure 3.11: IEEE LOM elements [365] [366]

Another simple model for describing learning content is presented in [188] as shown in figure 3.12, including keyword, subject, difficulty, author, language, title, content type and interactivity.



Figure 3.12: Learning Content Ontology [188]

In order to facilitate and speed up the learning objects annotation process for the enormous amount of educational resources available, we can use the automatic semantic annotation KIM platform [309] that was created by Popov et al. [310]. We additionally could augment KIM's annotation results with the ontology-based automatic annotation method that was presented by Jovanovic et al. [307] for learning objects based on IEEE LOM.

### 3.2.4 Enriching learning content

For enriching learning content, more semantic information could be produced and augmented into our semantic database using the LOD (linked open data) tool created by Garcia et al. [154] and DBpedia through the use of natural language processing, entity recognition, and semantic information. Nevertheless, the teacher's role is vital during the data augmentation phase, where the teacher must select and approve the valid extracted data to be added into the learning content.

In order to find relevant entities to display to the user, LOD [154] analyzes the learning content provided by the LMS system (Sakai LMS). Lessons content will initially be displayed without any enhancements when the teacher first accesses the application. The teacher will receive a list of checkable entities if they select the import option, which will cause an entity recognition algorithm to be run. These entities are listed with a confidence level that indicates the likelihood that they will really be included in the lesson content. The system will eventually display the lesson content with the enhancement applied once the teacher has chosen the pertinent entities as shown in figure 3.13. The problem with this system is that it builds upon a specific LMS (Sakai), but the authors planned to solve this problem in their future work.



Figure 3.13: Components of LOD (linked open data) tool [154] for enriching learning content

We are going to use the LOD tool [154] for semantic information enrichment after adding our modifications on the entity recognition and topic extraction algorithm and augmenting it with WikipediaMiner. And finally, searching Wikipedia, or other external resources, for the topic and related topics. At this stage it is the institution's or the teacher's decision to enable adding the new resources to the learning object repository or just adding the links to the learning object metadata as extra helpful resources.

#### 3.2.5 Knowledge discovery

We had created our own algorithm for topic identification using Wikipedia miner and machine learning techniques, which will help discover the semantic knowledge within text. We will provide a full explanation about our algorithm in chapter 4.

Another useful tool for knowledge discovery, which could be beneficial for our system, was proposed by Capuano et al. in [172], called Electronic Justice Relationship Management (eJRM). It uses legal ontologies created by enriching numerous thesauri with common sense knowledge from Wikipedia, in addition to other components such as:

- SEMANTO [183], a graphical ontology management system for knowledge discovery that allows for the updating, erasing, saving, and restoring of several ontology versions.
- The algorithm described in [184], which automatically links each ontological notion to a series of phrases that are weighted and correlate to Wikipedia subjects, thereby enhancing semantic information material from outside sources.
- The algorithm described in [185], to identify terms from many ontologies that correspond, creating an integrated ontology as a consequence.

And for semantic search they follow the following steps:

- From a user-input text, extract pertinent concepts using the knowledge discovery process described in [187].
- Using the common sense knowledge base of Wikipedia, extracted concepts are then enhanced with terminology from external sources.
- Concepts are then organized into categories based on how closely they match up with terms in the managed legal ontologies, and they are linked to the repository of legal documents maintained by the Italian Supreme Court of Cassation.
- By comparing the enhanced terms with those associated to the ontology concepts, it is feasible to find pertinent legal concepts linked to the input text while doing a semantic search for related material.

The knowledge discovery process of the eJRM tool [172] is shown in figure 3.14 as we understood from their paper.



Figure 3.14: The knowledge discovery process of the eJRM tool [172]

#### 3.2.6 Learning path generation

Another important service for students is the creation of learning paths that align to their learning goals and background. Capuano et al. in [172] presented a good idea for providing a training path for learnersas shown in figure 3.15 by:

- Extracting concepts from input text.
- Sorting the concepts according to their educational relationships
- For each concept, finding related learning material/materials from the learning object repository.



Figure 3.15: The training path generation process [172]

By augmenting the previous steps with the "prerequisite" and "difficulty" properties of the learning objects, in addition to the "knowledge level", "learning goal", "learning interests", "learning style", and "learning history" properties of the learner, we will have the ability to produce fully customizable learning paths for each student independently as shown in figure 3.16.



Figure 3.16: The customizable learning path generation process

### 3.2.7 Semantic recommender system

With the continuous development of e-learning more and more learning objects are generated, making finding the required learning object a hard task. In order to tackle this problem, our system must provide an up-to-date recommendation service that augments the required technologies for helping learners and ease the learning process.

The general process of e-learning recommendation is presented in [241] as shown in figure 3.17.



Figure 3.17: The general e-learning recommendation process [241]

The steps of the recommendation process are presented in [188], as shown in figure 3.18.



Figure 3.18: The recommendation process [188]

In addition to the previously explained types of recommenders that were discussed in chapter 2, section 2.5.1, the recommendation service will be a hybrid recommender of different types to eliminate the limitation of each individual type. It will apply:

- Collaborative filtering, by gathering feedback and previous learners' opinions about learning objects.
- Social-network recommender, by using already-existing social networks to spread educational content and stimulate demand for it while also making use of the knowledge gained from the learning group structure we deploy.
- Content-based recommender and ontology-based recommender, by using the knowledge amassed about the learner and the learning object to bring them together and deliver the necessary resources in accordance with preferences, objectives, and educational background.

The interaction between the different recommender types are shown in figure 3.19.



Figure 3.19: Our recommendation process

The recommendation process starts when a new learner enters the e-learning system, where the result of a questionnaire is analyzed, or a behavior pattern is downloaded from another system, providing the required information about the learner.

The learner profile information is stored into the learner ontology and fed also into a similarity function to calculate the similarity between the new learner and previous ones, in internal learning groups and external social network groups. For both types of learners groups, opinions and feedback about each learning resource are collected and stored into the learning object model and fed into the content ontology.

Based on the similarities to other users estimated by the system, it produces social network recommendations and collaborative recommendations.

Next, the learner profile information, in addition to the learning object model, are fed into different machine learning algorithms with adaptation rules to get the adaptation outcomes, which are in turn fed into a recommendation module that matches learners information with learning objects information to produce the content based recommendations.

The ontology recommendation list is produced by using the learner ontology, the content ontology, and the domain ontology as proposed in [188].

Finally, all recommendation lists are combined and redundancy is removed, and learning objects are sorted according to learners' needs in the recommendation augmentation module, to produce the final recommendation list.

We can present the recommendation list in a dynamic interactive knowledge graph to enhance search ability.

# 3.3 Cloud computing perspective

Now we are going to explore our system from the third perspective, which is cloud computing. As we saw in the previous section, there are many tasks to handle such as learning object storing, retrieving, and recommendation, data augmentation, learner's data storing and continuous observation, topic identification, data mining, and data analysis for dynamic adaptivity. All those tasks require a great computational power that, in turn, requires a big team of workers for operation and maintenance. Additionally, the required tasks could be scaled up or down according to the number of users and learning objects. The use of cloud computing services is the most feasible solution for this scenario.

According to [196] [193] [194] and [192], there are three service models for cloud computing: software as a service (SaaS), platform as a service (PaaS), and infrastructure as a service (IaaS), as we mentioned in chapter2, section 2.4, as shown in figure 3.20. According to [202], IaaS includes servers, virtualization, storage, and networking; PaaS includes middleware, operating system, and runtime in addition to all IaaS components; SaaS includes data and applications in addition to all PaaS components. We can also see that PaaS builds upon IaaS, and SaaS builds upon PaaS.



Figure 3.20: Service models for cloud computing [202]

Other categories could be branched from the previous main three such as: Data as a Service, Storage as a Service, and Hardware as a Service, as we explained in chapter 2, section 2.4, where Data as a Service refers to hosting and provision of data in various formats and from numerous sources, and Storage as a Service refers to hosting and delivery of virtualized storage on demand. Regarding our system needs, we will use:

- Platform-aaS cloud, to build our own applications and provide the required services including learning object storage, retrieval, and recommendation, topic identification, data augmentation, learner's data storage, continuous observation, data mining, data analysis for dynamic adaptivity, learning groups and learning activities management, and dynamic knowledge graphs generation and updating.
- Hardware-aaS cloud, to provide the required computation power for our system tasks.
- Data-aaS cloud, to provide the required learning resources and learners' data, at the beginning.
- Storage-aaS cloud, to provide the required expandable storage for learning resources and metadata.

Finally, our system will be provided for teachers, students, and institutions as Software-aaS, since it should include all the components and should be ready to use. After building our own database, we will provide it for other institutions and systems as Data-aaS. The cloud service models are shown in figure 3.21.



Figure 3.21: Required cloud services models for our system

Regarding the cloud computing deployment models, we are going to use a hybrid cloud, since we need:

- A private cloud for private data such as students and teachers' information.
- A community cloud for the students' learning groups and learning activities.
- A public cloud for learning resources that should be available for public use.

Those cloud computing deployment models are shown in figure 3.22.



Figure 3.22: Required cloud deployment models for our system

In [239] Dong et al. presented an e-learning ecosystem based on cloud computing, as we discussed in chapter 2, section 2.4.1. Their system is composed of three layers and four modules, as shown in figure 3.23.



Figure 3.23: Cloud computing infrastructure-based e-learning ecosystem architecture [239]

Another cloud-based reflective learning environment, presented in [206], depicts the three main components, cloud service, teacher side, and student side, as shown in figure 3.24.



Figure 3.24: Framework of the cloud-based reflective learning environment [206]

After identifying our system's services and its required cloud models, and exploring some examples of other systems in use, we are going to present the structure of our system from a cloud computing perspective, as shown in figure 3.25.

Our system will be based on a Platform-aaS cloud model to build our system services. However, we will provide it as a Software-aaS for institutions, teachers, and students, and for activating learning object reusability and social learning, we can provide learning objects, learning groups, and learning activities information as a Data-aaS for other institutions and to other e-learning systems.

Our system structure uses three main layers:

- Infrastructure layers, where we use a Storage-aaS cloud model to store our learning objects and a Hardware-aaS cloud model to manage this data. We provide this data in a public cloud.
- Content layer, where we use a Platform-aaS cloud model that contains Data-aaS for providing new students with information from other resources, and Hardware-aaS for processing current students and teachers' data. They are then augmented with observation

data, and other system processes are used to conclude specific data for other tasks. Learning groups and learning activities information are also processed and managed in this layer. This layer has a community cloud for learning groups inside and outside the institution, and a private cloud for specific information about students and teachers of the institution.

The application layer, where we build up the required applications, webpages, and interfaces required for each kind of user (institutions, teachers, students and other e-learning systems). Each of them is provided with the required services in an easy and effective way as a Software-aaS cloud.



Figure 3.25: Our system from the cloud computing perspective

## 3.4 Augmenting all the three perspectives

After we elaborated our system design from the three perspectives, we sum up all the main steps in one graph as shown in figure 3.26.



Figure 3.26: Our system design from the three perspectives

Our system has two main types of users: institutions and students, where teachers are augmented as a component inside the institution. Required information about students could be gathered from questionnaires, pre-tests, or from external learning groups. Learning objects and their metadata could be created by the institution, gathered from external institutions, or enriched from Wikipedia and other semantic resources.

If we have the required information about students and the learning objects, we can operate the system processes using a Hardware-aaS cloud to produce the required information and execute scalable operations to manage recommendations and adapt the e-learning system according to the student's needs depending on the information gathered during the observation process.

# 4. An algorithm for topic identification in textual learning resources

## **4.1 Introduction**

The algorithm we developed to extract main topics from textual learning resources is the primary contribution of our work. In this chapter, we'll discuss the algorithm and how its output may be used to cluster learning resources and build dynamic, interactive knowledge graphs.

Due to its dependability, effectiveness, and constant content updating by a sizable user community, Wikipedia and its semantic information on DBpedia can be used in a wide range of applications and fields. Consequently, it will be a rich information source from which our algorithm can learn and expand the body of knowledge that it uses to extract important topics. In addition, Wikipedia Miner is a crucial and helpful tool for drawing out this critical information from the vast amount of data on Wikipedia.

Several attempts have been made in the past to accomplish some of our goals. In [336] David and Ian presented Wikipedia Miner although they evaluated it for news articles. In [337] Olena et al. used Wikipedia and Wikipedia Miner for automatic topic indexing, however they did not mention the document sizes. In [321] Kino et al. presented a method for automatic topic identification using an encyclopedic graph derived from Wikipedia. They used the unsupervised Wikify service. However, the Wikify service is not able to produce results for larger resources such as many of the books in our dataset. In [319] Chris et al. presented TopCat, which was effective in identifying topics in collections of news articles. In [320] Veselin and Claire presented an algorithm for opinion topic identification. In [323] Freidrich et al. presented a framework for the identification of primary research topics. In [324] Khader et al. designed an ensemble method for automatic topic extraction from a collection of scientific publications. However, all this research focuses on news articles, opinion topic identification, research topics or scientific publications, which are smaller than the learning resources in our dataset.

Wikipedia Miner is a crucial and helpful tool for drawing out this critical information from the vast amount of data on Wikipedia. The method used by Wikipedia Miner begins with natural language processing of the text document to extract key words and phrases, gathering overlapping word n-grams, where n ranges between one and the maximum label length in Wikipedia, and determining which terms and phrases correspond to Wikipedia concepts.

However, this method has a significant limitation: the bigger the piece of text that needs to be processed, the longer it takes, and the less effective it is. In consequence, we complemented Wikipedia miner with other tools to overcome this limitation and enhance the efficiency of the results.

We applied TextRank [345], a text processing ranking strategy based on graphs. It is an unsupervised algorithm for automatically summarizing text that can identify the most crucial words in a document without the use of labeling or a training corpus.

The TextRank algorithm was later improved upon by Federico et al. in [367], by introducing the "BM25 ranking function", which is a ranking function widely used in the state of the art for information retrieval tasks. BM25 is a variation of the TF-IDF model using a probabilistic model. The results underwent substantial development and improvement, leading to this merging. By employing BM25, they were able to increase results by 2.92% over TextRank. Therefore, the combination of TextRank with modern information retrieval ranking functions such as BM25 creates a robust method for automatic summarization that performs better than previous standard techniques [367]. Therefore we used TextRank combined with BM25.

We applied a quantitative research methodology to evaluate our proposal. The experiments we performed on the algorithm are described in Chapter 5. Their objectives were to optimize a configuration parameter of the algorithm (the parent category level, which will be explained later in this section), to evaluate the precision of the main topics extracted by the algorithm and measure the time it takes to produce those topics, in comparison to another state of the art algorithm. The experiments were based on a dataset of more than 500 full text learning books we collected from several repositories. More explanations and details provided in chapter 5.

# 4.2 Algorithm phases

### 4.2.1. Extract main keywords

We started by extracting text from books, and then we applied text processing to extract main keywords using natural language processing techniques. We used the TextRank [345] graphbased ranking algorithm combined with the BM25 [367] ranking function. This technique was explained in chapter 2 (sections 2.8.2, 2.8.3). We applied that algorithm on the text extracted from the learning resources to extract main keywords. We used the implementation of this algorithm provided by Gensim [368] version 4.1.2, which is a very popular open-source library for unsupervised learning implemented in Python [358]. The output of this step is a sorted list of vertices "keywords" according to their importance "weight" in the text, where the top vertices are the most important.

## 4.2.2. Identify main topics

To identify the main topics in each learning resource, we have to understand the difference between keywords and main topics:

- Keywords are the important words in a text that could represent a topic mentioned in the book, and they are an exact match to text that appears in the book.
- Main topics are the most important topics found in the book according to a category from the Wikipedia database, which have a specific identifier and a URL [369]. They are not necessarily an exact match to words mentioned in the book, since they could also appear with semantically related words.

Example: a keyword could be the word "business", which appears in the book's text. A main topic related to that keyword could be the Wikipedia category titled "Business", with identifier 771152 and URL [**370**]. Therefore, main topics are pre-specified categories in Wikipedia, which could provide more information and link to related resources that are not necessarily mentioned in the input text.

After extracting the main keywords from the first phase, we use them as an input to the topic identification phase, where we use Wikipedia Miner's services (Wikify, and Search) to extract article IDs that contain all the keywords.

If the number of retrieved results is under a specific threshold, we reduce the number of input keywords and repeat this step. The reason is that Wikipedia Miner's services retrieve an article if it contains semantic relations with all the input keywords, so by reducing the keywords number we get more results. The output of this step is a list of article identifiers.

We reduce the number of input keywords by removing the last keyword from the sorted keywords list, because the last one is the least important one. By using this reduction method we enable Wikipedia Miner's services to focus on the most important keywords to find relations between them.

Next, we have to get the properties of each article by using more Wikipedia Miner's services (exploreArticle, suggest, exploreCategory, parentCategory):

- By using the "exploreArticle" service we extract article's definition and title.
- By using the "suggest" service we extract other suggested categories' identifiers to be processed and added to the results. They are processed using the "exploreCategory" service to extract categories' definition and title, and processed using the "parentCategory" service to extract the higher rank category for each one, and this step could be repeated as required to formulate a hierarchical structure of connected topics. After extracting all required categories we process them to produce a sorted list of them.

The output of all previous steps consists in a sorted list of main topics representing the input text, which could be further processed later to be presented to the user in a dynamic interactive connected knowledge graph. The structure of our algorithm is shown in figure 4.1.



Figure 4.1. Structure of our algorithm

The main topics extraction phase uses Wikipedia Miner where Wikipedia's articles, categories and redirects are represented as classes, and can be efficiently searched, browsed, and iterated over. This is a platform for sharing data mining techniques based on Wikipedia data. This data is gathered from Wikipedia's XML dumps. We used a dump of the English version of Wikipedia dated on 22 July 2011, which includes 3.3 million articles and 37.4 GB of uncompressed markup. Wikipedia Miner is the fundamental tool in our Topic Identification system that enables us to assign the right main topic to the learning resources. There is also a parent category feature that enables us to find the parent category of the specified category/topic.

In our initial tests we ran Gensim with the Wikify service [322] of Wikipedia Miner, which is used for keyword extraction and word sense disambiguation. However we found many missing results. To solve this problem we augmented the combination of Gensim and Wikipedia Miner (Wikify) with the Search service of Wikipedia Miner, which solved the problem by searching all available articles for those that are related to the identified keywords. The actual data of this experiment will be presented in the explanation of experiment 2 in Chapter 5.

For each learning resource, after extracting the main keywords with Gensim, we use Wikipedia Miner to extract main topics, then for each topic we identify the parent categories at different levels. We iterate over the retrieved categories to extract multi-level results. This step produces a hierarchy of topics that represents a multilayer topic structure as mentioned in [332]. When we mention the second level of the parent category we mean the second iteration of the parent category service, and so on.

As we can see from the examples in table 4.1, the higher the level the more general and comprehensive the results are, yet at some level those results are too broad and not so useful. We describe in Chapter 5 how we empirically chose the best level of parent category to use in our system.

Table 4.1. Examples of our algorithm's results.				
Example1 - book	Example1 - book main topic (Biology)			
1st level results:	(Immune system/Organ systems/Microbiology/Biology/Clinical pathology)			
2nd level results:	(Biology/Life/Natural sciences/Botany/Anatomy)			
3rd level results:	(Biology/Life/Society/Natural sciences/scientific disciplines)			
4th level results:	(Life/Biology/Nature/Natural sciences/Universe)			
Example2 - book	main topic (Math)			
1st level results:	(Mathematical analysis/Calculus/Subdivisions of mathematics/Analysis/Integral calculus)			
2nd level results:	(Subdivisions of mathematics/Mathematics/Abstraction/Dimension/Geometry)			
3rd level results:	(Mathematics/Subdivisions of mathematics/Abstraction/Structure/Dimension)			
4th level results:	(Structure/Scientific disciplines/Academic disciplines/Abstraction/Dimension)			

All the steps of our algorithm are presented in the next pseudo code and figure 4.1.

Pseudo code of our algorithm for topic identification			
-For each (Learning Resource) in the dataset:			
-Extract (Text)			
-Process (Text) with Gensim to extract (Keywords)			
-Process (Keywords) with Wikipedia Miner services (Wikify, Search) to extract (Article IDs)			
-Copy (Keywords) to (Wikify_Keywords)			
-Copy (Keywords) to (Search_Keywords)			
-While (number of retrieved Article IDs $< 20$ )			
-Process (Wikify_Keywords) with (Wikify) to extract topic IDs			
-Remove the last keyword from (Wikify_Keywords) list			
"to enable Wikify to find more results"			
-If length of (Wikify_Keywords) = $0$ :			
-For each Keyword in (Search_Keywords):			
-Process Keyword with (Search) to extract topic IDs			
-Process (Article IDs) with Wikipedia Miner services (exploreArticle, suggest,			
exploreCategory, parentCategory) to extract (Definition, Title)			
-Append all (Article IDs) in one Article IDs list			
-For each ID in Article IDs:			
-Process ID with (exploreArticle) to extract Definition and Title			
-Process ID with (suggest) to extract suggested Categories IDs			
-For each suggested Category ID:			
-Process ID with (exploreCategory) to extract Definition and Title			



## 4.3 Dynamic interactive knowledge graphs

For each learning resource we select the top five main topics retrieved by the algorithm. After assigning those main topics to the learning resource we will establish the base for the dynamic interactive knowledge graph that will contain all the learning resources. We insert each learning resource and main topics as nodes, then we create a link between each main topic and all related learning resources. The result is a connected graph of related learning objects as shown in figures 4.2, 4.3, 4.4, 4.5.



Figure 4.2: Relation between learning resource (in blue), and main topics (in yellow)



Figure 4.3: Relation between learning resources (in blue), and main topics (in yellow)

In figures 4.2 and 4.3 we can see the relation between learning resources (in blue), and main topics (in yellow).

In figure 4.4 we can see the relation between one learning resource (in blue), learning resource's chapters (in yellow), and main topics (in red). To obtain this graph we applied our algorithm separately on each learning resource and each chapter, and then we augmented all results in one graph.



Figure 4.4: Relation between one learning resource (blue), main topics (red), and chapters (yellow)

In figure 4.5 we can see the relation between learning resources (in blue), main topics (in yellow), and chapters (in red) but with more learning resources.



Figure 4.5: Relation between learning resources (blue), main topics (yellow), and chapters (red)

We provide all generated graphs with navigation tools for students to ease the exploration process, where they can zoom in and out, move the graph or the nodes, and rotate around the graph.

Learners can use our generated graphs in different useful ways such as:

- Finding the required main topic and browsing all related learning objects.
- Finding a learning object then identifying all related main topics.
- Or finding related learning objects and exploring common main topics between them.

Building knowledge graphs from the text in learning resources is another key feature in our platform, where we can focus on self-directed learning to enable students to identify the required topic or learning resource easily and in a joyful way. With this information and the topics our algorithm extracts, the system will be able to recommend learning resources or complete learning paths for other students with the same goals or learning background.

After we have explained our methodology and the structure of our topic identification algorithm, we are going to illustrate the experiments we did to evaluate it in the next chapter.

# 5. Experiments and Results

To evaluate our algorithm we conducted two experiments:

- In the first experiment we ran our algorithm on a dataset of 579 books with different configurations of parent category level, in order to analyze its accuracy and select the optimal value for that parameter.
- In the second experiment we compared the accuracy of our algorithm's results against another topic extraction technique, and we also compared computation time for both of them.

These experiments were run on a collection of 579 books in different subjects, levels, and from different learning repositories. We selected them for the levels of secondary school and first years of university/college.

The topics of the books included: Anatomy, Art, Astronomy, Automotive, Biology, Biostatistics, Business, Chemistry, Computational Physics, Computing, Economics, Education, Engineering, First Aid, Hardware, History, Humanities, Languages, Law, Math, Medicine, Organic Chemistry, Parenting, Physics, Physics and Environment, Psychology, Science, Social Sciences, Software, Statistics and Travel.

## 5.1 Experiment 1

We extracted the results from the algorithm in four different levels (1st, 2nd, 3rd, and 4th) depending on the number of parent categories used in Wikipedia Miner [**336**]. We used top-5 accuracy [**371**] to evaluate the results in each level to identify the most suitable level to use.

For each learning resource we retrieved the top five main topics, and then we evaluated each topic manually and gave each learning resource a score out of five. If the resource got five of five then all the five topics were correct. However, if the resource got two of five, then only two of the topics were correct. Next, we summed all the results and divided the total by the number of resources to find the average score (average true value).

Figure 5.1 shows, for each parent category level, the number of resources for which the algorithm got the five topics correct, four topics correct, etc. For example, with the 1st level parent category we got 77 results with zeros and 305 results with fives. Nonetheless, with the 4th level parent category we got 67 results with zeros and 77 results with fives.



Figure 5.1: Comparison between the four levels of parent-category

Table 5.1 lists, for each parent category level, the percentages of books for which at least one of the five topics was correct ("True topic identified"), the percentage of books for which none of the five topics was correct ("False topic identified") and the average number of correctly identified topics ("Average true", a value from 0 to 5).

For example, with the 1st level parent category we got 86.7% of the books with true topic identified, 13.3% of the books with false topic identified, and an average of 3,76 correctly identified topics out of 5. Nonetheless, with the 4th level parent category we got 88.4% of the books with true topic identified, 11.6% of the books with false topic identified, and an average of 2.64 correctly identified topics out of 5.

	True Topic Identified	False Topic Identified	Average True
4th level Parent-Category	88.4%	11.6%	2.64 of 5
3rd level Parent-Category	94.6%	5.4%	3.20 of 5
2nd level Parent-Category	95.9%	4.1%	3.57 of 5
1st level Parent-Category	86.7%	13.3%	3.76 of 5

Table 5.1: Comparison between the four levels

We can find that the results of using both the first and second level parent categories show a high number of books with a five of five in top-5 accuracy. In addition, the average number of correctly identified topics is higher (3.76 out of 5) for the first level than for the second level. Therefore, we chose the first level of the parent category in our algorithm for the rest of the experiments, although we retain the ability to use many levels to produce a hierarchical structure of main topics.

## 5.2 Experiment 2

The goal of this experiment is to compare the computation time and the accuracy of our algorithm against other topic identification techniques. As **Gensim** [359] and Wikipedia Miner [336] have the ability to identify topics from text, we compared them against our algorithm, which is built on top of Gensim and Wikipedia Miner as was explained at chapter 4.

In this experiment we sampled **350 books** from the dataset that we used in the first experiment. The books in this sample of the dataset contain between 95 and 1800 pages. In figure 5.2 we can see the distribution of the number of pages of the books in the dataset. For example, we have 134 books with a number of pages between 200 and 400.



Figure 5.2: Number of pages vs. number of books

At the beginning of our experiment we thought that the bigger the number of pages of the book, the longer the processing time it would take. However, we have seen that computation time correlates better with text size (in terms of number of characters) than with the number of pages of a book. That happens because we are processing text, and some books contain a big number of pages filled with charts, figures, and pictures, which are not processed by the algorithms. The histogram of the number of characters per book is shown in figure 5.3. We have 187 books with a number of characters of almost 1 million, 118 books with a number of characters between 1 to 2 million, and so on.



Figure 5.3: Text size vs. number of books

When we started our comparison we found that the maximum number of pages that we can process with Wikipedia Miner is between 25 and 35 pages. With more pages the Wikipedia Miner server frequently stopped responding. This circumstance forced us not to use Wikipedia Miner in this experiment. On the other hand, Gensim had no problem with the number of pages, so we continued our experiment comparing Gensim alone against our algorithm.

An earlier version of our algorithm was based only on Gensim and the "Wikify" Wikipedia Miner service. However, it did not produce any topic suggestion for a big portion of the books in the dataset, more than 200 books out of 350. As a result of that, the final version of our algorithm uses, in addition, the "Search" Wikipedia Miner service for the cases in which the "Wikify" service does not produce any result. The final version of the algorithm produced suggestions for all but 11 books in the dataset.

### 5.2.1 Computation time

First, we compared computation time for both Gensim and our algorithm, taking text size into account.

In figure 5.4 we have a dot plot that presents a data point for each book, identifying the relation between its number of characters and its computation time is seconds, with blue dots for our algorithm and red dots for Gensim. For example, we have for a book with around 5.4 million characters, Gensim processing time was 85.2 seconds, our algorithm processing time was 88.4 seconds.



Figure 5.4: Computation time vs. text size

And in figure 5.5 we group books by ranges of number of characters and show the average computation time for the books at each range. For example, the books with 4 to 5 million characters were processed with Gensim in 65.1 seconds and, with our algorithm, in 66.7 seconds.



Figure 5.5: Average computation time vs. text size

Figures 5.6 and 5.7 show the same (text size vs. computation time) data in the form of box plots, where minimum, maximum and median values for each range, as well as their 25 and 75 percentiles, can be identified. For example, in figure 11 the books that have 3 to 4 million characters with Gensim have a minimum of 40.6 seconds, median of 51.7 seconds, and maximum of 53.1 seconds of processing time. However in figure 12 the books that have 3 to 4 million characters with our algorithm have a minimum of 41.9 seconds, median of 49.1 seconds, and maximum of 49.4 seconds of processing time.



Figure 5.6: Gensim (computation time vs. text size)



Figure 5.7: Our algorithm (computation time vs. text size)

Table 5.2 presents the total **computation time** for the 350 books in the dataset and the average computation time per book for both Gensim and our algorithm.

	Total Time (s)	Average 7 (s)	Гіте
Gensim	5721	16.35	
Our algorithm	6739	19.25	

Table 5.2: Total computation time

We can see in the plots that our algorithm takes slightly more time than Gensim, which was expected as our algorithm runs Gensim internally. The overhead is due to the use of Wikipedia Miner services on the results provided by Gensim, including communication time caused by the running of a different server. In spite of these extra tasks, the average overhead of our algorithm is 17.7%, lower than 3 seconds per book.

After having analyzed computation times and their relation to text size, we will analyze the **accuracy** of the results provided by Gensim and our algorithm.

### 5.2.2 Accuracy

First, we present an example of the results that we got for three books we sampled from the dataset in table 5.3.

Book Information	Title: Principles of Accounting, Volume 1: Financial Accounting Pages: 1055 Category: Business	Title: College Physics for AP® Courses Pages: 1694 Category: Physics	Title: Anatomy & Physiology Pages: 1420 Category: Anatomy
Wikipedia miner	Fails to provide an answer	Fails to provide an answer	Fails to provide an answer
Gensim	Companies/company/ accounting/accountants/ account/accounts/ accountant/accountable/ accountancy/accounted	Credit/credited/credits/ figures/figure/figured/ figuring/energy/energies/ force	Figure/figures/figurative/ cell/cells/called/calling/ muscle/muscles/blood
Our Algorithm	Accountancy/Investment Financial markets/Accounting systems/Management/ Applied sciences/Business/ Business economics/Accountability /Legal entities	Physics/Fundamental physics/concepts/Concept s by field/Introductory physics/Physics education/Motion/Space Time / Force/ Phenomena	Anatomy /Biology /Tissues /Organs /Musculoskeletal system /Subjects taught in medical school /Cardiovascular system /Circulatory system /Medical specialties /Greek loanwords

Table	5.3:	Example	of results
-------	------	---------	------------

We can see from the results that, at least for these examples, our algorithm provides the most relevant and accurate results. In addition, whereas Gensim produces just words, our algorithm produces unambiguous Wikipedia categories, referenced by their unique identifier at Wikipedia. The identifiers are not shown in the table for readability.

In order to evaluate accuracy, in this experiment we made Gensim and our algorithm produce 10 topics for each book. Then, for each book, we evaluated these 10 results and computed accuracy (number of correct topics divided by 10). Topic correctness was decided by a human evaluator. We can see in table 5.4 that our system achieves a much better accuracy than using Gensim alone.

	p io accuracy	
	Average	
	accuracy	
Gensim	0.240857143	
<b>Our Algorithm</b>	0.712285714	

Table	5.4:	Top-10	accuracy
-------	------	--------	----------

Then we evaluated these topics by using top-k accuracy [**371**]. We compared top-1, top-5, and top-10 accuracy in figure 5.8 and in Table 5.5. For each book we evaluated the results according to the three levels. If the first topic returned by the algorithms is the main category of the book or close enough to it, then the book counts for top-1 accuracy. If one of the first 5 topics is the main category or close enough, then the book counts for top-5 accuracy. Likewise, if one of the

first 10 topics is the main category or close enough, the book counts for top-10 accuracy. Again, a human evaluator decided that book by book.

Table 5.5: Top-k accuracy.			
	TOP-1	TOP-5	<b>TOP-10</b>
Gensim	22.9%	45.5%	67.5%
Our algorithm	48.0%	74.0%	86.0%

T.1.1. F.F. T. 1



Figure 5.8: Top-K accuracy

We can see from the results that our algorithm has a higher value for all accuracy levels, and especially for top-1 accuracy.

### 5.2.3 Accuracy vs. text size

Now we want to analyze the relation between accuracy and text size in order to find how our algorithm is affected by the amount of information processed at a time. We start by plotting all accuracy values for all books against text size in figure 5.9. From the results, we can clearly see that higher accuracy values are produced by our algorithm over Gensim for all different text sizes.



Figure 5.9: Accuracy vs. text size (all)

Next, we present top 1, top 5, and top 10 values for Gensim and our algorithms, in groups of different text sizes in figure 5.10. From the results in this graph we can see that our algorithm's average accuracy for all levels is higher than Gensem's. We can see in this graph that there are 0-sized bars with text sizes between 5 and 7 million characters. The reason is the small number of books these groups contain, as we can see in figure 8 (Text size vs. number of books).



Figure 5.10: Average accuracy vs. text size (detailed)

Finally, the average accuracy values are calculated and compared to groups of text sizes in figure 5.11.


Figure 5.11: Average accuracy vs. text size

As the results show, our algorithm, with a small computation time overhead with respect to Gensim, provides much better results in terms of accuracy, and those accuracy values are consistent for all text sizes.

## 6. Conclusion and future work

In this thesis we designed an e-learning system that uses semantic information and cloud computing technologies, combined with other technologies, to enhance the learning process and provide a variety of learning tools and services.

Before we started the design process we explored the research field and studied many examples on different topics, including education, learning, e-learning, semantic information, recommendation systems, learning object annotation, cloud computing, and other topics.

We also developed an algorithm to process learning objects in order to extract main topics, where we start by extracting text, then we use Gensim to extract the main keywords from the text, and finally we use Wikipedia Miner with both the Wikify and Search services to extract main topics. Different levels of Wikipedia parent categories were possible, so we conducted an experiment to find out the best one to use in our algorithm. From the main topics extracted by our algorithm we produced interactive knowledge graphs to facilitate finding the required main topic, browsing all related learning objects, finding a specific learning object, identifying all related main topics, finding related learning objects, and exploring common main topics between learning objects.

To evaluate our algorithm we made two experiments. In experiment 1 we produced results with our algorithm for different levels of Wikipedia parent category. We found that the first level is the best choice to use. However, we can use different levels if a hierarchical topic structure is needed.

In experiment 2 we proved that the Wikify service of Wikipedia Miner does not work properly for the books in our dataset, as it does not produce results for more than 57% of the books. On the other hand, with our algorithm this happens for just 3% of them.

We also analyzed the computation time for both our algorithm and Gensim. As expected, our algorithm takes slightly more time than Gensim, due to it using Wikipedia Miner services on the results provided by Gensim. What is important to mention is that the experiment shows that the relationship between the computation time of our algorithm and text size is linear.

Finally, we showed that our algorithm largely outperforms Gensim in terms of the accuracy of the topics it extracts, with the additional advantage of producing them in an unambiguous way as references to Wikipedia categories. According to our experiments, text size does not affect the accuracy of the topics our algorithm extracts.

We can conclude that the main advantage of the algorithm we propose, in comparison to previous work, is its ability to work on large learning resources such as big-sized books that normally contain hundreds of pages and even reach two thousand pages in some cases.

## 6.1 Research Restrictions and Practical Implications

So far, we have applied our algorithm on textual educational resources only. Recent advances in speech-to-text techniques suggest that it could also probably work for extracting the main topics of other kinds of learning resources such as videos and audio. However, further research is needed in order to prove that.

Applying our algorithm requires the computational power of a high-end computer or a computer cluster, especially for running the Wikipedia Miner services, which work on a massive database of information extracted from Wikipedia. Since this task is intended to be run at the core of the e-learning system, this limitation would not affect learners, who would be able to use the system from the device of their choice.

By embedding the algorithm we proposed into the design of our e-learning system we can provide a new system that will help evolving many services such as learning object recommendation, topic identification, learning object enrichment, learning group suggestions, knowledge extraction, and dynamic interactive knowledge graphs.

By implementing our design we can provide learning object reusability and sharing amount the different e-learning platforms, reducing the required time and cost for generating new resources and make use of the massive amount of semantic information to enrich current learning resources and connect different concepts from other resources to produce a connected network of new dynamic knowledge graphs.

## 6.2 Future work

One line of future work is continuing with the implementation of the whole system we have designed in this research. This is a complex task that includes several research challenges.

We need to start by separating the implementation phase into three specializations: cloud computing and networks, semantic web and ontology engineering, data mining and data analysis. Within each one a qualified team of programmers have to develop the required tasks in such a way that makes all the processes work smoothly between the different specializations. The coordination between the development team is a challenging task and providing the required financial resources is also vital. Regarding the research challenges, we must follow up with the state of the art in all the involved technologies, and this requires another team to keep our system up to date.

After finishing the implementation phase, the testing phase starts by utilizing the system with real educational institutions, real students, and real teachers, where we must keep eye on every process and every learning object to make sure that the system complies with its requirements. Convincing educational institutions and students to use a new e-learning system could be challenging because they are on a specific schedule that they cannot postpone or change.

To enhance and develop our system, we can integrate other tools and technologies to analyze learning resources in different formats like video, audio, and graphics. This could produce a strong system for e-learning that will beat the huge amount of learning resources currently available and help learners find the resources they need. We will complete the designed system by providing the required tools for augmenting it inside different social networks that can encourage learners to spread their learning experience among their friends, and make use of the long times that they spend socializing to enhance the learners ontology database and collecting information about learners interests background, and goals. Another area to enhance our system is the expansion of the number of languages that it could handle for topic identification and learning object handling. The ability to handle multiple languages will, without any doubt, be a great advantage through which a bigger number of users could be acquired and a larger number of learning resources could be processed. This increase will enrich the scientific content, expand the circle of interaction between students from different cultures, and increase the knowledge base of the system to improve the quality of recommendations received by users.

Certainly, the great development in the field of artificial intelligence is not hidden from us, which can be used in the development of all aspects. We can use it to improve the quality of recommendations and extract main topics from learning resources, whatever their type or language. We can use it to analyze the behavior of students during their use of the system to find the appropriate learning model for them and give them suggestions for learning groups that suit their interests. We can also improve the search process for learning resources by developing an AI service supported by the knowledge base extracted from our system, based on semantic information, that can be understood by the computer, and then interactive methods could be provided for this services through text, audio or even video, providing and interactive electronic guidance or tutor for each student that could be adapted and tailored according to his own needs and requirements.

## References

- [1] (2022, June) worldpopulationreview.com-education-rankings-by-country. [Online]. https://worldpopulationreview.com/country-rankings/education-rankings-by-country
- [2] Clàudia Canals. (2022, June) caixabankresearch.com. [Online].
  https://www.caixabankresearch.com/en/economics-markets/labour-marketdemographics/education-and-economic-growth
- [3] (2023, Feb) wikipedia. [Online]. https://en.wikipedia.org/wiki/Spanish\_Civil\_War
- [4] (2022, July) wikipedia.org. [Online]. https://en.wikipedia.org/wiki/Meiji\_(era)
- [5] (2022, June) factsanddetails.com/japan. [Online]. https://factsanddetails.com/japan/cat23/sub150/entry-2794.html#:~:text=After%20World%20War%20II%2C%20Japanese,direction%20of%20th e%20Occupation%20forces
- [6] britannica.com. (2023, Feb) britannica.com. [Online].
  https://www.britannica.com/topic/education/Education-after-World-War-II
- [7] (2022, June) globalpartnership.org. [Online]. https://www.globalpartnership.org/benefitsof-education
- [8] Ajay Bhardwaj, "Importance of education in human life: A holistic approach," *International Journal of Science and Consciousness 2.2*, vol. 2, no. 2, pp. 23-28, 2016.
- [9] (2022, June) worldvision.ca/stories/education/why-is-education-important. [Online]. https://www.worldvision.ca/stories/education/why-is-education-important
- [10] Cameron Stuart Cameron John. (2005) unesco.org/The Economic benefits of increased literacy. [Online]. https://unesdoc.unesco.org/ark:/48223/pf0000145957
- [11] (2013, Oct.) Statistics Canada. [Online]. https://www150.statcan.gc.ca/n1/en/catalogue/89-555-X2013001
- [12] (2022, June) globalpartnership.org/benefits-of-education. [Online]. https://www.globalpartnership.org/benefits-of-education
- [13] (2022, June) worldpopulationreview.com. [Online].
  https://worldpopulationreview.com/country-rankings/most-educated-countries
- [14] M Woodhall George Psacharopoulos, *Education for development*.: Citeseer, 1993.
- [15] (2022, June) unesdoc.unesco.org. [Online].ttps://unesdoc.unesco.org/ark:/48223/pf0000190214

- [16] C Brannen, R Glennerster M Kremer, "The challenge of education and learning in the developing world," *Science*, pp. 297-300, 2013.
- [17] Preeti. Tarkar, "Impact of COVID-19 pandemic on education system," *International Journal of Advanced Science and Technology*, pp. 3812-3814, 2020.
- [18] (2022, June) worldbank.org. [Online]. https://www.worldbank.org/en/news/feature/2018/01/30/the-changing-wealth-ofnations-2018
- [19] Lawrence H., M. Ali Khan, and Richard H. Sabot. Summers, "Investing in all the people [with comments]," *The Pakistan development review*, vol. 31, no. 4, pp. 367-404, 1992.
- [20] (2022, June) cgdev.org/publication/education-and-developing-world. [Online]. https://www.cgdev.org/publication/education-and-developing-world
- [21] (2022, June) nobelprize.org. [Online].
  https://www.nobelprize.org/prizes/physics/1903/marie-curie/facts/
- [22] (2023, Fep) BBC. [Online]. https://www.bbc.com/mundo/noticias-59172486
- [23] (2023, Feb) wikipedia. [Online]. https://en.wikipedia.org/wiki/Ir%C3%A8ne\_Joliot-Curie
- [24] Michio. Kaku. (2022, Apr) Encyclopedia Britannica. [Online]. https://www.britannica.com/biography/Albert-Einstein
- [25] (2022, June) scopus.com/elearning1990-2022. [Online]. https://www.scopus.com/term/analyzer.uri?sid=15fb5ed34546346a7a0fb532772f1a8c&ori gin=resultslist&src=s&s=TITLE-ABS-KEY%28e-learning%29&sort=plff&sdt=b&sot=b&sl=25&count=97191&analyzeResults=Analyze+results&txGid=3485fdea f33acd61a2f4f51bc95fd6eb
- [26] (2022, June) scopus.com/education19902022. [Online]. https://www.scopus.com/term/analyzer.uri?sid=535b23787ffb445798912faaffee71f5&origi n=resultslist&src=s&s=TITLE-ABS-KEY%28education%29&sort=plff&sdt=b&sot=b&sl=24&count=2321113&analyzeResults=Analyze+results&txGid=e7adc4 32f2eebefd52e4b08be9ee2a71
- [27] (2022, november) Scopus. [Online]. https://www.scopus.com/term/analyzer.uri?sid=15fb5ed34546346a7a0fb532772f1a8c&ori gin=resultslist&src=s&s=TITLE-ABS-KEY%28e-learning%29&sort=plff&sdt=b&sot=b&sl=25&count=97191&analyzeResults=Analyze+results&txGid=3485fdea f33acd61a2f4f51bc95fd6eb
- [28] (2022, November) Scoupus. [Online]. https://www.scopus.com/term/analyzer.uri?sid=535b23787ffb445798912faaffee71f5&origi n=resultslist&src=s&s=TITLE-ABS-KEY%28education%29&sort=plff&sdt=b&sot=b&sl=24&count=2321113&analyzeResults=Analyze+results&txGid=e7adc4 32f2eebefd52e4b08be9ee2a71

- [29] (2022, November) Scopus. [Online]. https://www.scopus.com/term/analyzer.uri?sid=15fb5ed34546346a7a0fb532772f1a8c&ori gin=resultslist&src=s&s=TITLE-ABS-KEY%28e-learning%29&sort=plff&sdt=b&sot=b&sl=25&count=97191&analyzeResults=Analyze+results&txGid=3485fdea f33acd61a2f4f51bc95fd6eb
- [30] (2022, November) Scopus. [Online]. https://www.scopus.com/term/analyzer.uri?sid=535b23787ffb445798912faaffee71f5&origi n=resultslist&src=s&s=TITLE-ABS-KEY%28education%29&sort=plff&sdt=b&sot=b&sl=24&count=2321113&analyzeResults=Analyze+results&txGid=e7adc4 32f2eebefd52e4b08be9ee2a71
- [31] (2023, Feb) wikipedia. [Online]. https://en.wikipedia.org/wiki/Education
- [32] (2022, June) pediaa.com. [Online]. https://pediaa.com/difference-between-education-and-learning/
- [33] (2022, June) keydifferences.com. [Online]. https://keydifferences.com/difference-betweenlearning-andeducation.html#:~:text=Learning%20refers%20to%20an%20intellectual,providing%20kno wledge%2C%20through%20systematic%20instruction.&text=To%20regulate%20future%2 0actions%20and%20develop%20new%20knowl
- [34] (2023, Feb) wikipedia. [Online]. https://en.wikipedia.org/wiki/Learning
- [35] R. Gross, *Psychology: The science of mind and behaviour 6th Edition*, 6th ed. Hachette , UK: Hodder Education, 2012.
- [36] R. M., & Soloman, B. A. Felder. (2000) Learning styles and strategies. [Online]. https://www.andrews.edu/services/ctcenter/career-center/learning-stylesstrategies/learning-styles-and-strategies.pdf
- [37] Denis, and Jonas F. Soltis Phillips, "Perspectives on learning," in *Perspectives on learning*.: Teachers College Press, 2015, ch. 2.
- [38] Christine, and Penny Mukherji. Brain, "Understanding child psychology," in *Understanding child psychology.*, 2005, ch. 4.
- [39] Albert. Bandura, Social learning theory.: Morristown, 1971.
- [40] C. George. Boeree, Gestalt psychology.: Webspace. ship. edu., 2000.
- [41] George, Michael Klobuchar, and David Geelan. Bodner, The many forms of constructivism., 2001.
- [42] Edward W. Taylor, "Transformative learning theory," in *Transformative learning theory.*, 2018, ch. 1, pp. 301-320.
- [43] Mohamed Amine. Chatti, "Personalization in technology enhanced learning: A social software perspective," *RWTH Aachen*, 2010.

- [44] Georg. Siemens, "Connectivism: A learning theory for the digital age," International Journal of Instructional Technology and Distance Learning, 2005.
- [45] Stephen. Downes, "New technology supporting informal learning," *Journal of emerging technologies in web intelligence*, pp. 27-33, 2010.
- [46] Lone, Chris Jones, and Berner Lindström. Dirckinck-Holmfeld, *Analysing networked learning* practices in higher education and continuing professional development.: BRILL, 2009.
- [47] Fisteus JA, Mahmoud TM, Abd El-Hafeez T. Badawy A, "Topic Extraction and Interactive Knowledge Graphs for Learning Resources," *Sustainability*, vol. 14, no. 1, p. 226, Dec 26 2021.
- [48] H., Weber, C., Zenkert, J., Krumm, R., & Fathi, M. Abu-Rasheed, "Explainable Graph-Based Search for Lessons-Learned Documents in the Semiconductor Industry," *Intelligent Computing. Springer, Cham*, pp. 1097-1106, 2022.
- [49] Sigmund, J. Dexter Fletcher, and Alexander P. Wind. Tobias, "Game-based learning," Springer Science+Business Media, vol. 5, no. 38, pp. 485-503, 2014.
- [50] K. D., Abdullah, N. S., Ahmad, N., & Sahmin, S. Tania, "Continued Usage of E-Learning: A Systematic Literature Review," *Journal of Information Technology Management*, vol. 14, no. 5, pp. 245-254, 2022.
- [51] Michael G., and Greg Kearsley Moore, *Distance education: A systems view of online learning*.: Cengage Learning, 2011.
- [52] Michael G., and Greg Kearsley Moore, *Distance education: A systems view of online learning.*: Cengage Learning, 2011.
- [53] Daniel Y., and Yi-Shun Wang. Shee, "Multi-criteria evaluation of the web-based e-learning system: A methodology based on learner satisfaction and its applications," *Computers & Education*, vol. 50, no. 3, pp. 894-905, 2008.
- [54] Kamla Ali, and Hafedh Al-Shihi. Al-Busaidi, "Key factors to instructors' satisfaction of learning management systems in blended learning," *Journal of Computing in Higher Education*, vol. 24, no. 1, pp. 18-39, 2012.
- [55] Ken Ditha, et al. Tania, "Continued Usage of E-Learning: A Systematic Literature Review," Journal of Information Technology Management 14.5th International Conference of Reliable Information and Communication Technology (IRICT), pp. 245-254, 2022.
- [56] Kadriye O., et al. Lewis, "Leveraging e-learning in medical education," *Current problems in pediatric and adolescent health care*, vol. 44, no. 6, pp. 150-163, 2014.
- [57] Snigdha, and Snigdha Pattnaik Choudhury, "Emerging themes in e-learning: A review from the stakeholders' perspective," *Computers & Education*, vol. 144, 2020.
- [58] Erin Duffin. (2021, Sep.) statista.com. [Online].
  https://www.statista.com/statistics/1076776/us-k-12-students-reasons-using-digital-

learning-tools-level/

- [59] L. Yoon. (2021, Mar.) statista.com. [Online].
  https://www.statista.com/statistics/710512/south-korea-number-of-e-books-academic-libraries/
- [60] Amy Watson. (2021, Aug.) statista.com. [Online]. https://www.statista.com/statistics/652610/number-of-new-primary-school-booksreleased-in-norway/
- [61] the Babson Survey Research Group. (2016) learningconsortium.org. [Online]. https://onlinelearningconsortium.org/news\_item/report-one-four-students-enrolled-onlinecourses/
- [62] I. E., & Seaman, J. Allen, *Changing course: Ten years of tracking online education in the United States*, 01950th ed. Newburyport: PO Box 1238, 2013.
- [63] Hope Kentnor. (2015) digitalcommons.du.edu. [Online]. https://digitalcommons.du.edu/cgi/viewcontent.cgi?article=1026&context=law\_facpub
- [64] H. E. Kentnor, "Distance education and the evolution of online learning in the United States," *Curriculum and teaching dialogue*, vol. 17, no. 1, pp. 21-34, jan 2015.
- [65] J. L., Dickson-Deane, C., & Galyen, K. Moore, "e-Learning, online learning, and distance learning environments: Are they the same?," *The Internet and higher education*, vol. 14, no. 2, pp. 129-135, Mar 2011.
- [66] Institute of Education Sciences USA. (2020) National Center for Education Statics nces.ed.gov. [Online]. https://nces.ed.gov/fastfacts/display.asp?id=80
- [67] Vishal Dineshkumar. Soni, "Global Impact of E-learning during COVID 19," SSRN, 2020.
- [68] John McCarthy, "What is artificial intelligence," *Computer Science Department Stanford University*, pp. 1-15, Nov 2007.
- [69] Ajay K., Joshua S. Gans, and Avi Goldfarb. Agrawal, "What to expect from artificial intelligence," *MIT Sloan Management Review*, 2017.
- [70] Elizabeth D. Liddy, "Natural language processing," in Natural language processing., 2001.
- [71] Tim, James Hendler, and Ora Lassila. Berners-Lee, "The semantic web," *Scientific american*, pp. 34-43, 2001.
- [72] J. M., K. W. Chan, and Ian Gibson. Zheng, "Virtual reality," *Ieee Potentials*, pp. 20-23, 1998.
- [73] Thomas J. Shuell, "Cognitive conceptions of learning," Review of educational research, vol. 56, no. 4, pp. 411-36, Dec 1986.
- [74] E. K., Ekenberg, L., Hansson, H., Danielson, F. T., & Danielson, M. Kahiigi, "Exploring the

e-Learning State of Art," Electronic Journal of e-learning, vol. 6, no. 2, pp. 149-160, Apr 2008.

- [75] K. D., Abdullah, N. S., Ahmad, N., & Sahmin, S. Tania, "Continued Usage of E-Learning: A Systematic Literature Review," *Journal of Information Technology Management*, vol. 14, no. 5, pp. 245-254, Jan 2022.
- [76] D., Joy, M., & Sinclair, J. Al-Fraihat, "Evaluating E-learning systems success: An empirical study," *Computers in human behavior*, vol. 102, pp. 67-86, Jan 2020.
- [77] H., Teng, B. K., Alzahrani, A. I., & Alalwan, N. Al-Samarraie, "E-learning continuance satisfaction in higher education: a unified perspective from instructors and students," *Studies in higher education*, vol. 43, no. 11, pp. 2003-2019, Nov 2018.
- [78] AKM Najmul, and Nasreen Azad. Islam, "Satisfaction and continuance with a learning management system: Comparing perceptions of educators and students," *The International Journal of Information and Learning Technology*, Mar 2015.
- [79] A., Hassan, R., Aremu, A. Y., Hussain, A., & Lodhi, R. N. Shahzad, "Effects of COVID-19 in E-learning on higher education institution students: the group comparison between male and female," *Quality & quantity*, vol. 55, no. 3, pp. 805-826, Jun 2021.
- [80] F. J., de Lima, L. S. A., Cidade, P. I. M., Nobre, C. B., & Neto, M. L. R. de Oliveira Araújo, "Impact of Sars-Cov-2 and its reverberation in global higher education and mental health," *Psychiatry research*, vol. 288, p. 112977, Jun 2020.
- [81] G., & Chakraborty, D. Dash, "Digital transformation of marketing strategies during a pandemic: Evidence from an emerging economy during COVID-19," *Sustainability*, vol. 13, no. 12, p. 6735, Jun 2021.
- [82] Muhammad Ashraf. Fauzi, "E-learning in higher education institutions during COVID-19 pandemic: current and future trends through bibliometric analysis," *Heliyon*, vol. 16, p. e09433, May 2022.
- [83] (2022, november) Scopus. [Online]. https://www.scopus.com/term/analyzer.uri?sid=46d3a42c478156b105e13b55b20c421a&or igin=resultslist&src=s&s=TITLE-ABS-KEY%28e-learning%29&sort=plff&sdt=b&sot=b&sl=25&count=102640&analyzeResults=Analyze+results&txGid=73527f9 cb9b83e4e44e61755c946f303
- [84] Heejin. Chang, "The development of a learning community in an e-learning environment," *International Journal of Pedagogies and Learning*, vol. 7, no. 2, pp. 154-161, Nov 2012.
- [85] Gökhan, and Buket Akkoyunlu. Dağhan, "Modeling the continuance usage intention of online learning environments," *Computers in Human Behavior*, vol. 60, pp. 198-211, Jul 2016.
- [86] Z., Xiao, L., Van Toorn, C., Lai, Y., & Seo, C. Guo, "Promoting online learners' continuance intention: An integrated flow framework," *Information & Management*, vol. 53, no. 2, pp. 279-295, Mar 2016.

- [87] F. A., Ramos, A. S. M., Gouvêa, M. A., & da Costa, M. F. de Melo Pereira, "Satisfaction and continuous use intention of e-learning service in Brazilian public organizations," *Computers in Human Behavior*, vol. 46, pp. 139-148, May 2015.
- [88] B., & Zhang, C. Wu, "Empirical study on continuance intentions towards E-Learning 2.0 systems," *Behaviour & Information Technology*, vol. 33, no. 10, pp. 1027-1038, Oct 2014.
- [89] M., & Yuen, A. H. K. Cheng, "Student continuance of learning management system use: A longitudinal exploration," *Computers & Education*, vol. 120, pp. 241-253, May 2018.
- [90] K. D., Abdullah, N. S., Ahmad, N., & Sahmin, S. Tania, "Continued Usage of E-Learning: A Systematic Literature Review," *Journal of Information Technology Management*, vol. 14, no. 5, pp. 245-254, Jan 2022.
- [91] Branka, and Nenad Stefanovic. Arsovic, "E-learning based on the adaptive learning model: case study in Serbia," *Sādhanā*, vol. 45, no. 1, pp. 1-13, Dec 2020.
- [92] EDUCAUSE a nonprofit association. (2022, Oct.) library.educause.edu. [Online]. https://library.educause.edu/resources/2017/1/7-things-you-should-knowaboutpersonalized-learning
- [93] Philip. Kerr, "Adaptive learning," Elt Journal, vol. 70, no. 1, pp. 88-93, Jan 2016.
- [94] Declan, Vincent Wade, and Owen Conlan. Dagger, "Towards "anytime, anywhere "learning: The role and realization of dynamic terminal personalization in adaptive elearning.," in *Ed-Media 2003, World Conference on Educational Multimedia, Hypermedia and Telecommunications*, Hawaii, Jun 2003, pp. 32-35.
- [95] Paul, Ad Aerts, Bart Berden, Barend De Lange, Brendan Rousseau, Tomi Santic, David Smits, and Natalia Stash. De Bra, "AHA! The adaptive hypermedia architecture," in Proceedings of the fourteenth ACM conference on Hypertext and hypermedia, Nottingham, 2003, pp. 81-84.
- [96] Dante, Emanuela Ovcin, and Gaetano Morrone Del Corso, "A teacher friendly environment to foster learner-centered customization in the development of interactive educational packages," *IEEE Transactions on Education*, vol. 48, no. 4, pp. 574-579, Nov 2005.
- [97] Meg Coffin, and Jorge Pérez. Murray, "Informing and performing: A study comparingadaptive learning to traditional learning," *Informing Science: The International Journal of* an Emerging Transdicipline, vol. 18, pp. 111-125, 2015.
- [98] Sarvenaz. Hatami, ""Learning styles."," Elt Journal, vol. 67, no. 4, pp. 488-490, 2013.
- [99] Silverman L K Felder Richard M., "Learning and teaching styles in engineering education," *Engineering Education*, vol. 78, pp. 674 – 681, 1988.
- [100] L., Williams, S., Ousmanou, K., & Lubega, J. Sun, "Building personalised functions into dynamic content packaging to support individual learners," in *Proceedings of the 2nd European Conference on e-Learning*, Glasgow, 2003, pp. 439-448.

- [101] RevathyShomona P and Jacob G. Geetharamani R, "Prediction of users' webpage access behavior using association rule mining," *Sadhana*, vol. 40, no. 8, pp. 2353-2365, Dec 2015.
- [102] Herman Dwi. Surjono, "Adaptive E-Learning Model in Learning Personality Characters," International Conference on Online and Blended Learning 2019 (ICOBL 2019). Atlantis Press, vol. 22, pp. 18-22, May 2020.
- [103] Manuela, Fernando Bacao, and Tiago Oliveira. Aparicio, "An e-learning theoretical framework," *Educational Technology & Society*, vol. 19, no. 1, pp. 292-307, 2016.
- [104] J. W You, "Examining the Effect of Academic Procrastination on Achievement Using LMS Data in Elearning.," *Educational Technology & Society*, vol. 18, no. 3, pp. 64–74, 2015.
- [105] Amal, Jieun Kim, and Hokyoung Ryu Al-Dujaily, "Am I extravert or introvert? Considering the personality effect toward e-learning system," *Journal of Educational Technology & Society*, vol. 16, no. 3, pp. 14-27, Jul 2013.
- [106] Howard E Gardner, Frames of mind: The theory of multiple intelligences.: Basic books, 2011.
- [107] M. H., Al-Ahmari, A., Ahmad, A., Ameen, W., & Alkhalefah, H. Abidi, "Assessment of virtual reality-based manufacturing assembly training system," *The International Journal of Advanced Manufacturing Technology*, vol. 105, no. 9, pp. 3743-3759, Dec 2019.
- [108] F., Yildirim, P., Caliklar, Ş., & Yilmaz, R. M. Arici, "Research trends in the use of augmented reality in science education: Content and bibliometric mapping analysis," *Computers & Education*, vol. 142, p. 103647, Dec 2019.
- [109] J. L., DeJaegher, C. J., & Chao, J. Chiu, "The effects of augmented virtual science laboratories on middle school students' understanding of gas properties," *Computers & Education*, vol. 85, pp. 59-73, Jul 2015.
- [110] S., Miyauchi, T., Noguchi, S., & Yamashita, H. Matsutomo, "Real-time visualization system of magnetic field utilizing augmented reality technology for education," *IEEE transactions on magnetics*, vol. 48, no. 2, pp. 531-534, Jan 2012.
- [111] Jon. Peddie, Augmented reality: Where we will all live .: Springer, 2017.
- [112] J. C., Tsai, C. H., & Wu, M. Yen, "Augmented reality in the higher education: Students' science concept learning and academic achievement in astronomy," *Procedia-social and behavioral sciences*, vol. 103, pp. 165-173, Nov 2013.
- [113] T. H., Yang, S. J., & Hwang, G. J. Chiang, "An augmented reality-based mobile learning system to improve students' learning achievements and motivations in natural science inquiry activities," *Journal of Educational Technology & Society*, vol. 17, no. 4, pp. 352-365, Oct 2014.
- [114] L., Luckin, R., Seljeflot, S., & Woolard, A. Kerawalla, ""Making it real": exploring the potential of augmented reality for teaching primary school science," *Virtual reality*, vol. 10, no. 3, pp. 163-174, Dec 2006.

- [115] J., Jiménez-Rodríguez, M. Á., Torralba-Estelles, J., Garzón-Farinós, F., Pérez-Bermejo, M., & Fernández-Ehrling, N. Ferrer-Torregrosa, "Distance learning ects and flipped classroom in the anatomy learning: comparative study of the use of augmented reality, video and notes," *BMC medical education*, vol. 16, no. 1, pp. 1-9, Dec 2016.
- [116] J., Torralba, J., Jimenez, M. A., García, S., & Barcia, J. M. Ferrer-Torregrosa, "ARBOOK: Development and assessment of a tool based on augmented reality for anatomy," *Journal of Science Education and Technology*, vol. 24, no. 1, pp. 119-124, Feb 2015.
- [117] (2021, April) Statista. [Online]. https://www.statista.com/statistics/ 1085930/edtechexpenditure-forecast/
- [118] A. M., Alsulaimani, A. A., Allamnakhrah, A., Alalwan, A. A., Dwivedi, Y. K., & Rana, N. P. Baabdullah, "Usage of augmented reality (AR) and development of e-learning outcomes: An empirical evaluation of students'e-learning experience," *Computers & Education*, vol. 177, p. 104383, Feb 2022.
- [119] M. Y., Ong, S. K., & Nee, A. Y. Zhao, "An augmented reality-assisted therapeutic healthcare exercise system based on bare-hand interaction," *International Journal of Human–Computer Interaction*, vol. 32, no. 9, pp. 708-721, Sep 2016.
- [120] F., & Hagberg, J. Caboni, "Augmented reality in retailing: a review of features, applications and value," *International Journal of Retail & Distribution Management*, Aug 2019.
- [121] J., & Duffy, K. Scholz, "We ARe at home: How augmented reality reshapes mobile marketing and consumer-brand relationships," *Journal of Retailing and Consumer Services*, vol. 44, pp. 11-23, Sep 2018.
- [122] R., & Khoo-Lattimore, C. Yung, "New realities: a systematic literature review on virtual reality and augmented reality in tourism research," *Current issues in tourism*, vol. 22, no. 17, pp. 2056-2081, Oct 2019.
- [123] J., Barna, J., Janak, M., & Novakova-Marcincinova, L. Novak-Marcincin, "Augmented reality aided manufacturing," *Procedia Computer Science*, vol. 25, pp. 23-31, Jan 2013.
- [124] R., Erkoyuncu, J. A., Roy, R., & Torabmostaedi, H. Palmarini, "A systematic review of augmented reality applications in maintenance," *Robotics and Computer-Integrated Manufacturing*, vol. 49, pp. 215-228, Feb 2018.
- [125] S., Wang, X., & Chiang, F. K. Cai, "A case study of Augmented Reality simulation system application in a chemistry course," *Computers in human behavior*, vol. 37, pp. 31-40, Aug 2014.
- [126] M., Dede, C., & Mitchell, R. Dunleavy, "Affordances and limitations of immersive participatory augmented reality simulations for teaching and learning," *Journal of science Education and Technology*, vol. 18, no. 1, pp. 7-22, Fep 2009.
- [127] T., Lagerstam, E., Kärkkäinen, T., & Väänänen-Vainio-Mattila, K. Olsson, "Expected user experience of mobile augmented reality services: a user study in the context of shopping centres," *Personal and ubiquitous computing*, vol. 17, no. 2, pp. 287-304, Fep 2013.

- [128] R., & Cellary, W. Wojciechowski, "Evaluation of learners' attitude toward learning in ARIES augmented reality environments," *Computers & education*, vol. 68, pp. 570-585, Oct 2013.
- [129] K., Ab Aziz, N. A., Yusof, A. M., & Paul, A. Ab Aziz, "Potential for providing augmented reality elements in special education via cloud computing," *Procedia Engineering*, vol. 41, pp. 333-339, Jan 2012.
- [130] R. C., Chung, L. Y., & Huang, Y. M. Chang, "Developing an interactive augmented reality system as a complement to plant education and comparing its effectiveness with video learning," *Interactive Learning Environments*, vol. 24, no. 6, pp. 1245-1264, Aug 2016.
- [131] Y. Li, "Augmented Reality for remote education," in 3rd International Conference on Advanced Computer Theory and Engineering (ICACTE), Chengdu, China, 2010, pp. V3-187.
- [132] S. A., Elinich, K., Wang, J., Steinmeier, C., & Tucker, S. Yoon, "Using augmented reality and knowledge-building scaffolds to improve learning in a science museum," *International Journal* of Computer-Supported Collaborative Learning, vol. 7, no. 4, pp. 519-541, Dec 2012.
- [133] K. F., Chen, N. S., & Huang, S. Y. Hsiao, "Learning while exercising for science education in augmented reality among adolescents," *Interactive learning environments*, vol. 20, no. 4, pp. 331-349, Aug 2012.
- [134] H. K., Lee, S. W. Y., Chang, H. Y., & Liang, J. C. Wu, "Current status, opportunities and challenges of augmented reality in education," *Computers & education*, vol. 62, pp. 41-49, Mar 2013.
- [135] (2022, november) Scopus. [Online]. https://www.scopus.com/term/analyzer.uri?sid=c3e2a73de684babf719b123b375b7bbb&or igin=resultslist&src=s&s=TITLE-ABS-KEY%28semantic+web%29&sort=plff&sdt=b&sot=b&sl=27&count=61011&analyzeResults=Analyze+results&txGid=0e79a2f2 c7e237f4b7610360b353efcc
- [136] (2022, November) Scopus. [Online]. https://www.scopus.com/term/analyzer.uri?sid=c1e4f9812043dd5a5651c58fbb5122b6&ori gin=resultslist&src=s&s=TITLE-ABS-KEY%28ontology%29&sort=plff&sdt=b&sot=b&sl=23&count=165268&analyzeResults=Analyze+results&txGid=23ab2d 73b6831d1cb0ac01211d9eec50
- [137] V. K., & Kuechler Jr, W. L. Vaishnavi, "An approach to reducing e-commerce coordination costs in evolving markets using a semantic routing protocol," J. Electron. Commer. Res., vol. 4, no. 3, pp. 113-127, 2003.
- [138] S., Abhishek, K., Cohen, J. P., Cohen-Adad, J., & Hamarneh, G. Asgari Taghanaki, "Deep semantic segmentation of natural and medical images: a review," *Artificial Intelligence Review*, vol. 54, pp. 137-178, Jan 2021.
- [139] A., Sassi, S., & Faiz, S. Dridi, "Towards a semantic medical internet of things," in 2017 IEEE/ACS 14th International Conference on Computer Systems and Applications (AICCSA), Hammamet, Tunisia, 2017 October, pp. 1421-1428.

- [140] M. L., & Barnett, G. A. Doerfel, "A semantic network analysis of the International Communication Association," *Human communication research*, vol. 25, no. 4, pp. 589-603, Jun 1999.
- [141] A., Latif, K., Ahmad, H. F., Hur, A., Anwar, Z., & Bloodsworth, P. C. Razzaq, "Semantic security against web application attacks," *Information Sciences*, vol. 254, pp. 19-38, Jan 2014.
- [142] S., Kulkarni, M., Nayak, K., & Dumitraş, T. Indela, "Toward semantic cryptography APIs," in IEEE Cybersecurity Development (SecDev 2016), Boston, USA, November 2016, pp. 9-14.
- [143] (2022, November) Scopus. [Online]. https://www.scopus.com/term/analyzer.uri?sid=8965ace21f816f597e6d586e1c31a6b0&ori gin=resultslist&src=s&s=TITLE-ABS-KEY%28semantic+e-learning%29&sort=plff&sdt=b&sot=b&sl=34&count=2648&analyzeResults=Analyze+results&txGid=0fb7348fb 8ea25b7040ebd7aa3bf68b3
- [144] W., Webster, D., Wood, D., & Ishaya, T. Huang, "An intelligent semantic e-learning framework using context-aware Semantic Web technologies," *British Journal of Educational Technology*, vol. 37, no. 3, pp. 351-373, May 2006.
- [145] T., & Whitelock, D. Anderson, "The educational semantic web: Visioning and practicing the future of education," *Journal of interactive Media in Education*, vol. 1, pp. 1-15, May 2004.
- [146] M. Nilsson, "The semantic web: How RDF will change learning technology standards," in *The semantic web: How RDF will change learning technology standards*, Centre for User Oriented IT Design CID, Ed. Stockhom, , Sweden: NADA, Deptartment of Numerical Analysis and Computer Science, 2001, ch. 1, pp. 3-9.
- [147] N., Dolog, P., & Nejdl, W. Henze, "Reasoning and ontologies for personalized e-learning in the semantic web," *Journal of Educational Technology & Society*, vol. 7, no. 4, pp. 82-97, Oct 2004.
- [148] G., Gašević, D., & Devedžić, V. Šimić, "Semantic web and intelligent learning management systems," in *Workshop on applications of semantic web technologies for e-learning*, 2004.
- [149] A., Rogushina, J., Garcı, F., Martínez-Béjar, R., & Fernández-Breis, J. T. Gladun, "An application of intelligent techniques and semantic web technologies in e-learning environments," *Expert Systems with Applications*, vol. 36, no. 2, pp. 1922-1931, Mar 2009.
- [150] E. Junuz, "Preparation of the learning content for semantic e-learning environment," *Procedia-Social and Behavioral Sciences*, vol. 1, no. 1, pp. 824-828, Jan 2009.
- [151] L., & Maode, D. Rui, "A research on e-learning resources construction based on semantic web," *Physics Procedia*, vol. 25, pp. 1715-1719, Jan 1715-1719.
- [152] B. Dutta, "Semantic web based e-learning," in DRTC Conference on ICT for Digital Learning Environment- Documentation Research and Training Centre Indian Statistical Institute, Bangalore, 2006, pp. 2-14.

- [153] T., Begum, S. H., & Bernard, M. J. Sheeba, "Semantic web to e-Learning content," *International Journal*, vol. 2, no. 10, pp. 58-66, Oct 2012.
- [154] H., Gayo, J. E. L., & Paule-Ruiz, M. García-González, "Enhancing e-learning content by using semantic web technologies," *IEEE Transactions on Learning Technologies*, vol. 10, no. 4, pp. 544-550, Nov 2016.
- [155] M. Masud, "Collaborative e-learning systems using semantic data interoperability," *Computers in Human Behavior*, vol. 61, pp. 127-135, Aug 2016.
- [156] N. Ghatasheh, "Knowledge level assessment in e-learning systems using machine learning and user activity analysis," *International Journal of Advanced Computer Science and Applications*, vol. 6, no. 4, pp. 107-113, Apr 2015.
- [157] M. P., Delgado, M., & Pegalajar, M. C. Cuéllar, "Improving learning management through semantic web and social networks in e-learning environments," *Expert systems with applications*, vol. 38, no. 4, pp. 4181-4189, Apr 2011.
- [158] M. P., Delgado, M., & Pegalajar, M. C. Cuéllar, "Improving learning management through semantic web and social networks in e-learning environments," *Expert systems with applications*, vol. 38, no. 4, pp. 4181-4189, Apr 2011.
- [159] J. K., Niu, Z., & Mustafa, G. Tarus, "Knowledge-based recommendation: a review of ontology-based recommender systems for e-learning," *Artificial intelligence review*, vol. 50, pp. 21-48, Jun 2018.
- [160] N. W., Ferdiana, R., & Kusumawardani, S. S. Rahayu, "A systematic review of ontology use in E-Learning recommender system," *Computers and Education: Artificial Intelligence*, vol. 3, p. 100047, Jan 2022.
- [161] L. C. Belcadhi, "Personalized feedback for self assessment in lifelong learning environments based on semantic web," *Computers in Human Behavior*, vol. 55, pp. 562-570, Feb 2016.
- [162] A. E., Canós, J. H., & Penadés, M. C. Labib, "On the way to learning style models integration: a Learner's Characteristics Ontology," *Computers in Human Behavior*, vol. 73, pp. 433-445, Aug 2017.
- [163] M., Fernández-Breis, J. T., Castellanos-Nieves, D., Frutos-Morales, F., & Prendes-Espinosa, M. P. del Mar Sánchez-Vera, "Semantic Web technologies for generating feedback in online assessment environments," *Knowledge-Based Systems*, vol. 33, pp. 152-165, Sep 2012.
- [164] F. J., Alcaraz Calero, J. M., Bernal Bernabé, J., Marín Pérez, J. M., Martínez Pérez, G., & Gómez Skarmeta, A. F. García Clemente, "Semantic Web-based management of routing configurations," *Journal of Network and Systems Management*, vol. 19, pp. 209-229, Jun 2011.
- [165] A. H., Clemente, F. J. G., Pérez, M. G., & Pérez, G. M. Celdrán, "SeCoMan: A semanticaware policy framework for developing privacy-preserving and context-aware smart applications," *IEEE Systems Journal*, vol. 10, no. 3, pp. 1111-1124, Jan 2014.

- [166] F. A. M., Burguillo, J. C., & Nistal, M. L. Fonte, "An intelligent tutoring module controlled by BDI agents for an e-learning platform," *Expert Systems with Applications*, vol. 39, no. 8, pp. 7546-7554, Jun 2012.
- [167] J. K., Niu, Z., & Yousif, A. Tarus, "A hybrid knowledge-based recommender system for elearning based on ontology and sequential pattern mining," *Future Generation Computer Systems*, vol. 72, pp. 37-48, Jul 2017.
- [168] S., Loia, V., & Veniero, M. Clemente, "Applying cognitive situation awareness to collision avoidance for harbour last-mile area safety," *Journal of Ambient Intelligence and Humanized Computing*, vol. 5, pp. 741-745, Oct 2014.
- [169] C., Ching-López, A., Lefranc, G., Loia, V., & Herrera-Viedma, E. Porcel, "Sharing notes: An academic social network based on a personalized fuzzy linguistic recommender system," *Engineering Applications of Artificial Intelligence*, vol. 75, pp. 1-10, Oct 2018.
- [170] J. K., Niu, Z., & Yousif, A. Tarus, "A hybrid knowledge-based recommender system for elearning based on ontology and sequential pattern mining," *Future Generation Computer Systems*, vol. 72, pp. 37-48, Jul 2017.
- [171] C., & Melis, E. Ullrich, "Complex course generation adapted to pedagogical scenarios and its evaluation," *Journal of Educational Technology & Society*, vol. 13, no. 2, pp. 102-115, Apr 2010.
- [172] N., & Toti, D. Capuano, "Experimentation of a smart learning system for law based on knowledge discovery and cognitive computing," *Computers in Human Behavior*, vol. 92, pp. 459-467, Mar 2019.
- [173] M., Mangione, G. R., Miranda, S., & Orciuoli, F. Gaeta, "Adaptive Feedback Improving Learningful Conversations at Workplace," in *International Association for Development of the Information Society. LADIS International Conference e-Learning*, Prague, Czech Republic, 2013.
- [174] G., Gaeta, M., Granito, A., Orciuoli, F., & Loia, V. D'aniello, "Sustaining self-regulation processes in seamless learning scenarios by situation awareness," in 2015 IEEE International Multi-Disciplinary Conference on Cognitive Methods in Situation Awareness and Decision, Orlando, USA, 2015, pp. 101-105.
- [175] S., Mangione, G. R., Orciuoli, F., Gaeta, M., & Loia, V. Miranda, "Automatic generation of assessment objects and Remedial Works for MOOCs," in 2013 12th International Conference on Information Technology Based Higher Education and Training (ITHET), Turkey, 2013, pp. 1-8.
- [176] N., Dell'Angelo, L., Orciuoli, F., Miranda, S., & Zurolo, F. Capuano, "Ontology extraction from existing educational content to improve personalized e-Learning experiences," in 2009 IEEE International Conference on Semantic Computing, Berkeley, CA, USA, 2009, pp. 577-582.
- [177] N., Salerno, S., Mangione, G. R., & Pierri, A. Capuano, "Semantically connected learning resources fostering intuitive guided learning," *International Journal of Continuing Engineering Education and Life Long Learning 6*, vol. 24, no. 2, pp. 122-140, Jan 2014.
- [178] P., Orciuoli, F., Paolozzi, S., Ritrovato, P., & Toti, D. Del Nostro, "A semantic-based

architecture for managing knowledge-intensive organizations: the ARISTOTELE platform," in *Web Information Systems Engineering–WISE 2011 and 2012 Workshops*, Sydney Australia, 2013, pp. 133-146.

- [179] P., Orciuoli, F., Paolozzi, S., Ritrovato, P., & Toti, D. Del Nostro, "ARISTOTELE: a semantic-driven platform for enterprise management," in 2013 27th International Conference on Advanced Information Networking and Applications Workshops, Barcelona, Spain, 2013, pp. 44-49.
- [180] The Publications Office of the European Union. (2023, Jan) eurovoc. [Online]. http://eurovoc.europa.eu/
- [181] Corte Suprema di Cassazione Centro Elettronico di Documentazione. (2023, Jan) italgiure. [Online]. https://www.italgiure.giustizia.it/
- [182] (2023, Jun) DeJure. [Online]. https://www.iusexplorer.it
- [183] D., & Longhi, A. Toti, "SEMANTO: a graphical ontology management system for knowledge discovery," *Journal of Ambient Intelligence and Humanized Computing*, vol. 9, pp. 1075-1084, Aug 2018.
- [184] N., De Maio, C., Salerno, S., & Toti, D. Capuano, "A methodology based on commonsense knowledge and ontologies for the automatic classification of legal cases," in *In Proceedings of the 4th international conference on web intelligence, mining and semantics (WIMS14)*, Greece, 2014, pp. 1-6.
- [185] N., Longhi, A., Salerno, S., & Toti, D. Capuano, "Ontology-driven generation of training paths in the legal domain," *International Journal of Emerging Technologies in Learning (iJET)*, vol. 10, no. 7, pp. 14-22, Jan 2015.
- [186] D., Atzeni, P., & Polticelli, F. Toti, "A knowledge discovery methodology for semantic categorization of unstructured textual sources," in 2012 Eighth International Conference on Signal Image Technology and Internet Based Systems, Naples Italy, 2012, pp. 944-951.
- [187] D., & Rinelli, M. Toti, "On the road to speed-reading and fast learning with CONCEPTUM," in 2016 International Conference on Intelligent Networking and Collaborative Systems (INCoS), Ostrava, Czech Republic, 2016, pp. 357-361.
- [188] Z., Nakamura, Y., Jang, S., Kajita, S., & Mase, K. Yu, "Ontology-based semantic recommendation for context-aware e-learning," in *Ubiquitous Intelligence and Computing: 4th International Conference, Springer Berlin Heidelberg*, Hong Kong, China, 2007, pp. 898-907.
- [189] Michael J. O'Brien. (2015, Jul) researchgate. [Online]. https://www.researchgate.net/figure/Subterranean-portion-of-a-giant-leafcutter-ant-nest-in-Brazil-Concrete-was-poured-into\_fig3\_282628837
- [190] James F. Taulman. (2017, Apr) researchgate. [Online]. https://www.researchgate.net/figure/The-first-wild-honey-bee-nest-observed-in-a-forestsnag-den-at-the-Woodridge-Public-Use\_fig4\_316625159

- [191] Omrane Bouketir, Tarek Bouktir Sebaa Haddi. (2018, Dec) researchgate. [Online]. https://www.researchgate.net/figure/a-attacking-prey-b-hunting-prey-bywolves\_fig2\_329883862
- [192] D. C. Marinescu, Cloud computing: theory and practice.: Morgan Kaufmann, 2022.
- [193] L., Von Laszewski, G., Younge, A., He, X., Kunze, M., Tao, J., & Fu, C. Wang, "Cloud computing: a perspective studyNew," in *Generation Computing2010282137*, 2010, p. 146.
- [194] L., Luo, Z., Du, Y., & Guo, L. Qian, "Cloud computing: An overview," in *Cloud Computing: First International Conference, CloudCom 2009*, Beijing, China, December 1-4, 2009, pp. 626-631.
- [195] P., & Grance, T. Mell. (2009) "Draft NIST working definition of cloud computing". Referenced on June. 3rd, 15(32), 2. Online at http://csrc.nist.gov/.
- [196] T., Wu, C., & Chang, E. Dillon, "Cloud computing: issues and challenges," in 2010 24th IEEE international conference on advanced information networking and applications, 2010, April, pp. 27-33.
- [197] (2022, Dec) scopus. [Online]. https://www.scopus.com/term/analyzer.uri?sid=ac8b3a8162b2a5db41dfae28733bd43f&ori gin=resultslist&src=s&s=TITLE-ABS-KEY%28Cloud+Computing%29&sort=plff&sdt=b&sot=b&sl=30&count=124715&analyzeResults=Analyze+results&txGid=3410f35 cf90c0a48cafdf3782ba4bb17
- [198] (2022, Oct) trends.google. [Online]. https://trends.google.com/trends/explore?date=all&q=%2Fm%2F02y\_9m3,%2Fm%2F02 vnd10,%2Fm%2F012dtj13
- [199] Brady Forrest. (2006, Sept) radar.oreilly. [Online]. http://radar.oreilly.com/2006/09/zimkihosted-javascript-enviro.html
- [200] (2023, March) aws.amazon. [Online]. http://aws.amazon.com
- [201] (2023, March) azure.microsoft. [Online]. https://azure.microsoft.com/en-us/
- [202] H., Zeebaree, S., Zebari, R., Zeebaree, D., Ahmed, O., & Salih, A. Shukur, "Cloud computing virtualization of resources allocation for distributed systems," *Journal of Applied Science and Technology Trends*, vol. 1, no. 3, pp. 98-105, Jun 2020.
- [203] T. S., & Kadir, S. L. S. A. Hew, "Predicting the acceptance of cloud-based virtual learning environment: The roles of Self Determination and Channel Expansion Theory," *Telematics* and Informatics, vol. 33, no. 4, pp. 990-1013, Nov 2016.
- [204] Y. M. Cheng, "How does task-technology fit influence cloud-based e-learning continuance and impact?," *Education+ Training*, vol. 61, no. 4, pp. 480-499, May 2019.
- [205] M., & Wang, J. Jou, "Observations of achievement and motivation in using cloud computing driven CAD: Comparison of college students with high school and vocational high school

backgrounds," Computers in Human Behavior, vol. 29, no. 2, pp. 364-369, Mar 2013.

- [206] Y. T., Wen, M. L., Jou, M., & Wu, D. W. Lin, "A cloud-based learning environment for developing student reflection abilities," *Computers in Human Behavior*, vol. 32, no. 1, pp. 244-252, Mar 2014.
- [207] W. L., & Chau, P. Y. Shiau, "Understanding behavioral intention to use a cloud computing classroom: A multiple model comparison approach," *Information & Management*, vol. 53, no. 3, pp. 355-365, Apr 2016.
- [208] V., Colomo-Palacios, R., Soto-Acosta, P., & Misra, S. Stantchev, "Learning management systems and cloud file hosting services: A study on students' acceptance," *Computers in Human Behavior*, vol. 31, pp. 612-619, Feb 2014.
- [209] (2022, Dec) scopus. [Online]. https://www.scopus.com/term/analyzer.uri?sid=ed54cc200257bcc9dc22f561d1c8828d&ori gin=resultslist&src=s&s=TITLE-ABS-KEY%28Cloud+based+learning%29&sort=plff&sdt=b&sot=b&sl=35&count=22378&analyzeResults=Analyze+results&txGid=e284e871 5f604123fe1090694d4ae12e
- [210] A., Youssef, A., & Emam, A. Alshwaier, "A new trend for e-learning in KSA using educational clouds," *Advanced Computing*, vol. 3, no. 1, p. 81, Jan 2012.
- [211] K., Kefalas, P., Dranidis, D., Hatziapostolou, T., & Cowling, A. Pireva, "Cloud e-learning: A new challenge for multi-agent systems," in *Agent and Multi-Agent Systems: Technologies and Applications: Proceedings of the 8th International Conference KES-AMSTA 2014, Springer International Publishing*, Chania, Greece, June 2014, pp. 277-287.
- [212] S. T., Alam, S., Khan, Z. A., & Gupta, A. Siddiqui, "Cloud-based e-learning: using cloud computing platform for an effective e-learning," in *Smart Innovations in Communication and Computational Sciences: Proceedings of ICSICCS-2018*, Springer Singapore, 2019, pp. 335-346.
- [213] G., & Ahmed, I. S. Bibi, "A Comprehensive survey on e-learning system in cloud computing environment," *Eng. Sci. Technol. Int. Res. J*, vol. 1, no. 1, pp. 43-50, 2017.
- [214] A., Ansari, A. M., Barga, M. M., & Albhaishi, A. Alam, "Is mobile cloud computing efficient for E-learning?," in *Proceedings of the Second International Conference on Computer and Communication Technologies: IC3T 2015*, Springer India, 2016, pp. 109-123.
- [215] (2018, Mar) elearningindustry.com. [Online]. https://elearningindustry.com/7-cloud-based-authoring-tools-compared.
- [216] (2018, Feb ) easygenerator.com. [Online]. https://www.easygenerator.com/
- [217] (2018, Mar) lectoraonline.com. [Online]. http://lectoraonline.com/app/BN0057/product.html
- [218] (2022, Nov) www.lectoraonline.com. [Online]. https://www.lectoraonline.com/

- [219] (2018, Feb) ecoach.com. [Online]. https://ecoach.com/
- [220] (2022, Nov) e-coach.co.uk. [Online]. https://e-coach.co.uk/
- [221] (2018, Mar ) www.ruzuku.com. [Online]. https://www.ruzuku.com/?utm\_campaign=elearningindustry.com&utm\_source=%2F7cloudbased-authoring-tools-compared&utm\_medium=link
- [222] (2022, Nov) ruzuku. [Online]. https://www.ruzuku.com/
- [223] (2018, Mar) www.ispringsolutions.com. [Online]. https://www.ispringsolutions.com/ispring-learn/pricing.html
- [224] (2022, Nov) www.ispringsolutions.com. [Online]. https://www.ispringsolutions.com/ispring-suite
- [225] (2022, Nov) www.litmos.com. [Online]. https://www.litmos.com/platform/authoring-tools
- [226] (2022, Nov) www.smartsparrow.com. [Online]. https://www.smartsparrow.com/
- [227] (2022, Nov) coassemble.com. [Online]. https://coassemble.com/
- [228] (2022, Nov) www.elucidat.com. [Online]. https://www.elucidat.com/
- [229] Imed Bouchrika. (2023, Feb) research.com. [Online]. https://research.com/education/bestlms-for-schools
- [230] (2023, Mar) classroom.google.com. [Online]. https://classroom.google.com
- [231] (2023, Mar) wikipedia.org. [Online]. https://en.wikipedia.org/wiki/Google\_for\_Education
- [232] (2023, Mar) anthology.com. [Online]. https://www.anthology.com/enemea/products/teaching-and-learning/learning-effectiveness/blackboard-learn
- [233] (2023, Mar) www.instructure.com. [Online]. https://www.instructure.com/en-au/canvas
- [234] (2023, Mar) moodle.org. [Online]. https://moodle.org/
- [235] (2023, Mar) www.d2l.com. [Online]. https://www.d2l.com/brightspace/
- [236] (2023, Mar) aws.amazon.com. [Online]. https://aws.amazon.com/education/?wwpscards.sort-by=item.additionalFields.sortDate&wwps-cards.sort-order=desc
- [237] (2023, Mar) edu.google.com. [Online]. https://edu.google.com/why-google/customerstories/?modal\_active=none
- [238] (2023, Mar) class.com/collaborate. [Online]. https://www.class.com/collaborate
- [239] B., Zheng, Q., Yang, J., Li, H., & Qiao, M. Dong, "An e-learning ecosystem based on cloud computing infrastructure," in *Advanced Learning Technologies, 2009. ICALT 2009. Ninth IEEE*

International Conference, Riga, Latvia, July 2009, pp. 125-127.

- [240] (2022, Dec) scopus.com. [Online]. https://www.scopus.com/term/analyzer.uri?sid=eea10e15a85fec1a33575424fe979a74&origi n=resultslist&src=s&s=TITLE-ABS-KEY%28Learning+object+recommendation%29&sort=plff&sdt=b&sot=b&sl=45&count=1570&analyzeResults=Analyze+results&txGid=907e1e7b2 172a88a05f10f5
- [241] S. M., Siewe, F., & He, Y. Nafea, "A novel algorithm for course learning object recommendation based on student learning styles," in 2019 International Conference on Innovative Trends in Computer Engineering (ITCE), United States, 2019, February, pp. 192-201.
- [242] S. S., Prasad, P. W. C., Alsadoon, A., & Maag, A. Khanal, "A systematic review: machine learning based recommendation systems for e-learning," *Education and Information Technologies*, vol. 25, pp. 2635-2664, Jul 2020.
- [243] J., Yoo, J., Lim, H., Qiu, H., Kozareva, Z., & Galstyan, A. Kim, "Sentiment prediction using collaborative filtering," *Proceedings of the International AAAI Conference on Web and Social Media*, vol. 7, no. 1, pp. 685-688, 2013.
- [244] R., & Martinez, L. Yera, "Fuzzy tools in recommender systems: A survey," International Journal of Computational Intelligence Systems, vol. 10, no. 1, p. 776, 2017.
- [245] J. B., Frankowski, D., Herlocker, J., & Sen, S. Schafer, "Collaborative filtering recommender systems," *The adaptive web: methods and strategies of web personalization*, pp. 291-324, 2007.
- [246] M., & Mann, S. Sharma, "A survey of recommender systems: approaches and limitations," *International journal of innovations in engineering and technology*, vol. 2, no. 2, pp. 8-14, 2013.
- [247] G., & Tuzhilin, A. Adomavicius, "Toward the next generation of recommender systems: A survey of the state-of-the-art and possible extensions," *IEEE transactions on knowledge and data engineering*, vol. 17, no. 6, pp. 734-749, 2005.
- [248] J., Ortega, F., Hernando, A., & Alcalá, J. Bobadilla, "Improving collaborative filtering recommender system results and performance using genetic algorithms," *Knowledge-based* systems, vol. 24, no. 8, pp. 1310-1316, 2011.
- [249] X. Liu, "A collaborative filtering recommendation algorithm based on the influence sets of e-learning group's behavior," *Cluster Computing*, vol. 22, no. Suppl 2, pp. 2823-2833, Mar 2019.
- [250] A., & Ursino, D. Nocera, "An approach to providing a user of a "social folksonomy" with recommendations of similar users and potentially interesting resources," *Knowledge-Based Systems*, vol. 24, no. 8, pp. 1277-1296, Dec 2011.
- [251] P. Y., & Yang, H. C. Wang, "Using collaborative filtering to support college students' use of online forum for English learning," *Computers & Education*, vol. 59, no. 2, pp. 628-637, Sep 2012.

- [252] P., Tang, T. Y., & McCalla, G. Winoto, "Contexts in a paper recommendation system with collaborative filtering," *International Review of Research in Open and Distance Learning*, vol. 13, no. 5, pp. (56–75)https://www.irrodl.org/index.php/irrodl/article/view/1243, Feb 2012.
- [253] O., & El Bachari, E. Bourkoukou, "E-learning personalization based on collaborative filtering and learner's preference," *Journal of Engineering Science and Technology*, vol. 11, no. 11, pp. 1565-1581, 2016.
- [254] S., Shabanali-Fami, F., & Moradi, H. Fatahi, "An empirical study of using sequential behavior pattern mining approach to predict learning styles," *Education and Information Technologies*, vol. 23, pp. 1427-1445, Jul 2018.
- [255] M. M., Aldraiweesh, A., Alturki, U., Tortorella, R., Yang, J., Chang, T. W., & Graf, S. El-Bishouty, "Use of Felder and Silverman learning style model for online course design," *Educational Technology Research and Development*, vol. 67, no. 1, pp. 161-177, Feb 2019.
- [256] L., Liang, J., Zhu, Y., Luo, Y., Sun, L., & Zheng, X. Guo, "Collaborative filtering recommendation based on trust and emotion," *Journal of Intelligent Information Systems*, vol. 53, pp. 113-135, Aug 2019.
- [257] P., Kant, V., & Bharadwaj, K. K. Dwivedi, "Learning path recommendation based on modified variable length genetic algorithm," *Education and information technologies*, vol. 23, pp. 819-836, Mar 2018.
- [258] I., Elouahbi, R., El Khoukhi, F., Karite, T., & Zouiten, H. Kamsa, "Optimizing collaborative learning path by ant's optimization technique in e-learning system," in 15th International Conference on Information Technology Based Higher Education and Training (ITHET), ISTANBUL, 2016, pp. 1-5.
- [259] C., Tan, K. C., Li, H., & Hong, G. S. Zhang, "A cost-sensitive deep belief network for imbalanced classification," *IEEE transactions on neural networks and learning systems*, vol. 30, no. 1, pp. 109-122, May 2018.
- [260] F., Basile, P., Musto, C., Lops, P., Caputo, A., de Gemmis, M., & Semeraro, G. Narducci, "oncept-based item representations for a cross-lingual content-based recommendation process," *Information Sciences*, vol. 374, pp. 15-31, Dec 2016.
- [261] H., Hummel, H. G., & Koper, R. Drachsler, "Personal recommender systems for learners in lifelong learning networks: the requirements, techniques and model," *International Journal of Learning Technology*, vol. 3, no. 4, pp. 404-423, Jan 2008.
- [262] Y., Yu, B., Zhang, X., & Yu, Y. Chen, "Topic modeling for evaluating students' reflective writing: a case study of pre-service teachers' journals," in *Proceedings of the sixth international* conference on learning analytics & knowledge, Edinburgh, United Kingdom, 2016, April, pp. 1-5.
- [263] S. V., Pai, R. M., & MM, M. P. Kolekar, "Rule based adaptive user interface for adaptive Elearning system," *Education and Information Technologies*, vol. 24, pp. 613-641, Jan 2019.
- [264] J., Shen, X., Liu, H., Yi, B., & Zhang, Z. Shu, "A content-based recommendation algorithm

for learning resources," Multimedia Systems, vol. 24, no. 2, pp. 163-173, Mar 2018.

- [265] M. M., & Abdullah, N. A. Rahman, "A personalized group-based recommendation approach for Web search in E-learning," *IEEE Access*, vol. 6, pp. 34166-34178, Jun 2018.
- [266] C. K., Campos, F., Ströele, V., David, J. M. N., & Braga, R. Pereira, "BROAD-RSIeducational recommender system using social networks interactions and linked data," *Journal* of *Internet Services and Applications*, vol. 9, no. 1, pp. 1-28, Dec 2018.
- [267] L. O., Valencia-García, R., Rodríguez-González, A., Alor-Hernández, G., & Samper-Zapater, J. J. Colombo-Mendoza, "RecomMetz: A context-aware knowledge-based mobile recommender system for movie showtimes," *Expert Systems with Applications*, vol. 42, no. 3, pp. 1202-1222, Feb 2015.
- [268] A., Boratto, L., Stettinger, M., & Tkalčič, M. Felfernig, .: Cham: Springer, 2018, p. 302.
- [269] E., & Meziane, F. Aeiad, "An adaptable and personalised E-learning system applied to computer science Programmes design," *Education and Information Technologies*, vol. 24, no. 2, pp. 1485-1509, Mar 2019.
- [270] P., & Vicheanpanya, J. Phobun, "Adaptive intelligent tutoring systems for e-learning systems," *Procedia-Social and Behavioral Sciences*, vol. 2, no. 2, pp. 4064-4069, Jan 2010.
- [271] S., & Niu, Z. Wan, "An e-learning recommendation approach based on the self-organization of learning resource," *Knowledge-Based Systems*, vol. 160, pp. 71-87, Nov 2018.
- [272] A., Hedhili, A., & Chaari, W. L. Trifa, "Knowledge tracing with an intelligent agent, in an elearning platform," *Education and Information Technologies*, vol. 24, pp. 711-741, Jan 2019.
- [273] A., Wettayaprasit, W., & Gilbert, L. Nitchot, "Assistive tool for constructing knowledge structures and suggesting related study materials links," *Education and Information Technologies*, vol. 24, pp. 219-230, Jan 2019.
- [274] K. I., & Abdullah, N. A. Ghauth, "Learning materials recommendation using good learners' ratings and content-based filtering," *Educational technology research and development*, vol. 58, pp. 711-727, Dec 2010.
- [275] X., Niu, Z., Chen, W., Shi, C., Niu, K., & Liu, D. Zhao, "A hybrid approach of topic model and matrix factorization based on two-step recommendation framework," *Journal of Intelligent Information Systems*, vol. 44, pp. 335-353, Jun 2015.
- [276] S., & Satratzemi, M. Karga, "A hybrid recommender system integrated into LAMS for learning designers," *Education and Information Technologies*, vol. 23, pp. 1297-1329, May 2018.
- [277] M., Zhu, W., Zhang, W., Abidi, S. M. R., & Ali, S. Hussain, "Using machine learning to predict student difficulties from learning session data," *Artificial Intelligence Review*, vol. 52, pp. 381-407, Jun 2019.
- [278] X. W., Guo, Y., He, Y., Jiang, H., Wu, Y., & Li, X. Zhao, "We know what you want to buy: a demographic-based system for product recommendation on microblogs," in *Proceedings of the*

20th ACM SIGKDD international conference on Knowledge discovery and data mining, New York, USA, 2014, August, pp. 1935-1944.

- [279] R. Burke. (2007) Hybrid web recommender systems. The adaptive web: methods and strategies of web personalization, 377-408. [Online]. https://www.researchgate.net/profile/Robin-Burke-3/publication/200121024\_Hybrid\_Web\_Recommender\_Systems/links/5464ddc30cf2f5eb1 7ff314a/Hybrid-Web-Recommender-Systems.pdf
- [280] K., Manouselis, N., Ochoa, X., Wolpers, M., Drachsler, H., Bosnic, I., & Duval, E. Verbert, "Context-aware recommender systems for learning: a survey and future challenges," *IEEE transactions on learning technologies*, vol. 5, no. 4, pp. 318-335, Apr 2012.
- [281] P., & Ahmadian, S. Moradi, "A reliability-based recommendation method to improve trustaware recommender systems," *Expert Systems with Applications*, vol. 42, no. 21, pp. 7386-7398, Nov 2015.
- [282] P., De Gemmis, M., Semeraro, G., Narducci, F., & Musto, C. Lops, "Leveraging the linkedin social network data for extracting content-based user profiles," in *Proceedings of the fifth ACM conference on Recommender systems*, Chicago, 2011, October, pp. 293-296.
- [283] B., Rokach, L., & Freilikhman, S. Shapira, "Facebook single and cross domain data for recommendation systems," User Modeling and User-Adapted Interaction, vol. 23, pp. 211-247, Apr 2013.
- [284] E. A., Phithakkitnukoon, S., & Dantu, R. Baatarjav, "Group recommendation system for facebook," in On the Move to Meaningful Internet Systems: OTM 2008 Workshops: OTM Confederated International Workshops and Posters, Monterrey, Mexico, November 9-14, 2008, pp. 211-219.
- [285] S., Banihashem, S. Y., Zin, N. A. M., Noah, S. A. M., & Malaysia, K. Shishehchi,
  "Ontological approach in knowledge based recommender system to develop the quality of elearning system," *Australian Journal of Basic and Applied Sciences*, vol. 6, no. 2, pp. 115-123, 2012.
- [286] J., Ignatov, D. I., Kuznetsov, S. O., & Dedene, G. Poelmans, "Formal concept analysis in knowledge processing: A survey on applications," *Expert systems with applications*, vol. 40, no. 16, pp. 6538-6560, Nov 2013.
- [287] G., Métral, C., Teller, J., Tweed, C., Roussey, C., Pinet, F., Kang, M.A. and Corcho, O. Falquet, , Ontologies in Urban development projects, Ed. London: Springer , 2011, pp. 9-38.
- [288] C., Porcel, C., Bernabé-Moreno, J., & Herrera-Viedma, E. Martinez-Cruz, "A model to represent users trust in recommender systems using ontologies and fuzzy linguistic modeling," *Information Sciences*, vol. 311, pp. 102-118, Aug 2015.
- [289] S., Hsiao, I. H., & Brusilovsky, P. Sosnovsky, "Adaptation "in the wild": Ontology-based personalization of open-corpus learning material," in 21st Century Learning for 21st Century Skills: 7th European Conference of Technology Enhanced Learning, EC-TEL 2012, Saarbrücken,

Germany, September 18-21, 2012, pp. 425-431.

- [290] Q., Gao, F., & Wang, H. Han, "Ontology-based learning object recommendation for cognitive considerations," in WCICA 2010 : 8th World Congress on Intelligent Control and Automation, Jinan, China, 2010, July, pp. 2746-2750.
- [291] C., & Zhekova, D. Klaussner, "Lexico-syntactic patterns for automatic ontology building," in Proceedings of the Second Student Research Workshop associated with RANLP 2011, Hissar, Bulgaria, 2011, September, pp. 109-114.
- [292] S., Yoon, V., & Thomas, M. A. Elnagar, "An automatic ontology generation framework with an organizational perspective," in 53rd Hawaii International Conference on System Sciences, HICSS 2020, Maui, Hawaii, USA, January , 2020, pp. 4860-4869.
- [293] D. A., & Solomou, G. D. Koutsomitropoulos, "A learning object ontology repository to support annotation and discovery of educational resources using semantic thesauri," *IFLA journal*, vol. 44, no. 1, pp. 4-22, Mar 2018.
- [294] D., Sarkar, S., & Ghose, S. Roy, "A comparative study of learning object metadata, learning material repositories, metadata annotation & an automatic metadata annotation tool," *Advances in Semantic Computing*, vol. 2, pp. 103-126, 2010.
- [295] (2022, Dec) scopus. [Online]. https://www.scopus.com/term/analyzer.uri?sid=b3b9a67ef3e3f2c23b6e01ebb4f4da65&ori gin=resultslist&src=s&s=TITLE-ABS-KEY%28Learning+object+annotation%29&sort=plff&sdt=b&sot=b&sl=41&count=3070&analyzeResults=Analyze+results&txGid=85dc9961a 8fbf9194b64c5cbd92
- [296] X., Cardinaels, K., Meire, M., & Duval, E. Ochoa, "Frameworks for the Automatic Indexation of Learning Management Systems Content into Learning Object Repositories," in World Conference on Educational Multimedia, Hypermedia & Telecommunications, EDMEDLA 2005, Montreal, Canada, 2005 Jun, pp. 1407-1414.
- [297] (2022, Dec) dublincore.org. [Online]. http://dublincore.org/
- [298] (2022, Dec) wikipedia.org. [Online]. https://en.wikipedia.org/wiki/Dublin\_Core
- [299] (2022, Dec) standards.ieee.org. [Online]. https://standards.ieee.org/ieee/1484.12.1/7699/
- [300] (2022, Dec) ieeexplore.ieee.org. [Online]. https://ieeexplore.ieee.org/document/9262118
- [301] T., Nyulas, C., Noy, N. F., & Musen, M. A. Tudorache, "WebProtégé: A collaborative ontology editor and knowledge acquisition tool for the web," *Semantic web*, vol. 4, no. 1, pp. 89-99, Jan 2013.
- [302] (2022, Dec) w3.org. [Online]. https://www.w3.org/TR/2008/WD-skos-reference-20080829/skos.html

- [303] (2022, Dec) ukoln.ac.uk. [Online]. http://www.ukoln.ac.uk/metadata/dcdot/
- [304] H., Giles, C. L., Manavoglu, E., Zha, H., Zhang, Z., & Fox, E. A. Han, "Automatic document metadata extraction using support vector machines," in 2003 Joint Conference on Digital Libraries, Houston, Texas, USA, May 2003, pp. 37-48.
- [305] C., & Inman, D. Jenkins, "Server-side automatic metadata generation using Qualified Dublin Core and RDF," in *Proceedings 2000 Kyoto International Conference on Digital Libraries: Research and Practice*, Kyoto, Japan, 2000, November, pp. 262-269.
- [306] Y., Zhu, Q., & Cao, Y. Li, "Automatic metadata generation based on neural network," in Proceedings of the 3rd international conference on Information security, Shanghai, China, 2004, November, pp. 192-197.
- [307] J., Gasevic, D., & Devedzic, V. Jovanovic, "Ontology-based automatic annotation of learning content," *International Journal on Semantic Web and Information Systems (IJSWIS)*, vol. 2, no. 2, pp. 91-119, Apr 2006.
- [308] K., Meire, M., & Duval, E. Cardinaels, "Automating metadata generation: the simple indexing interface," in *Proceedings of the 14th international conference on World Wide Web*, Chiba, Japan, 2005, May, pp. 548-556.
- [309] (2022, Dec) w3.org. [Online]. https://www.w3.org/2001/sw/wiki/KIM\_Platform
- [310] B., Kiryakov, A., Kirilov, A., Manov, D., Ognyanoff, D., & Goranov, M. Popov, "KIMsemantic annotation platform," in *The Semantic Web-ISWC 2003: Second International Semantic Web Conference*, Sanibel Island, FL, USA, 2003, October, pp. 834-849.
- [311] M., Kiryakov, A., Simov, K., & Petrov, S. Damova, "Mapping the central LOD ontologies to PROTON upper-level ontology," in Ontology Matching, Proceedings of the ISWC Workshop, November 2010, pp. 61-72.
- [312] (2022, Dec) gate.ac.uk. [Online]. https://gate.ac.uk/
- [313] (2022, Dec) scopus.com. [Online]. https://www.scopus.com/term/analyzer.uri?sid=bd0450a61ac10890c31a3fc53c46e8fa&orig in=resultslist&src=s&s=TITLE-ABS-KEY%28Topic+extraction%29&sort=plff&sdt=b&sot=b&sl=31&count=39017&analyzeResults=Analyze+results&txGid=fa79abd2 2fd8a13956fab5187d4286d9
- [314] (2022, Dec) scopus.com. [Online]. https://www.scopus.com/term/analyzer.uri?sid=cd42634420f8071c24cd6b0c17af987c&ori gin=resultslist&src=s&s=TITLE-ABS-KEY%28entity+extraction%29&sort=plff&sdt=b&sot=b&sl=32&count=12051&analyzeResults=Analyze+results&txGid=ca71a46f 8c62e2ad45303c8a15473773
- [315] M. W., Mohamed, A., & Yap, B. W. Berry, *Supervised and unsupervised learning for data science*. Cham, Switzerland: Springer Nature Switzerland AG 2020, 2019.

- [316] R. P., & Thabtah, F. Bunker, "A machine learning framework for sport result prediction," *Applied computing and informatics*, vol. 15, no. 1, pp. 27-33, Jan 2019.
- [317] T., Höpken, W., Fuchs, M., & Lexhagen, M. Menner, "Topic detection: identifying relevant topics in tourism reviews," in *Information and Communication Technologies in Tourism 2016: Proceedings of the International Conference*, Bilbao, Spain, 2016, February, pp. 411-423.
- [318] S., Dumais, S. T., Furnas, G. W., Landauer, T. K., & Harshman, R. Deerwester, "Indexing by latent semantic analysis," *Journal of the American society for information science*, vol. 41, no. 6, pp. 391-407, Sep 1990.
- [319] Chris, Robert Cooley, and Jason Rennie. Clifton, "Topcat: Data mining for topic identification in a text corpus," *IEEE transactions on knowledge and data engineering*, pp. 949-964, 2004.
- [320] Veselin, and Claire Cardie. Stoyanov, "Topic identification for fine-grained opinion analysis," Proceedings of the 22nd International Conference on Computational Linguistics (Coling 2008), pp. 817-824, 2008.
- [321] Kino, Rada Mihalcea, and William Moen. Coursey, "Using encyclopedic knowledge for automatic topic identification," *Proceedings of the Thirteenth Conference on Computational Natural Language Learning (CoNLL-2009)*, pp. 210-218, 2009.
- [322] Rada, and Andras Csomai. Mihalcea, "Wikify! Linking documents to encyclopedic knowledge," Proceedings of the sixteenth ACM conference on Conference on information and knowledge management, pp. 233-242, 2007.
- [323] A., Freidrich, B., Fehlman, W., Frolow, I., & Engels, D. W. Crane, "A Novel Methodology to Identify the Primary Topics Contained Within the COVID-19 Research Corpus," *SMU Data Science Review*, 2020.
- [324] A. K., Khader, A. T., Al-Betar, M. A., Naim, S., Makhadmeh, S. N., & Alyasseri, Z. A. A. Abasi, "A novel ensemble statistical topic extraction method for scientific publications based on optimization clustering," *Multimedia Tools and Applications*, pp. 37-82, 2021.
- [325] W., Wang, Y., Gao, C., & Yang, F. Zhou, "Emerging topic identification from app reviews via adaptive online biterm topic modeling," *Frontiers of Information Technology & Electronic Engineering*, vol. 23, no. 5, pp. 678-691, May 2022.
- [326] M. A., & Fard, F. H. Hadi, "Aobtm: Adaptive online biterm topic modeling for version sensitive short-texts analysis," in 2020 IEEE international conference on software maintenance and evolution (ICSME), Adelaide, Australia, 2020, September, pp. 593-604.
- [327] (2022, Dec) nltk.org. [Online]. http://www.nltk.org
- [328] (2023, April) wiktionary.org. [Online]. https://en.wiktionary.org/wiki/Wiktionary:Main\_Page
- [329] (2023, April) github.com. [Online]. https://github.com/shibing624/pycorrector

- [330] (2023, April) wikipedia.org. [Online]. https://en.wikipedia.org/wiki/Pointwise\_mutual\_information#:~:text=In%20statistics%2C %20probability%20theory%20and,if%20the%20events%20were%20independent.
- [331] (2023, April) rapidminer.com. [Online]. https://rapidminer.com/glossary/data-miningtools/
- [332] X., Guang-Ri, Q., Zhi-Ming, X., & Ya-Dong, W. Yong-Dong, "Research on text hierarchical topic identification algorithm based on the dynamic diverse thresholds clustering," 2009 *International Conference on Asian Language Processing. IEEE*, pp. 206-210, 2009.
- [333] Zhijun, Weilin Pan, and Yang Wang. Yang, "Group Cooperative Learning Model and Its Application Based on Group Knowledge Graph," 2021 IEEE 3rd International Conference on Computer Science and Educational Informatization (CSEI), pp. 324-327, 2021.
- [334] (2023, April) wikipedia.org. [Online]. https://www.wikipedia.org/
- [335] Jing, Zhijing Liu, and Hui Zhao. Wang, "Group recommendation using topic identification in social networks," *ixth International Conference on Intelligent Human-Machine Systems and Cybernetics. Vol. 1*, pp. 355-358, 2014.
- [336] David, and Ian H. Witten. Milne, "An open-source toolkit for mining Wikipedia," *Artificial Intelligence*, pp. 222-239, 2013.
- [337] Olena, Ian H. Witten, and David Milne. Medelyan, "Topic indexing with Wikipedia," *Proceedings of the AAAI WikiAI workshop*, pp. 19-24, 2008.
- [338] A., Milne, D., Frank, E., & Witten, I. H. Huang, "Clustering documents using a wikipediabased concept representation," *Pacific-Asia Conference on Knowledge Discovery and Data Mining. Springer, Berlin, Heidelberg*, 2009.
- [339] Vivi, Katja Filippova, and David N. Milne. Nastase, "Summarizing with Encyclopedic Knowledge," *TAC*, 2009.
- [340] Marcos Antonio Mouriño, Roberto Pérez Rodríguez, and Luis Anido Rifón. García, "Leveraging Wikipedia knowledge to classify multilingual biomedical documents," *Artificial intelligence in medicine*, pp. 37-57, 2018.
- [341] Marcos Antonio Mouriño, Roberto Pérez Rodríguez, and Luis Anido Rifón. García, "Wikipedia-based cross-language text classification," *Information Sciences*, pp. 12-28, 2017.
- [342] C., Gasparetti, F., Limongelli, C., Sciarrone, F., & Temperini, M. De Medio, "Automatic extraction of prerequisites among learning objects using wikipedia-based content analysis," *International conference on intelligent tutoring systems. Springer, Cham*, pp. 375-381, 2016.
- [343] M. A., Pérez-Rodríguez, R., Anido-Rifon, L., & Vilares-Ferro, M. Mouriño-García,
  "Wikipedia-based hybrid document representation for textual news classification," *Soft Computing*, pp. 6047-6065, 2018.

- [344] Gabriela, Monica Vladoiu, and Zoran Constantinescu. Moise, "a multi-agent system for evaluation and classification of OERs and OCW based on quality criteria," *E-Learning Paradigms and Applications. Springer, Berlin, Heidelberg*, pp. 185-227, 2014.
- [345] Rada, and Paul Tarau. Mihalcea, "Textrank: Bringing order into text," *Proceedings of the 2004 conference on empirical methods in natural language processing*, pp. 404-411, 2004.
- [346] J. M. Kleinberg, "Authoritative sources in a hyperlinked environment," *Journal of the ACM (JACM)*, vol. 46, no. 5, pp. 604-632, Sep 1999.
- [347] S., & Page, L. Brin, "The anatomy of a large-scale hypertextual web search engine," *Computer networks and ISDN systems*, vol. 30, no. 1-7, pp. 107-117, Apr 1998.
- [348] P., Van der Laan, G., & Talman, D. Herings, "Measuring the power of nodes in digraphs," Tinbergen Institute, Technical report 2001, October.
- [349] R., Tarau, P., & Figa, E. Mihalcea, "PageRank on semantic networks, with application to word sense disambiguation," in COLING 2004: Proceedings of the 20th International Conference on Computational Linguistics, Geneva, Switzerland, 2004, pp. 1126-1132.
- [350] P. D. Turney, "Learning to extract keyphrases from text," National Research Council. Institute for Information Technology, technical report ERB-1057, 1999.
- [351] E., Paynter, G. W., Witten, I. H., Gutwin, C., & Nevill-Manning, C. G. Frank, "Domain-Specific Keyphrase Extraction," in *JCAI'99: Proceedings of the 16th International Joint Conference* on Artificial Intelligence, Stockholm, Sweden, 1999, August, pp. 668-673.
- [352] A. Hulth, "Improved automatic keyword extraction given more linguistic knowledge," in Proceedings of the 2003 conference on Empirical methods in natural language processing, Augu, Japan, 2003, pp. 216-223.
- [353] Y., Li, W., Wei, F., & Lu, Q. Ouyang, "Learning similarity functions in graph-based document summarization," in *Computer Processing of Oriental Languages. Language Technology for the Knowledge-based Economy: 22nd International Conference, ICCPOL 2009*, Hong Kong, March 26-27, 2009, pp. 189-200.
- [354] G., Singhal, A., Mitra, M., & Buckley, C. Salton, "Automatic text structuring and summarization," *Information processing & management*, vol. 33, no. 2, pp. 193-207, Mar 1997.
- [355] S., & Zaragoza, H. Robertson, "The probabilistic relevance framework: BM25 and beyond," *Foundations and Trends*® *in Information Retrieval*, vol. 3, no. 4, pp. 333-389, Dec 2009.
- [356] S., Walker, S., Jones, S., Hancock-Beaulieu, M., Gatford, M. Robertson, "Okapi at TREC-3," in *Proceedings of The Third Text REtrieval Conference, TREC 1994*, Gaithersburg, Maryland, USA, November 2-4, 1994, pp. 109–126.
- [357] Y., Zhai, C. Lv, "Lower-bounding term frequency normalization," in *Proceedings of the 20th ACM Conference on Information and Knowledge Management, CIKM 2011*, Glasgow, United Kingdom, October 24-28, 2011, pp. 7–16.

- [358] Radim, and Petr Sojka. Rehurek, "Software framework for topic modelling with large corpora," *In Proceedings of the LREC 2010 workshop on new challenges for NLP frameworks*, 2010.
- [359] R., & Sojka, P. Řehůřek. (2011) Gensim—statistical semantics in python. [Online]. https://citeseerx.ist.psu.edu/document?repid=rep1&type=pdf&doi=b55fe23d7290f59d14e 51e7813f5950f5ff08b2b
- [360] D. M., Ng, A. Y., & Jordan, M. I. Blei, "Latent dirichlet allocation," *Journal of machine Learning research*, vol. 3, pp. 993-1022, Jan 2003.
- [361] G. W. Stewart, "On the early history of the singular value decomposition," *SLAM review*, vol. 35, no. 4, pp. 551-66, Dec 1993.
- [362] (2023, May) wikipedia.org. [Online]. https://en.wikipedia.org/wiki/Singular\_value\_decomposition
- [363] (2022, Dec) wikipedia.org. [Online]. https://en.wikipedia.org/wiki/FOAF
- [364] (2022, Dec) wikipedia.org. [Online]. https://en.wikipedia.org/wiki/Myers%E2%80%93Briggs\_Type\_Indicator
- [365] (2022, Dec) researchgate.ne. [Online]. https://www.researchgate.net/figure/The-elementsof-IEEE-LOM-standard\_fig1\_291327371
- [366] A. D., Solomou, G. D., Koutsomitropoulos, D. A., & Papatheodorou, T. Alexopoulos, "Enhancing digital repositories with learning object metadata," *Handbook of research on e-Learning standards and interoperability: frameworks and issues*, pp. 246-263, 2011.
- [367] Federico, Federico López, Luis Argerich, and Rosa Wachenchauzer. Barrios, "Variations of the similarity function of textrank for automated summarization," *arXiv preprint arXiv*, 2016.
- [368] (2021, Dec) pypi.org. [Online]. https://pypi.org/project/gensim/
- [369] (2021, Dec) wikipedia.org. [Online]. https://en.wikipedia.org/wiki/Wikipedia:FAQ/Categories
- [370] (2021, Dec) wikipedia.org. [Online]. https://en.wikipedia.org/wiki/Category:Business
- [371] Suman, and Oriol Vinyals. Ravuri, "Classification accuracy score for conditional generative models," *arXiv preprint arXiv*, 2019.
- [372] Tim, and James Hendler. Berners-Lee, "Publishing on the semantic web," *Nature*, pp. 1023-1024, 2001.
- [373] (2022, June) dartmouthlearning.net. [Online]. https://www.dartmouthlearning.net/pdf/Literacy-Employment.pdf
- [374] Prabhat. S. (2011, June) differencebetween.net. [Online]. http://www.differencebetween.net/miscellaneous/difference-between-education-and-

learning

- [375] (2022, June) differencebetween.net. [Online]. http://www.differencebetween.net/miscellaneous/difference-between-education-and-learning
- [376] C. Alliance, "Security guidance for critical areas of focus in cloud computing v3. 0," *Cloud Security Alliance*, vol. 15, pp. 1-176, 2011.
- [377] (2018, Mar) elearningindustry.com. [Online]. https://elearningindustry.com/7-cloud-based-authoring-tools-compared
- [378] R., & McKeown, K. R. Barzilay, "Sentence fusion for multidocument news summarization," *Computational Linguistics*, vol. 31, no. 3, pp. 297-328, Sep 2005.
- [379] G., & Radev, D. R. Erkan, "Lexrank: Graph-based lexical centrality as salience in text summarization," *Journal of artificial intelligence research*, vol. 22, pp. 457-479, Dec 2004.
- [380] L., Brin, S., Motwani, R., Winograd, T. Page, "The pagerank citation ranking: Bringing order to the web," in *Proceedings of the 7th International World Wide Web Conference*, Brisbane, Australia, 1998, pp. 161–172.